COLLECTIVE AGREEMENT

between

THE GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION

the members of which are:

FOUR SEASONS HOTEL
HYATT REGENCY VANCOUVER
RENAISSANCE VANCOUVER HOTEL HARBOURSIDE
THE WESTIN BAYSHORE, VANCOUVER

(each Hotel hereinafter referred to as an "Employer")

and

UNITE HERE LOCAL 40

(Affiliated with UNITE HERE International Union, BC Federation of Labour, Canadian Labour Congress and Change To Win and hereinafter referred to as the "Union")

TABLE OF CONTENTS

| ART | ICLE SUBJECT | PAGE |
|------|--|------|
| ART | ICLE 1 - INTRODUCTION | 11 |
| 1.01 | PURPOSE OF THIS AGREEMENT | |
| 1.01 | IDENTITY OF THE "EMPLOYER" | |
| ART | ICLE 2 - DURATION AND INTEGRITY OF AGREEMENT | 11 |
| 2.01 | DURATION OF AGREEMENT | 11 |
| 2.02 | EXCLUSION OF LABOUR RELATIONS CODE SECTIONS 50(2) AND 50(3) | 12 |
| 2.03 | WORK INTERRUPTION PROHIBITED | 12 |
| 2.04 | CONTRACTED SERVICES | 12 |
| 2.05 | IMPACT OF LEGISLATION | 12 |
| ART | ICLE 3 - UNION RECOGNITION AND RELATED | 13 |
| 3.01 | RECOGNITION OF EXCLUSIVE BARGAINING AGENT | 13 |
| 3.02 | REFUSAL TO WORK WITH NON-UNION EMPLOYEES AND RECOGNITION OF LEGAL PICKET LINES | 13 |
| 3.03 | PERFORMANCE OF BARGAINING UNIT WORK | 14 |
| 3.04 | HARASSMENT AND DISCRIMINATION | 14 |
| 3.05 | UNION BUTTONS | 14 |
| 3.06 | FAIR LABOUR SERVICES, PRODUCTS AND MATERIALS | 15 |
| 3.07 | UNION HOUSE CARD OR BAR CARD CONTRACT | 15 |
| 3.08 | UNION INVESTIGATION OF THE STANDING OF EMPLOYEES' CONDITIONS | 15 |
| ART | ICLE 4 - UNION SECURITY | 15 |
| 4.01 | MEMBERSHIP | 15 |
| 4.02 | NEW EMPLOYEES | 16 |

| ART | ICLE SUBJECT | PAGE |
|------|---|-------|
| 4.03 | CHECK-OFF: ASSIGNMENT OF WAGES | 16 |
| 4.04 | CHECK-OFF: PROCESS AND PROCEDURES | 16 |
| 4.05 | AN EMPLOYEE'S FAILURE TO MAINTAIN MEMBERSHIP IN GOOD STANDING | 18 |
| 4.06 | ADDITIONAL UNION ASSESSMENT | 18 |
| ART | ICLE 5 - SHOP STEWARDS AND SHOP CHAIRPERSON | 18 |
| 5.01 | SHOP STEWARDS | 18 |
| 5.02 | SHOP CHAIRPERSON | 19 |
| 5.03 | SHOP CHAIRPERSON AND OTHER SHOP STEWARDS | 19 |
| 5.04 | MANAGEMENT AND SHOP STEWARDS MEETING | 20 |
| 5.05 | SHOP STEWARDS MEETINGS | 20 |
| ART | ICLE 6 - MANAGEMENT RIGHTS | 21 |
| 6.01 | MANAGEMENT RIGHTS | 21 |
| ART | ICLE 7 - HIRING PROCESS AND PROBATION PERIOD | 21 |
| 7.01 | HIRING PROCESS | 21 |
| 7.02 | PROBATION PERIOD | 22 |
| ART | ICLE 8 - EMPLOYEE TRAINING | 22 |
| 8.01 | PRIVATE AND GOVERNMENT SUBSIDIZED TRAINING PROGR | AMS22 |
| ART | ICLE 9 - HOURS OF WORK AND RELATED | 23 |
| 9.01 | NORMAL STRAIGHT TIME HOURS OF WORK | 23 |
| 9.02 | SPLIT SHIFTS | 23 |
| 9.03 | SHIFT HOURS | 24 |

gvhea-local40 ii

| ARTI | CLE SUBJECT | PAGE | |
|--------|---|------|--|
| 9.04 | MAXIMIZING THE LENGTH OF SHIFTS | 25 | |
| 9.05 | ASSIGNMENT OF SHIFTS BY SENIORITY | 25 | |
| 9.06 | DAYS OFF | 25 | |
| 9.07 | TIME WORKED ON SIXTH AND SEVENTH CONSECUTIVE DAYS | 25 | |
| 9.08 | PAYMENT FOR TIME IN LIEU OF BREAKS | 25 | |
| 9.09 | UNPAID MEAL BREAKS | 26 | |
| 9.10 | REST PERIODS | 26 | |
| 9.11 | EMPLOYEE'S RESPONSIBILITY: WORK START TIME | 26 | |
| 9.12 | WORK SCHEDULES | 27 | |
| 9.13 | CHANGES IN WORK SCHEDULES | 28 | |
| 9.14 | SCHEDULING OF OVERTIME | 29 | |
| 9.15 | EXCHANGING SHIFTS | 30 | |
| ARTI | CLE 9.1 - SCHEDULING OF SHIFTS | 30 | |
| 9.1.01 | THE MAIN PRINCIPLES | 31 | |
| 9.1.02 | CATEGORIES OF EMPLOYEES | 31 | |
| 9.1.03 | TRANSFERS BETWEEN CATEGORIES | 31 | |
| 9.1.04 | THE FRAMEWORK FOR THE SYSTEM | 33 | |
| 9.1.05 | THE SELECTION OF SHIFTS BY EMPLOYEES | 34 | |
| 9.1.06 | SPECIAL PROVISIONS FOR PART-TIME BANQUET EMPLOYEES. | 35 | |
| 9.1.07 | SPECIAL PROVISIONS FOR ALL BANQUET EMPLOYEES | 36 | |
| 9.1.08 | SCHEDULING PROVISIONS WAIVED IN SPECIAL CASES | 36 | |

| ARTI | CLE SUBJECT | PAGE |
|-------|---|------|
| ARTI | CLE 10 - SENIORITY | 37 |
| 10.01 | SENIORITY ENTITLEMENT DEFINED | 37 |
| 10.02 | ELIGIBILITY FOR SENIORITY ENTITLEMENT | 38 |
| 10.03 | ACCRUAL OF SENIORITY | 38 |
| 10.04 | LOSS OF SENIORITY | 38 |
| 10.05 | SENIORITY LISTS | 39 |
| 10.06 | SENIORITY AND LEAVE OF ABSENCE | 39 |
| ARTI | CLE 11 - PROMOTIONS AND TRANSFERS AND LAY-OFF AND RECALL | 39 |
| 11.01 | PROMOTION | 39 |
| 11.02 | TRANSFERS | 40 |
| 11.03 | PROMOTION AND TRANSFER TRIAL PERIOD | 40 |
| 11.04 | DEMOTIONS AND SENIORITY | 42 |
| 11.05 | LAY-OFF AND RECALL PROCEDURE | 43 |
| ARTI | CLE 12 - WAGE ADMINISTRATION | 43 |
| 12.01 | WAGE RATES | 43 |
| 12.02 | SUBSTITUTION PAY | 44 |
| 12.03 | WAGE RATE CONDITIONS | 45 |
| 12.04 | NEW CLASSIFICATIONS AND WAGE RATES | 45 |
| 12.05 | PAYMENT OF WAGES UPON DISCHARGE, LAY-OFF OR RESIGNATION | 45 |
| 12.06 | SPECIAL PROVISIONS FOR ELECTION DAYS | 45 |
| 12.07 | BANQUET GRATUITY AGREEMENT | 46 |
| 12.08 | OVERNIGHT PREMIUM | 47 |

gvhea-local40 iv

| ARTI | ICLE SUBJECT | PAGE |
|-------|---|------|
| ARTI | CLE 13 - STATUTORY HOLIDAYS | 48 |
| 13.01 | RECOGNIZED STATUTORY HOLIDAYS | 48 |
| 13.02 | ELIGIBILITY FOR STATUTORY HOLIDAY PAY | 48 |
| 13.03 | EMPLOYEES WHO DO NOT WORK ON A STATUTORY HOLIDAY. | 48 |
| 13.04 | EMPLOYEES WHO PERFORM WORK ON A STATUTORY HOLIDA | Y49 |
| 13.05 | CALCULATION OF PAY FOR STATUTORY HOLIDAYS | 49 |
| 13.06 | STATUTORY HOLIDAY DURING EMPLOYEE'S VACATION | 50 |
| 13.07 | LOSS OF STATUTORY HOLIDAY PAY FOR FAILURE TO REPORT | 51 |
| 13.08 | ANNIVERSARY OF EMPLOYMENT | 52 |
| ARTI | CLE 14 - ANNUAL VACATIONS | 53 |
| 14.01 | ANNUAL VACATION PAY: EMPLOYEES WITH LESS THAN ONE YEAR OF SERVICE | 53 |
| 14.02 | ANNUAL VACATION AND VACATION PAY ENTITLEMENTS: EMPLOYEES WITH ONE YEAR OR MORE OF COMPLETED CONSECUTIVE SERVICE | 53 |
| 14.03 | DEFINITIONS | 54 |
| 14.04 | APPLICATIONS FOR ANNUAL VACATIONS | 54 |
| 14.05 | SCHEDULING OF ANNUAL VACATIONS | 54 |
| 14.06 | PREFERENCE FOR ANNUAL VACATIONS BY SENIORITY | 55 |
| 14.07 | SPLITTING OF ANNUAL VACATIONS | 56 |
| 14.08 | ANNUAL VACATIONS TO BE TAKEN WITHIN TIME LIMITS | 56 |
| 14.09 | POSTING OF ANNUAL VACATION SCHEDULES | 57 |

| ARTI | CLE SUBJECT | PAGE |
|-------|---|------|
| ARTI | CLE 15 - HEALTH CARE PLAN, PENSION PLAN AND WAGE LOSS PROTECTION | 57 |
| 15.01 | CONTRIBUTIONS FOR HEALTH CARE AND PENSION PLANS | 57 |
| 15.02 | CONTRIBUTIONS FOR PENSION | 59 |
| 15.03 | PENSION PLAN QUALIFICATIONS | 59 |
| 15.04 | DEEDS OF TRUST | 59 |
| 15.05 | PAYMENT OF CONTRIBUTIONS TO ADMINISTRATORS | 59 |
| 15.06 | MONTHLY STATEMENTS WITH CONTRIBUTIONS | 60 |
| 15.07 | PENALTIES FOR DEFAULT | 60 |
| 15.08 | INVESTIGATION OF THE EMPLOYER'S PAYROLL RECORDS | 60 |
| 15.09 | EMPLOYER CEASING TO BE A MEMBER OF THE GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION | 60 |
| 15.10 | EMPLOYER CEASING TO PARTICIPATE IN THE HEALTH CARE PLAN | 60 |
| 15.11 | TABLE OF CONTRIBUTIONS | 61 |
| 15.12 | WAGE LOSS PROTECTION | 61 |
| 15.13 | WORK INJURY PAY | 61 |
| 15.14 | CONTRIBUTIONS ON BEHALF OF EMPLOYEES IN RECEIPT OF WORKERS' COMPENSATION TIME-LOSS BENEFITS | S61 |
| 15.15 | POSSIBLE INTRODUCTION OF GOVERNMENT PROGRAMS | 62 |
| ARTI | CLE 16 - LEAVES OF ABSENCE | 62 |
| 16.01 | LEAVE OF ABSENCE: EMPLOYEE ELECTED TO UNION OFFICE | E62 |
| 16.02 | LEAVE OF ABSENCE: UNION CONVENTIONS AND EDUCATIONAL PROGRAMS | 62 |
| 16.03 | LEAVE TO APPEAR AS WITNESS | 63 |

gvhea-local40 Vi

| ARTI | CLE SUBJECT | PAGE |
|-------|---|-------------|
| 16.04 | BEREAVEMENT LEAVE | 63 |
| 16.05 | MATERNITY LEAVE | 64 |
| 16.06 | ADOPTION LEAVE | 65 |
| 16.07 | PATERNITY LEAVE | 65 |
| 16.08 | PAID PERSONAL DAYS | 65 |
| 16.09 | LEAVE FOR MILITARY SERVICE | 65 |
| 16.10 | LEAVE FOR JURY DUTY OR AS WITNESS FOR CROWN | 66 |
| 16.11 | GENERAL LIMITATION ON LEAVES OF ABSENCE | 66 |
| 16.12 | MULTIPLE APPLICATIONS FOR LEAVES OF ABSENCE | 66 |
| ARTI | CLE 17 – ROOM ATTENDANT WORKLOAD | 66 |
| 17.01 | WORKLOAD CONDITIONS | 66 |
| 17.02 | ROOM ATTENDANT WORKLOAD COMMITTEE | 67 |
| 17.03 | REST PERIODS | 67 |
| ARTI | CLE 18-MISCELLANEOUS EMPLOYEE ENTITLEMENTS | 68 |
| 18.01 | PROTECTED WORKING CONDITIONS | 68 |
| 18.02 | CAFETERIA, KITCHEN AND DINING LOUNGE MEALL ALLOWAN | ICES68 |
| 18.03 | EMPLOYEE ATTENDANCE AT STAFF MEETINGS | 68 |
| 18.04 | EMPLOYEES RETURNING TO WORK AFTER ILLNESS OR INJURY | ·69 |
| 18.05 | NO INDIVIDUAL CONTRACTS OR AGREEMENTS | 69 |
| 18.06 | HEALTH LAWS AND LOCKUP FACILITIES FOR EMPLOYEES | 71 |
| 18.07 | COMPENSATION TO EMPLOYEES RE: ENFORCEMENT OF HOUSE FOR PATRONS | |
| 18.08 | SEVERANCE PAY | 71 |
| 18 09 | SUCCESSOR OWNER ORLIGATIONS | 72 |

gvhea-local40 Vii

| ARTI | CLE SUBJECT | PAGE |
|-------|--|------|
| 18.10 | LIMITATION ON EMPLOYEE ENTITLEMENTS | 73 |
| 18.11 | INDUSTRIAL FIRST AID TRAINING OF EMPLOYEES | 73 |
| 18.12 | BAGGAGE HANDLING FEES | 73 |
| 18.13 | ROOM SERVICE GRATUITIES | 73 |
| ARTI | CLE 19 – EMPLOYEE CONDUCT AND DRESS | 73 |
| 19.01 | HOUSE RULES GOVERNING CONDUCT OF EMPLOYEES | 73 |
| 19.02 | CONTROL OF ABSENTEEISM | 74 |
| 19.03 | AUTHORITY RE: CHEQUES AND CREDIT | 75 |
| 19.04 | UNIFORM MODE OF DRESS | 75 |
| 19.05 | UNCONVENTIONAL MODE OF DRESS | 76 |
| 19.06 | SPECIAL UNIFORMS | 76 |
| 19.07 | TELEPHONE AND MAILING ACCESS | 76 |
| ARTI | CLE 20 – LIQUOR CONTROL LEGISLATION AND REGULATION | IS76 |
| 20.01 | EMPLOYEES SERVING LIQUOR | 76 |
| 20.02 | CHANGES IN REGULATIONS | 76 |
| ARTI | CLE 21 – DISCIPLINE AND DISCHARGE OF EMPLOYEES | 77 |
| 21.01 | DISCIPLINE AND DISCHARGE OF EMPLOYEES | 77 |
| 21.02 | PERSONNEL FILES. | 77 |
| 21.03 | WRITTEN RESPONSES TO WRITTEN DISCIPLINE | 77 |
| 21.04 | COMMUNICATION DURING GRIEVANCES | 77 |
| 21.05 | EMPLOYEES RIGHT TO REPRESENTATION | 77 |
| 21.06 | THE EXPUNGING OF CERTAIN DISCIPLINARY PENALTIES | 78 |
| ARTI | CLE 22 – GRIEVANCE AND ARBITRATION PROCEDURES | 78 |
| 22.01 | DEFINITION AND RECOGNITION OF A GRIEVANCE | 78 |

gvhea-local40 Viii

| ARTI | CLE SUBJECT | PAGE |
|-------|--|-----------------|
| 22.02 | GRIEVANCE PROCEDURE | 79 |
| 22.03 | ESTABLISHING THREE MEMBER BOARD | 80 |
| 22.04 | ARBITRATION HEARING AND AWARD | 81 |
| 22.05 | AUTHORITY OF THE ARBITRATION BOARD | 81 |
| 22.06 | OPTIONAL INVESTIGATION PROCEDURE | 81 |
| 22.07 | POWERS OF ARBITRATION BOARD | 83 |
| 22.08 | TIME LIMITS ARE DIRECTORY NOT DESCRIPTIVE | 83 |
| 22.09 | PERSONS AUTHORIZED TO DEAL WITH GRIEVANCES | 83 |
| 22.10 | TROUBLESHOOTER PROCEDURE | 84 |
| 22.11 | FAST TRACK MED/ARB PROCESS | 85 |
| ARTI | CLE 23 - DEFINITIONS AND JOB DESCRIPTIONS | 87 |
| 23.01 | OBJECTIVE INTERPRETATION | 87 |
| 23.02 | TIME SPAN REFERENCES | 87 |
| 23.03 | SPECIFIC DEFINITIONS | 87 |
| ARTI | CLE 24 - TECHNOLOGICAL CHANGE | 88 |
| 24.01 | TECHNOLOGICAL CHANGE | 88 |
| ARTI | CLE 25 - THE RIGHTS OF EMPLOYEES TO PRIVACY IN THE WORKPLACE | 89 |
| ARTI | CLE 26 – LETTERS OF UNDERSTANDING | 90 |
| 26.01 | | 90 |
| 26.02 | | 90 |
| SIGN | ATURE PAGE | 91 |
| APPE | ENDIX 1 - LIST OF DEPARTMENTS | 92 |
| APPE | ENDIX 2 - WAGE RATES - NOTE | 94 |
| | FOUR SEASONS HOTEL(Jul07/Jul08/Jan | 09/Jul09/Jan10) |

gvhea-local40 iX

| ARTICLE | SUBJECT | PAGE |
|--------------|---|--------------------------------------|
| HYATT | REGENCY VANCOUVER | (Jul07/Jul08/Jan09/Jul09/Jan10) |
| | SANCE VANCOUVER HOTEL URSIDE | (Jul07/Jul08/Jan09/Jul09/Jan10) |
| | STIN BAYSHORE, VANCOUVER | (Jul07/Jul08/Jan09/Jul09/Jan10) |
| APPENDIX 3 - | LIST OF SPECIFIC WORK WEE | KS95 |
| APPENDIX 4 - | MAXIMUM NUMBER OF REGU EMPLOYEES AT ANY ONE TIM | LAR BANQUET E96 |
| LETTERS OF | UNDERSTANDING | 98 |
| LOU #1 | RE: CONTRACTED SERVICES [A | rticle 2.04]98 |
| LOU #2 | RE: OPTIONAL INVESTIGATOR I [Article 21.06(d)] | PROCEDURE101 |
| LOU #3 | RE: HYATT REGENCY VANCOU | VER [Article 9]102 |
| LOU #4 | RE: BANQUET SERVER EXTRAS | [Article 4]103 |
| LOU #5 | RE: PROVISIONS FOR BREAKING | SENIORITY TIES105 |
| LOU #6 | RE: RETIREMENT POLICIES | 107 |
| LOU #7 | RE: HYATT REGENCY VANCOU' ITEMS ON THE EMPLOYER' | VER: SALE OF SUNDRY S PREMISES109 |
| LOU #8 | RE: RENAISSANCE VANCOUVER SALE OF SUNDRY ITEMS ON EMPLOYER'S PREMISES | |
| LOU #9 | RE: PERFORMANCE OF BARGAIN BY MANAGEMENT PERSON | NING UNIT WORK NEL114 |
| WAGE PAGES | | 110 |
| FOUR SEASON | IS VANCOUVER HOTEL | 116 |
| HYATT REGEN | NCY VANCOUVER HOTEL | 124 |
| RENAISSANCE | E VANCOUVER HOTEL HARBOU | RSIDE137 |
| THE WESTIN E | BAYSHORE, VANCOUVER | 147 |

ARTICLE 1 INTRODUCTION

1.01 PURPOSE OF THIS AGREEMENT

- (a) The purpose of this Agreement is to set forth and establish the terms and conditions of employment for those employees who come within the scope of this Agreement, so that stable and harmonious relationships may be established and maintained between the Employer and the Union, to the mutual benefit of the parties to this Agreement.
- (b) Further, the purpose of the Agreement is to facilitate the peaceful adjustment of all disputes and grievances in accordance with Article 21 of this Agreement, to prevent strikes, lockouts, waste, unnecessary expense, and avoidable delays in carrying out the most efficient and effective operations of the Employer's business and to enhance the living standards and working conditions of the employees.

1.02 IDENTITY OF THE "EMPLOYER"

For the purpose of this Agreement the Employer shall mean the present Employer signatories to this Agreement and/or any purchaser, lessee or transferee of the operation.

ARTICLE 2 DURATION AND INTEGRITY OF AGREEMENT

2.01 DURATION OF AGREEMENT

- (a) This Agreement shall be for the period from and including July 1, 2007 to and including June 30, 2010. Thereafter, the Agreement shall continue in full force and effect from year to year subject to the right of either party to serve notice to commence bargaining as provided for in the Labour Relations Code.
- (b) During the period when negotiations are being conducted between the parties for the renewal of this Agreement, the present Agreement shall continue in full force and effect until:
 - (i) the Union commences a legal strike; or
 - (ii) the Employer commences a legal lockout; or
 - (iii) the parties enter into a new or further Agreement.
- (c) During the continuation period provided for in (b) above, neither party shall attempt to take any action or make any changes in the terms and conditions of

employment, which would be inconsistent with the express terms of this Agreement.

(d) Notice to commence collective bargaining must be in written form and must be delivered either by registered mail, or personally delivered. In the event that such notice is personally delivered, a delivery receipt must be obtained.

2.02 EXCLUSION OF LABOUR RELATIONS CODE SECTIONS 50(2) AND 50(3)

The operation of Sections 50(2) and 50(3) of the Labour Relations Code is hereby excluded from, and does not apply to this Agreement.

2.03 WORK INTERRUPTION PROHIBITED

The Union agrees during the term of this Agreement there will be no slowdown or strike, stoppage of work or refusal to work or to continue to work. The Employer agrees that during the term of this Agreement there will be no lockout.

2.04 CONTRACTED SERVICES

Subject only to the provisions contained in the Letter of Understanding which is appended to, and which forms a part of this Agreement, the Employer agrees that all work coming under the jurisdiction of the Union, performed by anyone on behalf of, or at the instance of the Employer, directly or indirectly under contract or subcontract, shall be performed by employees who are members of the Union, or who shall become members in accordance with the terms and conditions as set out in the Agreement.

2.05 IMPACT OF LEGISLATION

- (a) The parties recognize and agree that they cannot be obligated or bound by any term, condition or provision, which would be contrary to any existing federal or provincial legislation or regulations passed pursuant thereto. In the event that any term, condition or provision, or part thereof, which is incorporated into this Agreement, whether by inadvertence, error or misunderstanding, is in fact or in law contrary to such federal or provincial legislation or regulation, then such term, condition or provision or part thereof, is void and of no effect.
- (b) In the event that existing federal or provincial legislation makes invalid any provision of this Agreement, the remaining provisions shall remain in effect for the term of the Agreement. The Employer and the Union shall confer to settle upon a mutually agreeable provision to be substituted for the provision(s) so altered or invalidated.

ARTICLE 3 UNION RECOGNITION AND RELATED

3.01 RECOGNITION OF EXCLUSIVE BARGAINING AGENT

- (a) The Employer recognizes the Union as the sole and exclusive bargaining agent for the employees in the bargaining unit described in the Certification issued under the Labour Laws of the Province of British Columbia, subject to the exclusions already recognized by the parties, or subsequently ordered by the Labour Relations Board or recognized by the parties.
- (b) For the purposes of this Agreement, the terms "employee" or "employees" shall be understood to mean those persons employed by the Employer for whom the Union is the recognized bargaining agent in (a) above.
- (c) The Union recognizes the Greater Vancouver Hotel Employers Association as the sole and exclusive agent for the employers named in the accreditation order issued by the Labour Relations Board.

3.02 REFUSAL TO WORK WITH NON-UNION EMPLOYEES AND RECOGNITION OF LEGAL PICKET LINES

- (a) Refusal on the part of Union members to work with non-Union employees, pertaining to the bargaining unit, shall not be deemed to be a breach of this Agreement. In the event that any employee or group of employees intends to exercise this right, the Employer must first be served with written notice in advance of the exercise of the right. The written notice must be provided by the Union Office.
- (b) No employee shall be required to cross a legal picket line arising from a strike or lockout. For the purposes of this Article 3.02, "legal picket line" shall mean only a picket line expressly permitted under Section 65 of the Labour Relations Code.
- (c) The Union agrees whenever practicable to give the Employer advance notice of the probable implementation of picket lines which might affect the Employer's operation.

3.03 PERFORMANCE OF BARGAINING UNIT WORK

No person whose regular job is not in the bargaining unit will work on any job for which rates are established by this Agreement, except for the purpose of instruction, experimentation, management training, in which case trainees shall not displace or

replace any employee in the aforesaid classifications except in cases of emergency when regular employees are not available.

3.04 HARASSMENT AND DISCRIMINATION

- (a) All employees have the right to work in an environment free from harassment, including sexual harassment, and discrimination.
- (b) "Harassment" means any unwelcome physical contact, comments, gestures, body language, posting or distribution of material, or other behaviour which has the purpose or effect of interfering with an employee's work performance or creating a hostile or offensive work environment.

"Sexual Harassment" includes any of the conduct described above which is of a sexual nature or which is directed at an employee on the basis of that employee's gender.

"Discrimination" means any conduct which is prohibited under the B.C. Human Rights Act, regulations and amendments made thereto, and shall include discrimination on the basis of an employee's membership in the Union or participation in its activities.

- (c) It is agreed between the parties that there is an obligation and desire to eliminate any and all sexual harassment in the workplace. This obligation applies equally to the Employer, the Union and all employees.
- (d) Complaints of harassment or discrimination will be thoroughly investigated. Alleged failure by any party to deal with a harassment or discrimination complaint may be the subject of a grievance pursuant to this Agreement.

3.05 UNION BUTTONS

The parties agree that all Union employees are entitled to wear a Union Button on duty, provided that the manner in which the button is worn shall not detract from the style of the uniform or costumes normally worn by the employee.

3.06 FAIR LABOUR SERVICES, PRODUCTS AND MATERIALS

The Employer undertakes, wherever possible and practical, to use services, products and other materials necessary to the proper functioning of the Hotel, which are manufactured, provided or produced under fair labour conditions.

3.07 UNION HOUSE CARD OR BAR CARD CONTRACT

The signing of a Union House or Bar Card Contract with the Union is not compulsory. Where an Employer does choose to sign such a contract, the Union shall provide the appropriate card, and the Employer shall display it in a location selected by the Employer. If the display location is not acceptable to the Union, the Union can cancel the House or Bar Card contract and require the Employer to return the card.

3.08 UNION INVESTIGATION OF THE STANDING OF EMPLOYEES' CONDITIONS

- (a) The Employer shall allow the properly authorized representative of the Union to investigate the standing of all employees' conditions, to see that this Agreement is being enforced. The Employer is entitled to require an individual to substantiate that he is an authorized representative of the Union.
- (b) When access is required for purposes of such investigation, the Union representative will notify the Employer in advance.
- (c) Access will not be unreasonably denied by the Employer.
- (d) The investigation must not result in any disruption with the Employer's operations or affairs, and it must not result in any employee or employees neglecting their work duties and responsibilities.
- (e) This right is restricted and limited to the investigation of all employees covered by the certification.

ARTICLE 4 UNION SECURITY

4.01 MEMBERSHIP

- (a) All employees who are now members of the Union or who may become members, shall remain members in good standing as a condition of employment.
- (b) All new employees shall be required to become members of the Union. The Union is entitled to determine the eligibility of newly hired employees for admission into membership in the Union, according to the Union's International Constitution, provided that the eligibility criteria and the manner of their administration are lawful in this Province.

(c) For the purpose of this Article, the term "good standing" is defined to refer only and be limited to the payment of Union membership dues and initiation fees.

4.02 NEW EMPLOYEES

- (a) The Employer agrees to acquaint each new employee with the fact that a collective agreement is in effect, and with the conditions of employment which are set out in this Article 4, and which deal with Union Security and the Check-off of Union Dues.
- (b) The Employer agrees that at each orientation meeting which it holds for new employees, it will provide the Shop Chairperson or, in his absence, a Shop Steward with the opportunity to address such new employees for the purpose of acquainting them with the obligations of an employee to the Union.
- (c) Where for the purposes of Paragraph (b), the Shop Chairperson or a Shop Steward attends an orientation meeting during his working hours, he shall suffer no loss of pay. He shall, however, not be paid overtime in the event such attendance were to extend beyond the end of his scheduled shift, and he shall not be paid for attendance at such meetings which are held outside his working hours.

4.03 CHECK-OFF: ASSIGNMENT OF WAGES

- (a) All employees as a condition of employment, shall sign an authorization of check-off before commencing work.
- (b) The Union agrees to supply the Employer with the necessary assignment of wages forms. Such forms must specifically authorize the deduction of initiation fees or reinstatement fees as the case might be, Union dues, fines, assessments and arrears, as required by Article 4.04.

4.04 CHECK-OFF: PROCESS AND PROCEDURES

- (a) The Employer agrees to deduct initiation fees or reinstatement fees as the case might be, Union dues, fines, assessments and arrears, upon receipt of the appropriate assignment of wages form, signed by each employee.
- (b) Upon commencement of employment, each new employee will be required to sign the appropriate assignment of wages form. In the event that the Employer's files do not contain the necessary assignment of wages for any existing employee, such employees shall, upon demand, sign and present the appropriate Assignment of Wages form.

- (c) All monies deducted from employees' earnings pursuant to this Article, are to be forwarded to the Secretary of the Union, together with a list of employees to whom the monies are to be credited, and the names, addresses and social insurance numbers of new employees hired, on or before the 15th day of the month in which the monies were deducted.
- (d) It is the responsibility of the Union to advise the Employer in writing as to the amount of money to be deducted for initiation fees, Union dues, fines, assessments and arrears, and of any changes in the amounts to be deducted. In the event that any amount to be deducted is changed from the amounts specified in the assignment of wages form signed by the employees, the Employer can require the employees to sign new forms reflecting the new amounts to be deducted, prior to making such deductions.
- (e) The Union recognizes and agrees that the Employer's obligation to deduct such dues is expressly restricted to making only such deductions as are permitted by law, and as are authorized by the valid assignment of wages form executed by each employee.
- (f) Provided that on or before the fifteenth day of February in any year the Union has furnished the Employer with income tax receipts for the immediately preceding calendar year's Union dues, the Employer will distribute such income tax receipts to all current employees.
- (g) In the case of the termination of employment for any reason of an employee who has worked for the Employer during the month in which the termination occurs, the Employer will deduct the appropriate amount of Union dues from such employee's final paycheque, and shall forward such amount to the Union in accordance with the provisions of Paragraph (c).
- (h) In the event that the Union alleges any violation by the Employer of this Article, notice of such alleged violation shall be given to the Employer in writing. If the matter is not resolved between the Employer and the Union, either party may then refer the issue directly to arbitration.

4.05 AN EMPLOYEE'S FAILURE TO MAINTAIN MEMBERSHIP IN GOOD STANDING

- (a) Upon notice in writing from the Union to the Employer that an employee:
 - (i) is not a member of the Union;

- (ii) has not signed a written assignment of wages to pay initiation fees;
- (iii) has revoked his/her written assignment of wages to pay initiation fees, Union dues or Union assessments;
- (iv) is suspended from the Union;
- (v) has been expelled from the Union; or
- (vi) has resigned from the Union;

the Employer shall immediately discontinue the employment of such employee.

(b) The Union shall indemnify the Employer and hold it blameless against any and all suits, claims, demands and liabilities that may arise for the purposes of complying with the provisions of this clause.

4.06 ADDITIONAL UNION ASSESSMENT

In addition to any other Union dues, fines, assessments and arrears which the Employer is required by this Article 4 to deduct from its employees, and to remit to the Union as a condition of their employment, the Employer shall deduct from each employee, and shall remit monthly to the Union four (4) cents for each hour of employment performed by the employee.

ARTICLE 5 SHOP STEWARDS AND SHOP CHAIRPERSON

5.01 SHOP STEWARDS

- (a) The Union is entitled to appoint or elect from among the employees a reasonable number of Shop Stewards, provided that no more than one (1) Shop Steward is appointed or elected from each department.
 - (b) The duties of a Shop Steward shall be to assist in the reporting and resolution of all grievances within the department or departments for which he is responsible.
 - (c) The Employer agrees to recognize as Shop Stewards those employees whose names the Union has first submitted in writing to the Employer as having been appointed or elected. The Union agrees to advise the Employer in writing of any changes made by appointment or election from time to time.

5.02 SHOP CHAIRPERSON

- (a) The Shop Stewards will elect from their ranks a person who will be the Shop Chairperson.
- (b) The Shop Chairperson will be responsible for the adjustment or resolution of grievances which progress beyond the first step of the grievance procedure.
- (c) The Shop Chairperson will be recognized by the Employer as the official spokesperson on behalf of the Union.

5.03 SHOP CHAIRPERSON AND OTHER SHOP STEWARDS

- (a) The first obligation of any Shop Chairperson or Shop Steward is the fulfillment of his responsibilities as an employee. Neither the Shop Chairperson nor any Shop Steward is entitled to engage in Union activities during his working hours, other than for such involvement as is necessary for the reporting and resolution of grievances.
- (b) Neither the Shop Chairperson nor any Shop Steward may leave his assigned work area for the purposes set out in this Article 5, without having obtained prior permission from his department head or, in the absence of his department head, from an alternate designated by his department head. Such permission will not be unreasonably withheld.
- (c) The necessary time which is spent by the Shop Chairperson and by the Shop Stewards during their regular working hours, for the reporting and resolution of grievances, or for attendance at meetings specifically provided for in this Article 5, shall be considered to be time worked for all purposes covered by this Agreement.
- (d) Under no circumstances shall the Shop Chairperson or any Shop Steward take any action or issue any instruction which will interfere with the operations or affairs of the Employer, or with the management or direction of the work force.
- (e) Neither the Shop Chairperson nor any Shop Steward shall be discriminated against or disciplined for the proper performance of his duties on behalf of the Union.
- (f) The Employer agrees that either the Shop Chairperson or a designated Union official may post official communications from the Union to its members on the designated Union bulletin boards within the hotel. Only materials which are consistent with the provisions of the Agreement may be included in a notice which is posted. Any such notice must be correctly dated at the time of posting and may be removed after it has been posted for two (2) weeks,

- unless a specific arrangement has been made between the Shop Chairperson or the designated Union official and the Employer for a particular notice to be posted for a longer period of time.
- (g) The Employer agrees that the Union may use bulletin boards on the Employer's premises which shall be for the exclusive use of the Union. The number, location and size of such bulletin boards shall be determined by mutual agreement of the Employer and the Union.

5.04 MANAGEMENT AND SHOP STEWARDS MEETING

- (a) Upon request a person or persons designated by the Employer and empowered to act on a subject, will meet with the Shop Stewards on a monthly basis, to review problems that may arise concerning the application and operation of the Collective Agreement.
- (b) All Stewards will be permitted to attend such meetings without loss of pay, but there must be no resulting overtime or other premium costs to the Employer.
- (c) Minutes shall be kept as a record of the matters discussed during these meetings.
- (d) At the discretion of the Employer, accommodation will be provided for such meetings at no cost to the Union.
- (e) Accommodation can only be denied or withdrawn because there are no facilities available, or are no longer available. In such cases, the meeting must be rescheduled at the earliest possible date.

5.05 SHOP STEWARDS MEETINGS

(a) Up to a maximum of four (4) times in any calendar year, on a quarterly basis, the Shop Stewards shall be entitled to meet on their own under the terms of this Article 5.05, in preparation for a meeting with the designated representatives of the Employer pursuant to the provisions of Article 5.04(a). In any case where the Shop Stewards have already prepared an agenda for a meeting with the Employer representatives, and have agreed upon the arrangements for such a meeting with the Employer, including a tentative agenda, the Shop Stewards shall be permitted to attend their own meeting for a maximum of two (2) hours, without loss of pay in accordance with the provisions of Article 5.04(b), and shall be entitled to accommodation provided by the Employer in accordance with the provisions of paragraphs (d) and (e) of Article 5.04.

(b) Business agents and other officials of the Union authorized by the Union President shall be entitled to attend any of the meetings of Shop Stewards contemplated by paragraph (a) of this Article 5.05.

ARTICLE 6 MANAGEMENT RIGHTS

6.01 MANAGEMENT RIGHTS

- (a) The entire management of the operation including discipline of the employees is vested exclusively in the Employer at his place of business.
- (b) In the exercise of management rights, the Employer will not treat any employee in an unfair or discriminatory manner and will observe the provisions of this Agreement at his place of business.

ARTICLE 7 HIRING PROCESS AND PROBATION PERIOD

7.01 HIRING PROCESS

Where the Union desires to institute a dispatch hall relative to the GVHEA hotels, it will so advise the Association. The following will then apply:

- (a) The Union will, each month, provide each GVHEA Employer with an "available members list", including classifications of members who are on lay-off from GVHEA hotels and who are registered with the Union dispatch hall. When a member registers with the dispatch hall, he will fill out a "profile sheet" which includes his employment related information, such as his qualifications and prior work experience.
- (b) Along with the available members list, the Union will send out copies of the profile sheets for the members whose names appear on the list.
- (c) When the Employer intends to hire to fill a vacancy or a newly created position, the Employer will first consider those members whose names appear on the current available members list. The Employer is entitled to choose a candidate which the Employer considers to be the best qualified and most suitable.
- (d) When the Employer hires a member whose name appears on the list, the Employer will immediately notify the dispatch hall so that the member's name can be struck from the list.

7.02 PROBATION PERIOD

Newly hired employees shall serve a probationary period of ninety (90) calendar days.

ARTICLE 8 EMPLOYEE TRAINING

8.01 PRIVATE AND GOVERNMENT SUBSIDIZED TRAINING PROGRAMS

- (a) In the event the Employer wishes to participate in any training program for the benefit of persons who will not be directly employed by the Employer, and who will not receive remuneration of any kind from the Employer, and where such program is subsidized by the Government of Canada, by the Government of British Columbia, or by some other private or public body, and whether such program is delivered directly by one of such bodies or indirectly by some other agency, the Employer shall provide the Union with advance notice of its intention, and shall advise the body or agency which will be delivering the training program, of the requirement that such body or agency secure the approval of the Union for the delivery of the program.
- (b) In any case where the Employer participates in any training program as contemplated by paragraph (a), the Employer agrees that it will not
 - (i) provide any remuneration to any trainee in the absence of the Union's consent;
 - (ii) displace or replace currently employed bargaining unit employees, cause any loss of hours for such employees, or interfere with the hiring of new employees into the bargaining unit, or the rehiring of laid off employees who have recall rights under the Agreement;
 - (iii) continue any such program in the event of a legal strike or lockout; or
 - (iv) commence or continue any such program in the event of receipt of notice of cancellation of the program from the Union.

ARTICLE 9 HOURS OF WORK AND RELATED

9.01 NORMAL STRAIGHT TIME HOURS OF WORK

(a) The normal straight time hours of work assigned by the Employer shall conform with the following guidelines:

- (i) subject to the provisions of Article 9.03(d), not more than eight (8) hours in any one (1) day;
- (ii) except in the case of an employee who has given prior written authorization on a Supplementary Employee Record of Scheduling Preferences, not more than five (5) working days in any seven (7) day period;
- (iii) not more than forty (40) hours in any five (5) working day period.
- (b) Any hours which the Employer requires an employee to work in excess of the above shall be paid at double the straight time hourly rate.

9.02 SPLIT SHIFTS

- (a) Where split shifts are assigned by the Employer, they must conform with the following guidelines:
 - (i) no shift less than seven (7) hours may be split;
 - (ii) no shift may be split more than once;
 - (iii) no part of a split shall be less than two and one-half (2-1/2) hours;
 - (iv) all split shifts must be worked within a twelve (12) hour period.
- (b) Split shifts will be allowed on the following basis:
 - (i) Only two (2) split shifts during an employee's work week are permitted in the following departments:
 - -Public House (Beer Parlour)
 - -Neighbourhood Pub
 - -Cocktail Lounge
 - -Desk (including Bellperson)
 - -Kitchen
 - (ii) Split shifts are allowed during an employee's work week in the following departments:
 - -Banquets
 - -Room Service
 - -Restaurants
 - -Dining Rooms
 - -Cabarets (Specialty Rooms, Discos, Show Lounges).

- (iii) Split shifts are not allowed in the following departments:
 - -Housekeeping
 - -Maintenance
- (c) The Employer is not obligated to pay split shift premium except where time off between split segments exceeds one (1) hour.
- (d) Where premiums are applicable, they will be as follows:
 - (i) eight (8) hours straight time pay for seven (7) hours worked;
 - (ii) nine (9) hours straight time pay for eight (8) hours worked.

9.03 SHIFT HOURS

All shifts assigned by the Employer must conform with the following guidelines:

- (a) four (4) hour shifts will be the minimum shift permitted in any one (1) day;
- (b) shifts of 5, 6, 7 or 8 hours may be assigned, subject to the provisions of Article 9.05;
- (c) all hours worked up to and including eight (8) hours in any one (1) day will be paid at the straight time rate.
- (d) shifts of more than eight (8) hours may be assigned as parts of a compressed work week, subject to the requirement for the prior written agreement of the Union President or by another Union official duly authorized by him in each specific case.
- (e) Employees will be provided eight (8) hours rest between scheduled shifts occurring over consecutive days and if not so provided, overtime will be payable on all hours worked on the second shift.

9.04 MAXIMIZING THE LENGTH OF SHIFTS

- (a) While the Employer is entitled to schedule shifts of various lengths as provided for in this Agreement, the Employer is obligated to first build and maintain shifts of 4, 5, 6, 7 or 8 hours as the only recognized shifts. Wherever possible, all 8 hour shifts will be scheduled before 7 hour shifts are scheduled, with a similar progression downward to 4 hour shifts.
- (b) Except as provided in Article 9.13, once an employee has been scheduled for a shift, he will be provided with work for the scheduled hours. This provision is not applicable in the case of banquet or catering department employees.

9.05 ASSIGNMENT OF SHIFTS BY SENIORITY

- (a) Within each departmental classification, the employees with the most seniority are entitled to the longest shifts.
- (b) Where an employee is scheduled for less than eight (8) hours in a day, the shift cannot be extended unless by consent of the employee.

9.06 DAYS OFF

- (a) Except in the case of an employee who has given prior written authorization on a Supplementary Employee Record of Scheduling Preferences, all service employees shall receive two (2) consecutive days off in each seven (7) days; and
- (b) All other employees shall receive two (2) days off in each seven (7) days, but the days off need not be consecutive.

9.07 TIME WORKED ON SIXTH AND SEVENTH CONSECUTIVE DAYS

Subject only to the provisions of Article 9.1.05 (c), double time shall be paid for all work performed on an employee's sixth and seventh consecutive days of employment.

9.08 PAYMENT FOR TIME IN LIEU OF BREAKS

- (a) Those employees who are assigned shifts which are unbroken by rest periods and/or meal breaks, shall be paid according to the following schedule:
 - 5 unbroken hours of work for 5-1/2 hours pay;
 - 6 unbroken hours of work for 6-1/2 hours pay;
 - 7 unbroken hours of work for 8 hours pay;
 - 8 unbroken hours of work for 9 hours pay.
- (b) No employee shall be entitled to grieve purported violations of this clause, except for the most recent period up to maximum of one (1) month.
- (c) No shift other than identified in this clause can be worked unbroken.

9.09 UNPAID MEAL BREAKS

Employees working shifts of 5 to 8 hours are entitled to an unpaid meal break between the 3rd and 5th hour of work. Such meal break shall not be less than one-half (1/2) hour on the employee's own time.

9.10 REST PERIODS

- (a) All employees are entitled to rest periods in accordance with the following schedule:
 - (i) 4 hours one ten (10) minute rest period;
 - (ii) 5 hours one ten (10) minute rest period;
 - (iii) 6 hours one ten (10) minute rest period;
 - (iv) 7 hours two (2) ten (10) minute rest periods;
 - (v) 8 hours two (2) ten (10) minute rest periods.
- (b) Such rest periods are part of the employee's assigned hours of work and the rest period time is paid for by the Employer.
- (c) Time to commence when the employee arrives at the assigned rest area or a total of fifteen (15) minutes from the employee's work station.

9.11 EMPLOYEE'S RESPONSIBILITY: WORK START TIME

- (a) Employees shall be in their respective assigned working locations, ready to commence work at their designated starting times, and they shall not leave their working locations at times or in a manner inconsistent with the terms of this Agreement.
- (b) It shall be the responsibility of each employee to familiarize himself with the work schedules posted in accordance with Article 9.12 (b), and to report for work in accordance with the information contained in such work schedules, or in accordance with any subsequent notice of change provided to him by the Employer pursuant to Article 9.13.
- (c) All employees will know the approximate time when the weekly or bi-weekly work schedules are posted in their respective departments. They will be expected, if at work, to review those schedules themselves.
- (d) For any employee who is not scheduled to work between the time when his own work schedule is posted and the time of the first shift assigned to him under the new work schedule, he will be expected to obtain the necessary information in any of the following ways:
 - (1) by requesting a co-worker to survey the new work schedule on his behalf, and to notify him of the shifts assigned to him on the new work schedule;
 - (2) by travelling personally to his place of work in order to survey the new work schedule; or

- (3) by speaking directly in person to his supervisor via telephone in order to request details of the first shift assigned to him on the new work schedule.
- (e) Those employees who choose to familiarize themselves with their new work schedules by telephoning their supervisors, will be expected to telephone during the normal working hours, but should make every effort to avoid making those calls at times of the day when the supervisor is likely to be particularly occupied in giving service to his/her customers.
- (f) Regardless, however, of which way an individual employee chooses to become familiar with his/her work schedule, and particularly in those cases where the employee chooses to rely upon a co-worker for the requisite information, each individual employee is deemed to retain the overall responsibility set forth in Article 9.11 of the Agreement.

9.12 WORK SCHEDULES

- (a) There shall be placed in a conspicuous place a work schedule specifying the names and classifications of each employee, days off of each employee and the starting and finishing time of each employee, and the Employer shall keep said schedule up to date.
- (b) In recognition of the provisions of Article 9.13(a), each work schedule referred to in Paragraph (a) of this Article 9.12, shall be posted at least forty-eight (48) hours in advance of the first shift contained on the schedule.
- (c) If any change is made to the work schedules which the Employer posts pursuant to Paragraph (b) of this Article, notice of such change shall be given directly to the affected employees by the Employer.
- (d) The Employer shall make every reasonable effort to introduce a system by which any changes that are made to a previously posted work schedule, shall be made as clearly and legibly as possible, shall be dated, and shall as far as possible indicate that the change has been made by a person authorized to do so.

9.13 CHANGES IN WORK SCHEDULES

- (a) In situations other than emergencies, the scheduled employees are entitled to forty-eight (48) hours notice of any change in their respective work schedules.
- (b) In emergency situations which are beyond the control of the Employer, as in the case of the failure of an employee to report for an assigned schedule, the

- Employer may give notice of less than forty-eight (48) hours, but not less than twenty-four (24) hours, when changing work schedules.
- (c) The notices referred to in Paragraphs (a) and (b) of this Article 9.13 shall be given directly to the affected employees by the Employer.
- (d) Employees who become aware that they are not going to be able to report for work as scheduled, are obligated to provide the Employer with notice at the earliest possible time, or to have someone else notify the Employer on their behalf, to allow the Employer time to cover the absence.
- (e) Employees who are unable to reach their own Manager will notify a Manager on Duty and, in addition to providing their name and department, will provide their absence reason. Voice mail or electronic mail are not acceptable means of reporting an absence.
- (f) Employees whose schedules are changed without the advance notice specified, cannot be disciplined if they advise that they cannot comply with the changed starting and finishing times for the first shift of the new schedule.
- (g) In situations where an employee has not been provided with notice of a change in his work schedule, and the employee reports as scheduled before the change, the employee shall be provided with work and/or pay as follows:
 - (i) two (2) hours' pay unless the employee is unfit to perform his duties or he has failed to comply with the Industrial Health and Safety Regulations of the Workers' Compensation Board, or
 - (ii) where the employee commences work, four (4) hours work and/or pay unless his work is suspended because of inclement weather or other reasons completely beyond the control of the Employer, in which case Paragraph (i) above applies.
- (h) In any emergency situation covered by Paragraph (b) on this Article 9.13 where an employee who was scheduled to work, fails to report for his shift, the Employer shall be required to offer the vacant shift to the most senior employee who
 - i) was not already scheduled to work on the day in question, and
 - ii) whose scheduled hours have not yet been maximized, and
 - iii) is able and willing to work the shift without the Employer having to incur an overtime or other penalty.

(i) In complying with these provisions, if the Employer is unable to reach an employee to inform them of an available shift, the Employer will continue to contact alternate employees in order of seniority and pursuant to the preceding terms.

9.14 SCHEDULING OF OVERTIME

In any case where no employee is available to work a shift or the extension of a shift at straight time, and the Employer thereupon determines that it is necessary to assign the work on an overtime basis, the following provisions will govern the assignment of the overtime.

- (a) Where the Employer's determination was made forty- eight (48) hours or more in advance of the start of the overtime shift,
 - (i) the overtime hours will be offered to the most senior employee within the classification and department;
 - (ii) if the most senior employee declines the offer, the overtime hours will be offered to other employees within the classification and department in the order of their seniority;
 - (iii) if no employee within the classification and department is willing to work the overtime hours, the Employer may assign them to any other available employee without further restriction; and
- (b) Where the Employer's determination was made less than forty-eight (48) hours in advance of the start of the overtime shift, the overtime hours will be assigned in accordance with the provisions of Paragraph (a) of this Article 9.14, but subject to the following provisions:
 - (i) The Employer shall not be required to offer the overtime hours to any employee who has already been assigned overtime work during the same week, until the overtime hours have been offered to, and refused by all other qualified employees who have not already been assigned overtime work during the same week; and
 - (ii) The Employer shall not be required to call in a senior employee to work the overtime if there is a junior employee already at work and otherwise eligible to work the overtime hours as an extension of his shift.

9.15 EXCHANGING SHIFTS

(a) Subject to the provisions of paragraphs (b) and (c), an employee, having reached agreement with another employee in his own classification, may in

writing apply for permission from his supervisor to exchange shifts with that other employee. The Employer reserves the right to approve or disapprove any such application, but providing (1) the application is submitted at least forty-eight (48) hours prior to the commencement of the shift which occurs first, (2) each of the said employees possesses the necessary qualifications to perform the other employee's job, as set out in Article 11.01 of this Agreement, and (3) the other employee is the most senior of the employees who is scheduled to work on the day in question and is willing to agree to the exchange, the Employer's approval will not be unreasonably withheld.

- (b) In any case where, as a consequence of an exchange of shifts, a junior employee works in a given day or week more hours than a senior employee within his classification, or works on a shift that a senior employee would have preferred to work, the senior employee will have no cause to grieve the matter.
- (c) An exchange of shifts will not be permitted if, as a consequence one or both of the employees would qualify for a premium payment of any kind.

ARTICLE 9.1 SCHEDULING OF SHIFTS

9.1.01 THE MAIN PRINCIPLES

The main principles and guidelines for the operation of the new system of seniority, are as follows.

It is the clear intention of the parties that those employees who are willing and able to commit themselves on a continuing basis to employment by their Employer, should be given priority of treatment in the matter of scheduling of shifts.

The Employers agree that they will make every reasonable effort to maintain a suitable number of competent employees in order to support the new system.

It is agreed that, subject only to the legitimate business interests of the Employer, senior employees should be given the first opportunity to select their days off and specific a.m., p.m. and midnight shifts from among available shifts.

The parties acknowledge that the nature of the hotel industry is characterized by wide fluctuations in business levels and activity, and they therefore agree that it is in the best interests of all the parties that the collective agreement provide the maximum possible degree of stability for the largest number of employees.

The parties also agree that it is in the overall interests of the entire bargaining unit that senior employees, once having selected their days off and a.m., p.m. and

midnight shifts, shall be bound by their choices for the four (4) month election period.

9.1.02 CATEGORIES OF EMPLOYEES

- (a) All employees will be designated as either Part-Time or Regular, depending solely on their availability.
- (b) Any employee who is prepared to make himself available without restriction, will qualify for Regular status, and any employee who is unwilling to do so, will be given Part-Time status.
- (c) An employee with Part-Time status, other than one employed in the Banquets Department, shall be required to register his availability for specific times of the day or days of the week.

9.1.03 TRANSFERS BETWEEN CATEGORIES

- (a) Any Part-Time employee, including one who was previously a Regular employee, may elect to transfer to Regular status, but, subject only to the provisions of paragraph (b) herein, may only do so during an election period, and will be given a seniority start date that coincides with the date of his transfer to Regular status.
- (b) Any Part-Time employee who, during the course of a period of four months, is the successful applicant in a competition for a full-time position, whether vacated by a Regular employee or newly established, may elect to transfer to Regular status at the time of his appointment, and if so electing, he will be given a seniority start date that coincides with the date of his transfer to Regular status.
- (c) On the second occasion when a Regular employee is unable to select a schedule of shifts or to accept a specific shift assigned to him, for any reason other than one considered to be legitimate under the provisions of Article 13.05(d) of this Agreement, he will be designated as Part-Time effective immediately.
- (d) A Part-Time employee who registers his availability for specific times of the day or days of the week, and who, during the course of any four (4)-month period, for other than a bona fide reason declines two (2) shifts that are assigned to him at a time when he had registered his availability, will have his employment terminated effective immediately.
- (e) A Part-Time employee who, without prior authorization, registers himself as completely unavailable for an entire four-month period, will have his employment terminated effective immediately.

- (f) Any Regular employee who decides to revert to Part-Time status, and who previously enjoyed Part-Time status within the same classification, will be given the seniority start date which he previously had on the Part-Time seniority list.
- (g) Any other Regular employee who decides to revert to Part-Time status, will transfer his seniority start date with him from his Regular seniority list to his new Part-Time seniority list.
- (h) Where the Employer has posted to hire for a position which is part-time, and where the candidate hired for that position registers his/her availability as unrestricted and so qualifies him/herself for Regular status, such status will not take effect until the next election period.
- (i) When a Regular employee requests a change to Part-Time status, such change will take effect on the first schedule following two full weeks from the date of request unless the Employer is able to accommodate the transfer to Part-Time at an earlier date. No penalty will be imposed on employees who require an immediate status change for bone fide emergency reasons.

9.1.04 THE FRAMEWORK FOR THE SYSTEM

- (a) Each calendar year will be divided into three segments, January to and including April; May to and including August; and September to and including December.
- (b) The first seven days of the months of December, April and August of every year, shall be designated as the election period for the immediately following period of four months.
- (c) Two weeks prior to the commencement of any election period, department heads will publish a schedule that will represent as closely as possible, the core hours that are expected to be available in that department during the following period of four months.
- (d) During the course of any election period, every employee who has not already done so, other than a Part-Time employee employed in the Banquets Department, shall be required to complete an Employee Record of Scheduling Preferences form (the "Preference form") on which shall be registered, in the case of each Regular employee, his preferences regarding weekly days off and a.m., p.m. and midnight shifts, and in the case of each

Part-Time employee, those days of the week and times of each day when he will be available.

- (e) At the time of hiring, each new employee, other than a Part-Time employee employed in the Banquets Department, shall be required to complete a Preference form on which, depending on his status as either a Regular employee or a Part-Time employee, shall be registered his preferences or availability in accordance with the provisions of paragraph (d) above.
- (f) The preferences registered on the Preference forms, shall for each Regular employee, be taken into account by the Employer throughout the ensuing period of four months, or, in the case of an employee first hired during the course of any such period, during the remaining balance of the period.
- (g) The availability registered on the Preference forms, shall for each Part-Time employee, other than one employed in the Banquets Department, be binding on him throughout the ensuing period of four months, and shall remain in force until replaced by another Preference form.
- (h) Following commencement of his initial four month period, any employee who fails to file a new Preference form during any election period, other than a Part-Time employee employed in the Banquets Department, shall be assumed to have registered the preferences he selected for the previous four month period.

9.1.05 THE SELECTION OF SHIFTS BY EMPLOYEES

- (a) Following publication of the representative schedules referred to in Article 9.1.04(c), each Regular employee shall be entitled in descending order of seniority, to select his own weekly schedule of five shifts, subject only to the requirement that the legitimate business needs of the Employer be observed.
- (b) In selecting his weekly schedules of shifts, and when accepting supplementary shifts during the course of any given work week, any service employee who has provided prior written authorization on a Supplementary Employee Record of Scheduling Preferences form (the "Supplementary form"), will be entitled to split his days off for the purpose of maximizing his hours in any given work week or in order to obtain his preferred days off, and in any such circumstances the Employer will not be required to pay overtime rates or incur any other penalty.
- (c) In selecting his weekly schedules of shifts, and when accepting supplementary shifts during the course of any given work week, any employee who has provided prior written authorization on a Supplementary form, will be entitled to select shifts which result in him working more than five consecutive days or more than five days in a seven day period, providing

that this is done only for the purpose of maximizing his hours. In any such circumstances the Employer will not be required to pay overtime rates or any other penalty. Each Supplementary form shall remain in force until replaced by another Supplementary form. No employee may in any event work more than five days within any work week as defined in paragraph (d) herein, unless he is paid at double his straight time hourly rate.

- (d) For the purposes of this Article 9.1 only, "work week" means a fixed sevenday period established for scheduling purposes by the Employer, which may not subsequently be altered without the consent of the Union. The specific work weeks in effect at the date of executing this Agreement are set out in the Appendix numbered 3 which is attached to, and which forms part of the Agreement, and may not be amended without the mutual consent of both parties.
- (e) Any Regular employee who has registered a preference to work less than full-time hours, shall be entitled, in descending order of seniority, to be excused from working his weekly schedule of five shifts in any specific work week, providing that there is an employee within the same department and classification available to work the shift or shifts in question at normal straight-time rates.
- (f) Having selected his preferred weekly schedule of shifts, a Regular employee will not be entitled to make changes in his weekly days off or in the a.m., p.m. or midnight shifts he selected, for the balance of the period of four months, unless those changes can be accomplished without causing disruption to any other employee.
- (g) Where, after having made his selection of a weekly schedule of shifts, a Regular employee discovers that one or more of his chosen shifts becomes unavailable, he shall be offered a substitute shift in place of any missing shift that in the first instance constitutes part of a Part-Time pool of shifts, or, if none exists, then one that constitutes part of the weekly schedule of shifts of the most junior Regular employee in his classification.

9.1.06 SPECIAL PROVISIONS FOR PART-TIME BANQUET EMPLOYEES

(a) During the last week of December, April and August in each year, and during each succeeding week, on or before a day and time in that week to be clearly stipulated in advance by the Banquet Manager, each Part-Time Banquet department employee shall be entitled to complete a Banquet Department Availability Form on which he may register those days of the immediately following week and those times of each of those days when he will be available.

- (b) The availability registered by a Part-Time Banquet department employee on his form, shall be binding on the employee for the week to which it applies.
- (c) A Part-Time Banquet department employee who, without prior authorization, fails to register himself as available on a realistic basis for at least one shift every two (2) weeks during any four-month period, may have his employment terminated effective immediately.
- (d) When the Banquet Manager schedules Part-Time Banquet department employees for work in any week, he shall be obliged to schedule first, in descending order of their seniority, those employees who registered their availability on the form first referred to in the paragraph (a) above.
- (e) If, after having scheduled to work all those employees who had previously registered their availability on the form first referred to in Article 9.1.06(a), the Banquet Manager requires additional employees to work, he shall offer the work to other employees in descending order of their seniority within their respective classifications, it being clearly understood and accepted that where he is unable to make immediate and direct contact with a senior employee, he will be entitled to proceed to offer the work to the next most senior employee.

9.1.07 SPECIAL PROVISIONS FOR ALL BANQUET EMPLOYEES

- (a) The maximum number of Banquet Department employees who may be granted Regular status in the several member hotels of the Association at any one time, shall be as set forth in the Appendix numbered 5 which is attached to, and which forms part of the Agreement.
- (b) Whenever a vacancy occurs among the Regular employees employed by the Employer in its Banquet department in a classification listed in Appendix 5, the Employer shall post a notice of the vacancy, and the vacancy shall be filled by the most senior of the Part-Time employees within the classification in question who apply.
- (c) In any case where there are fewer Part-Time employees who apply than there are vacancies, the Employer may fill the position from elsewhere in the Hotel or from outside the workforce.
- (d) If a Part-Time employee employed by the Employer in its Banquet department in a classification listed in Appendix 5, is interested in attaining Regular status, and takes an approved leave of absence, he will be responsible for registering his interest in a potential vacancy with the Employer, and where he has done so, the Employer will be obligated to consider his registration of interest as an application for any vacancy among

- the Regular employees in his classification that might occur during the employee's leave of absence.
- (e) Any Part-Time employee who attains Regular status pursuant to this Article 9.01.7, shall be placed at the bottom of the Regular seniority list for his classification with a date equivalent to the date of his transfer to that status.

9.1.08 SCHEDULING PROVISIONS WAIVED IN SPECIAL CASES

On the understanding that the Employer is not required to create any particular shift or shifts which would in any way disrupt the operation of his hotel, the General Manager and the Union President may agree that special grounds exist for permitting an employee to limit his/her availability to work, and the Employer may therefore assign such limited hours to an eligible employee at any time. The grounds that will be recognized for such special treatment are

- (i) where an employee has been absent for an extended period of time for medical reasons arising from either occupational or non-occupational disability, and where there exists a reputable medical recommendation that the employee be provided with limited hours;
- (ii) where an employee wishes to undertake an educational course related to his/her future in the hospitality industry;
- (iii) where an employee has decided to retire within the reasonably near future, and where the employee wishes to undergo a progressive reduction in the number of days or hours he/she works; or
- (iv) where it is considered that compassionate or other appropriate grounds exist.

It is clearly understood that any such arrangements will only be entered into if the particular General Manager determines that the arrangements are appropriate in the circumstances.

ARTICLE 10 SENIORITY

10.01 SENIORITY ENTITLEMENT DEFINED

(a) For the purposes of this Agreement, "classification seniority" shall be defined as an employee's total length of continuous service within his current classification within a particular department in the Employer's operation.

- (b) For the purposes of this Agreement, the term "department" shall be understood to mean those departments identified within this Agreement.
- (c) Classification seniority shall be used to determine the order of layoff and recall within a classification within a particular department.
- (d) In any case where a Part-Time employee believes that he is working a majority of his hours in a classification other than the one for which he was hired, or other than the one to which he was subsequently transferred, he may apply to the Employer to be transferred to such other classification, and if it can be established that he has been working more hours in another classification, he shall be transferred into the new classification, and shall be given a seniority start date that coincides with the date of his transfer into the new classification.
- (e) Annual vacation entitlement will be determined by the employee's total years of service in the hotel and the employee shall be granted holidays according to that established seniority.
- (f) For the purposes of this Agreement, "departmental seniority" shall be defined as an employee's total length of continuous service within a particular department.
- (g) In any instance where two or more regular employees are awarded the same start date for seniority purposes, the tie between them shall be broken in accordance with the provisions contained in the Letter of Understanding which is appended to, and which forms a part of this Agreement.

10.02 ELIGIBILITY FOR SENIORITY ENTITLEMENT

- (a) The Employer will maintain the practice of making every reasonable effort to apply the principles of seniority to temporary employees.
- (b) Regular employees shall possess seniority rights which are superior to those of Part-Time employees.

10.03 ACCRUAL OF SENIORITY

Each employee will be granted seniority rights which are related to the date when he

- (i) commenced his employment with the Employer,
- (ii) entered his current department, or
- (iii) entered his current classification, as the case might be.

10.04 LOSS OF SENIORITY

An employee will lose all his seniority rights where he:

- (a) voluntarily terminates his employment;
- (b) is discharged for just and reasonable cause;
- (c) is on lay-off more than six (6) consecutive months;
- (d) does not return to work on the date specified following an approved leave of absence other than medical; or
- (e) receives severance pay in accordance with the provisions of Articles 18.10, 18.11 or 23; or
- (f) is deemed to have abandoned his job pursuant to Article 19.02(h).
- (g) notwithstanding the foregoing, when an employee is laid-off because of the closure of an employer's facilities for renovations, the employee's seniority will be frozen for the duration of that layoff period, even if the layoff period is longer than six (6) consecutive months.
- (h) Part-Time employees who have not worked a shift for six months because of the availability registered on their Record of Scheduling Preferences form will lose their seniority. The Employer will provide advance notice in this regard to the employee after four months of inactive work.

10.05 SENIORITY LISTS

- (a) The Employer agrees to post updated departmental seniority lists within one (1) week of the conclusion of each election period, containing the following information:
 - (i) the employee's name;
 - (ii) the date from which the employee's service seniority is calculated;
 - (iii) the employee's job classification;
 - (iv) the date from which the employee's classification seniority is calculated: and
 - (v) for an employee who has transferred from Part-Time to Regular status within the same classification, his Part-Time seniority date.
- (b) The seniority lists shall remain posted until they are replaced after the conclusion of the immediately following election period.
- (c) Any objection to the accuracy of a posted seniority list must be lodged with the Employer prior to the last day of the month in which the election period falls. Thereafter, the posted lists will be deemed to be valid and correct for all purposes of this Agreement.
- (d) At the time of posting, a copy of each seniority lists shall be given to the Union.

10.06 SENIORITY AND LEAVE OF ABSENCE

No employee shall have the right to claim seniority if he or she has been on a leave of absence in excess of three (3) months, except as provided in Articles 16.01, 16.02, 16.05, 16.06, 16.07, 16.08 and 16.09.

ARTICLE 11 PROMOTIONS AND TRANSFERS AND LAY-OFF AND RECALL

11.01 PROMOTION

- (a) The Employer, when considering applicants for promotion, will apply seniority, provided however that the employee who claims the right to exercise his seniority for the purpose of such promotion possesses the primary qualifications of character, integrity, attitude, efficiency and ability to satisfactorily perform the full measure of the work required.
- (b) The G.V.H.E.A. Employers undertake, wherever feasible, and with as few exceptions as possible, to provide the maximum possible notice of bargaining unit position vacancies by the posting of notices.

It is clearly understood by both parties that the G.V.H.E.A. Employers in giving this undertaking will, from time to time, be faced with situations which make it impossible to provide adequate, or indeed, any notice to interested employees.

Wherever an Employer provides notice of a vacancy in accordance with this undertaking, that notice will identify the position by its classification and its department, the pay rate for the classification, an approximation of the available hours, where possible, and any other information which the Employer considers relevant or useful regarding the working conditions. All such notices shall state that the position is open to male and female employees.

11.02 TRANSFERS

- (a) Transfers offered by the Employer from one department to another will take place only with the consent of the employee.
- (b) Transfers from one department to another cannot take place unless there is a vacancy or a new position has been created, and no employee will be laid off because of such transfer.
- (c) In any case where an employee is transferred to another classification within the same department, he shall retain his existing employment seniority status

- and his existing departmental seniority start date, but shall receive a new classification seniority start date which shall coincide with the effective date of his transfer.
- (d) In any case where an employee is transferred to another department, he shall retain his existing seniority status, but shall receive new classification and departmental seniority start dates, both of which shall coincide with the effective date of his transfer.

11.03 PROMOTION AND TRANSFER TRIAL PERIOD

- (a) Any employee who is granted a promotion or transfer appointment by the Employer, shall be on a trial period for up to ninety (90) calendar days. During this trial period, the employee must demonstrate that he can satisfy the requirements of the work performance criteria for the job, to the satisfaction of the Employer.
- (b) Should the employee be unable to satisfy the requirements of the work performance criteria in the trial period, or should he decide during the trial period that he does not want to continue in the job, then the employee may be returned to his former job. In such cases, the Employer shall have the right to require all employees who changed job positions in consequence of the promotion or transfer, to move back into their job positions and wage rates, which they occupied prior to the promotion or transfer.
- (c) Notwithstanding the provisions of Paragraph (a) of this Article 11.03, in any case where the promotion or transfer appointment is made to a position that has been vacated on what is deemed to be a temporary basis, by an employee who has been granted a leave of absence pursuant to Article 16.05, 16.06, 16.07, 16.08 or 16.09 of this Agreement, the trial period of ninety (90) calendar days shall be extended to cover the entire period of the relevant leave of absence, and the right of the employee to be returned to his former job shall apply to any such extended period, together with all other provisions of Articles 11.03(a) and 11.03(b).
- (d) Any employee who is granted a promotion or transfer appointment by the Employer to a position outside the bargaining unit, shall be entitled to return to his former job within thirty (30) calendar days of such appointment, if he is unable to satisfy the requirements of his new job, or if he does not want to continue in the new job. In any such case, the Employer shall have the right to require all employees who changed job positions in consequence of the promotion or transfer, to move back into their job positions and wage rates which they occupied prior to the promotion or transfer, and in any such case the employee who is returned to his former job from outside the bargaining unit, shall be returned with all the rights and obligations which he possessed prior to his promotion or transfer.

- (e) Commencing on the first day of work in a position outside the bargaining unit, an employee promoted or transferred pursuant to paragraph (d) of this Article 11.03, shall relinquish all of his benefits, rights and obligations as a member of the bargaining unit other than the right to be returned to his former job during the thirty (30) calendar days immediately following his appointment, and the Union shall be neither entitled nor required to represent such employee for any purpose other than to enforce the right established by Paragraph (d) of this Article 11.03.
- (f) In any case where an employee who was granted a promotion or transfer appointment by the Employer to a position outside the bargaining unit, is returned to his former job within thirty (30) calendar days of such appointment,
 - (i) the Employer shall retain the responsibility for maintaining the employee's health and welfare coverage until such time as the employee has requalified for coverage under the Health Care Plan referred to in Article 15;
 - (ii) the Employer shall immediately resume making its pension contributions to the Pension Plan on behalf of the employee, in the amounts specified in Article 15 of this Agreement; and
 - (iii) the employee shall be required to pay to the Union such fees and dues as may be assessed in order to obtain reinstatement, including retroactive dues, where applicable.
- (g) In order to avoid any misunderstanding on the part of an employee who is granted a promotion or transfer appointment by the Employer to a position outside the bargaining unit, regarding entitlement to the severance pay provisions contained in Article 17 of this Agreement, it is agreed by the parties that
 - (i) if he is returned to his former job within thirty (30) calendar days of an appointment, severance pay to which he may be entitled upon his eventual termination, shall be calculated on the basis of his total length of service with the Employer, i.e. ignoring the period of service outside the bargaining unit; and
 - (ii) if he remains outside the bargaining unit at the end of the period of the said thirty (30) calendar days, any entitlement to severance pay which he would have possessed in the event his employment had been terminated while he was a member of the bargaining unit, will have been abandoned.

- (h) The entitlement established by Paragraph (d) of this Article 11.03, may not be exercised by any individual employee more frequently than once during his tenure as a bargaining unit employee.
 - (i) In any case where the Employer decides to promote or transfer an employee to a position outside the bargaining unit, the Employer shall notify the Union in writing in advance, stipulating the effective date of the appointment.

11.04 DEMOTIONS AND SENIORITY

- (a) When layoffs occur within any classification, the employee with the least seniority in the classification shall be the first employee to be laid off, it being understood that any such employee who is laid off without an expectation of being recalled within the ensuing period of sixty (60) days, shall have the right to be transferred within the same department, to any other classification in which he had previously held seniority rights, and in which the Employer considers him able satisfactorily to perform the full measure of the work required.
- (b) Any employee transferred to another classification pursuant to the provisions of Paragraph (a), shall have seniority within such other classification according to his length of service in the department, as measured by the date when he commenced his employment in the department.

11.05 LAY-OFF AND RECALL PROCEDURE

- (a) In the event of a lay-off, the order of lay-off within any affected classification shall be as follows: probationary employees in reverse order of their seniority; then Part-Time employees in reverse order of their seniority; then regular employees in reverse order of their seniority.
- (b) Employees who restrict their availability for hours of work or work schedules will not be protected by their seniority for recall.
- (c) Employees shall be recalled in the inverse order to that in which they were laid off.
- (d) An employee who has been laid off and wishes to be recalled must insure that the Employer has a current phone number and address for purposes of recall. Failure on the part of the employee to provide this information may result in the employee forfeiting his recall rights.
- (e) The Employer agrees that recall notification will be by direct contact (including personal contact and telephone contact), registered mail. Any employee failing to report for duty within sixty (60) hours, excluding

Saturday and Sunday from the time of such notification, shall be considered to have resigned without notice.

ARTICLE 12 WAGE ADMINISTRATION

12.01 WAGE RATES

- (a) Subject to Paragraphs (b) and (c) following, the minimum wage rates provided in the appendix numbered 2 which is attached to, and which forms part of the Agreement, shall cover the job descriptions and classifications of labour within the jurisdiction of Local 40 and shall remain in effect throughout the specified or extended term of this Agreement.
- (b) Newly hired employees will be paid in accordance with the following:
 - (i) For the first six (6) months of employment seventy-five (75%) percent of the applicable classification wage rate contained in the wage appendix.
 - (ii) For the second six (6) months of employment eighty-seven and one-half (87-1/2%) percent of the applicable classification wage rate contained in the wage appendix.
- (c) For purposes of the administration of this Article, any employee who is transferred or promoted within the service of the same hotel, or to another GVHEA hotel within the same corporation, will be given credit for their period of employment prior to the transfer or promotion and will not be considered to be a new hire.
- (d) No present employee will suffer a reduction in wages currently being paid, because of this provision, provided he remains employed at the hotel.
- (e) In the application of this Article 12.01, no employee shall be paid an hourly wage rate lower than that set by the Employment Standards Act of the Province of British Columbia, or pursuant to its Regulations.

12.02 SUBSTITUTION PAY

- (a) Subject to the provisions of Paragraphs (b) and (c) of this Article 12.02, all employees shall be entitled to receive for each hour of work the actual hourly wage rate which applies to the classification in which the work is performed.
- (b) When an employee employed in the Kitchen Department works twenty (20) hours or less in any one (1) week in any higher classification than that in

- which he is regularly employed, he shall be paid by the hour for the hours worked at the higher classification.
- (c) When an employee employed in the Kitchen Department works more than twenty (20) hours in any one (1) week in any higher classification than that in which he is regularly employed, he shall be paid at the rate of the highest classification for all hours worked in such a week.

12.03 WAGE RATE CONDITIONS

- (a) The wage rates outlined in the attached appendices are minimum wage rates and they do not prevent the Employer from paying a higher wage rate.
- (b) The Employer is not entitled to pay wage rates lower than those contained in the appendices, except as specifically provided for elsewhere in this Agreement.
- (c) In cases where the Employer has granted an employee a rate higher than that provided in the Agreement, the premium cannot be withdrawn unless it was granted on a conditional basis and the condition has been exhausted or withdrawn. Otherwise, such a premium can only be withdrawn at the time when the wage rates are being renegotiated.

12.04 NEW CLASSIFICATIONS AND WAGE RATES

It is agreed that job classifications and wage rates not specifically set out in the attached appendices of this Agreement shall be included in the schedule by mutual consent of both parties to this Agreement. If unable to agree, either party may invoke the grievance procedure as defined in this Agreement.

12.05 PAYMENT OF WAGES UPON DISCHARGE, LAY-OFF OR RESIGNATION

- (a) When an employee resigns, the Employer will pay all wages owing to the employee within six (6) calendar days of the date of his resignation.
- (b) When an employee is laid off or discharged, the Employer shall pay all wages owing to him within forty-eight (48) hours, exclusive of Saturdays, Sundays or holidays.
- (c) When an employee is laid off or discharged, the Employer will provide the employee with an EI "Record of Employment" form which will indicate the reason for the separation from employment.

12.06 SPECIAL PROVISIONS FOR ELECTION DAYS

- (a) Whenever the Employer is required by law to close any of its operations on an election day, those employees who would otherwise have been scheduled to work on that day, shall be paid their regular wages for the day, and may be required by the Employer to work during their regularly scheduled shift in the performance of tasks which are generally related to their normal duties.
- (b) Any employee who wishes to take advantage of his right to be free from work for the clear period established by applicable legislation, in order to vote in a Federal or Provincial general election, may do so by electing one of the following options at least forty- eight (48) hours in advance of the date when the work schedule for his department is normally posted.
 - (i) He may request his department head in writing to shorten his regular daily shift in order to increase the amount of his clear period to a maximum of the amount established by the applicable legislation. In such case the Employer will be required to comply with the employee's request, but will not be required to pay him for the time not worked.
 - (ii) He may request his department head in writing to reschedule his daily shift in order to maintain his regular number of hours. In such case the Employer may accommodate the request by modifying the starting time of his shift to the minimum extent required to produce the clear period required by the applicable legislation, or may decline the request. Where the Employer declines such a request, the employee will be permitted to arrive late or to leave early, as the case might be, and will be paid at his normal straight time rate for the time not worked.
- (c) The provisions of Paragraph (b) apply in their entirety to Municipal elections, in which case the clear period shall, in the absence of legislation to the contrary, be the same as for Provincial general elections.

12.07 BANQUET GRATUITY AGREEMENT

- (a) The current banquet gratuity and service charge levied on all clients of the four hotels Catering/Banquets departments for service rendered in connection with the food and beverage component of a function shall not be reduced during the term of the collective agreement.
- (b) The apportionment of the total amount of gratuities and service charges between the Employer and the bargaining unit employees will be as follows:

The Westin Bayshore, Vancouver 60% Employees 40% Employer

| Four Seasons Hotel | 60% Employees | 40% Employer |
|-----------------------------|---------------|--------------|
| Hyatt Regency Vancouver | 60% Employees | 40% Employer |
| Renaissance Hotel Vancouver | 60% Employees | 40% Employer |

With respect to the Westin Bayshore, the banquet captain service charges will be paid entirely by the employer.

- (c) The bargaining unit employees eligible to receive a portion of the pool of gratuities and service charges belonging to the bargaining unit and the point system used for distribution purposes will remain unchanged from what prevailed in each hotel on June 30, 2007.
- (d) If a client refuses to pay a pre-negotiated service charge in whole or in part, the appropriate portion as set forth in paragraph (b) will be deducted from the bargaining unit portion.
- (e) On each bi-weekly pay day, the Employer shall publish, with respect to the pay period, the total amount of Food and Beverage Revenues collected by function and the total amount of service charges collected by function on behalf of bargaining unit employees.
- (f) On each bi-weekly pay day, the Employer will publish a service charge allocation sheet showing the total amount of service charges collected on behalf of all bargaining unit employees, the name of each eligible employee, the total hours worked by him/her during the pay period, the total number of points earned, and the value of one point applicable to the pay period.
- (g) The accumulated total dollar amount of all the service charge portions earned by an employee during any pay period shall be indicated on his/her next biweekly pay cheque.
- (h) In the event an eligible bargaining unit employee or the Union claims that an error was made with respect to the distribution of the service charge, a meeting will be held as soon as possible between the Employee or Union and the Employer to attempt to resolve the matter. If the matter is not resolved, it may be the subject of a grievance.
- (i) A Banquet Gratuity Committee comprised of two representatives from the Employer and two representatives from the Union will be established at each hotel to review on a regular basis the administration of the arrangements described in this Article, including review of documentation relevant to the banquet and beverage gratuities.

12.08 OVERNIGHT PREMIUM

Employees will receive a premium of \$0.50 per hour for all hours worked during shifts which commence between the hours of 2200 and 0100.

ARTICLE 13 STATUTORY HOLIDAYS

13.01 RECOGNIZED STATUTORY HOLIDAYS

- (a) The following statutory holidays shall be recognized for the purposes of this Agreement:
 - New Year's Day; Good Friday; Victoria Day; Canada Day; B.C. Day; Labour Day, Thanksgiving Day; Remembrance Day; Christmas Day; and Boxing Day.
- (b) Canada Day will invariably be observed for the purposes of this Agreement on July 1st.

13.02 ELIGIBILITY FOR STATUTORY HOLIDAY PAY

- (a) An employee shall establish eligibility for statutory holiday pay for any statutory holiday other than one which occurs during a period of his annual vacation, by
 - (i) working on the statutory holiday; or
 - (ii) working on both his last regularly scheduled shift immediately preceding the holiday and his first regularly scheduled shift immediately following the holiday; or
 - (iii) working on either his last regularly scheduled shift immediately preceding the holiday or his first regularly scheduled shift immediately following the holiday, and being on a leave of absence approved by the Employer on the other of such two days.
- (b) In the case of an employee's absence from either his last regularly scheduled shift immediately preceding a statutory holiday or his first regularly scheduled shift immediately following such statutory holiday, because of sickness or accident, the Employer is entitled to require a doctor's certificate

as proof of such sickness or accident, and any abuse of this provision by an employee may be cause for discipline.

13.03 EMPLOYEES WHO DO NOT WORK ON A STATUTORY HOLIDAY

Each employee who is eligible to receive pay for a statutory holiday, and who does not perform work on the holiday, shall receive a normal day's pay for the holiday, calculated in accordance with the formula contained in Article 13.05.

13.04 EMPLOYEES WHO PERFORM WORK ON A STATUTORY HOLIDAY

Each employee who

- (a) is eligible to receive pay for a statutory holiday;
- (b) is scheduled by the Employer to work on such holiday and
- (c) does perform work on such holiday,

shall receive a normal day's pay for such holiday, calculated in accordance with the formula contained in Article 13.05, and, in addition to such normal day's pay, shall be paid one and one-half (1-1/2) times his normal straight time wage rate for the first eight (8) hours he works on the statutory holiday, and shall be paid two and one-half (2-1/2) times his normal straight time wage rate for any hours he works in excess of eight (8) hours.

13.05 CALCULATION OF PAY FOR STATUTORY HOLIDAYS

- (a) For the purposes of this Article 13, and subject to the provisions of Paragraphs (b), (c) and (d), a normal day's pay shall be calculated by multiplying
 - (i) an employee's normal straight time hourly wage rate, exclusive of all premium payments, by
 - (ii) the straight time hours he actually worked during the two (2) week period immediately preceding the week in which the statutory holiday occurs, divided by ten (10), and by calculating the resulting product to two places of decimals.
- (b) Whenever Christmas Day or Boxing Day occur during the two (2) week period immediately preceding the week in which Boxing Day or New Year's Day occur, then for the purpose of calculating a normal day's pay for Boxing Day or New Year's Day in the case of an employee who had performed work on the immediately preceding Christmas Day or Boxing Day, the

number of hours which shall be deemed to have been worked by him on Christmas Day or Boxing Day, shall be the greater of

- (i) the number of hours actually worked by him on the holiday, to a maximum of eight (8); or
- (ii) the number of hours credited to him for the holiday pursuant to the formula contained in Paragraph (a).
- (c) In any case where an employee who has established eligibility to receive pay for a statutory holiday pursuant to the provisions of Article 13.02, was, if a regular employee, scheduled to work for less hours on Christmas Day or Boxing Day than the length of his normal shift for the day on which the holiday occurs, or was absent from work during the two (2) week period immediately preceding the week in which such statutory holiday occurs, for any of the reasons specified in Paragraph (d), the calculation of a normal day's pay shall incorporate the number of straight time hours actually worked by the employee in the most recent two (2) week period worked by him/her in which he/she was not absent.
- (d) The reasons which are referred to in Paragraph (c), and which, when approved by the Employer, shall be considered to be legitimate absences for the purpose of determining the number of hours to be incorporated into the formula for calculating a normal day's pay, are: adoption leave; anniversary of employment holiday leave; annual vacation leave; bereavement leave; leave of absence pursuant to Article 16.02(a); leave of absence to appear as a witness pursuant to Article 16.03; leave of absence to serve on a jury or as a witness for the Crown; maternity leave; paternity leave; statutory holiday leave; receipt of weekly indemnity benefits pursuant to the GVHEA/Local 40 Health Care Plan; on leave of absence as the result of a bona fide injury which would have qualified for weekly indemnity benefits pursuant to the GVHEA/Local 40 Health Care Plan except for the existence of any waiting period or eligibility restrictions established by the Trustees of the said Plan; or receipt of time-loss benefits paid by the Workers' Compensation Board.
- (e) In any case where an employee who has established eligibility to receive pay for a statutory holiday pursuant to the provisions of Article 13.02, was absent from work during the two (2) week period immediately preceding the week in which such statutory holiday occurs, on a leave of absence approved pursuant to Article 16.02(a) of the Agreement, the Union shall, upon being billed by the Employer, reimburse to the Employer such portion of the employee's normal day's pay for such statutory holiday as results from the leave of absence.

13.06 STATUTORY HOLIDAY DURING EMPLOYEE'S VACATION

- (a) In the case of a statutory holiday which occurs during a period of his annual vacation, an employee, if eligible to receive pay for the statutory holiday, will receive a normal day's pay for such holiday calculated in accordance with the formula contained in Article 13.05.
- (b) For the purposes of this Article 13.06, an employee shall establish eligibility or statutory holiday pay by
 - (i) working on both his last regularly scheduled shift immediately preceding a period of his annual vacation and his first regularly scheduled shift immediately following a period of his annual vacation; or
 - (ii) working on either his last regularly scheduled shift immediately preceding a period of his annual vacation or his first regularly scheduled shift immediately following a period of his annual vacation, and being on a leave of absence approved by the Employer on the other of such two days.
- (c) In the case of an employee's absence from either his last regularly scheduled shift immediately preceding a period of his annual vacation or his first regularly scheduled shift immediately following such period of his annual vacation, because of sickness or accident, the Employer is entitled to require a doctor's certificate as proof of such sickness or accident, and any abuse of this provision by an employee may be cause for discipline.
- (d) In the case of an employee who, having received a normal day's pay, wishes to be granted a day off without further pay in recognition of a statutory holiday which occurs during a period of his annual vacation, he shall provide reasonable advance notice in writing, indicating whether he wishes to be granted the working day immediately preceding the start of such annual vacation period, the working day immediately following the end of such annual vacation period, or some other day within the thirty (30) calendar day period immediately following the statutory holiday. The Employer will make every reasonable effort to accommodate such a request, and nothing contained herein shall prevent the Employer and the employee from extending the aforementioned thirty (30) calendar day period, if they mutually agree to do so.

13.07 LOSS OF STATUTORY HOLIDAY PAY FOR FAILURE TO REPORT

(a) If an employee is scheduled to work on a statutory holiday, but fails to report for work on such holiday, the Employer is entitled to require such employee to show that there was reasonable cause for his absence.

(b) If such employee has not obtained the Employer's approval to be absent, or is unable to show that there was reasonable cause for his failure to report for work on the statutory holiday as scheduled, he shall forfeit whatever pay he would have received for such holiday, and in addition such failure may be cause for discipline.

13.08 ANNIVERSARY OF EMPLOYMENT

- (a) In the event of the proclamation of Heritage Day by the Government of Canada, or in the event of the proclamation of any other holiday by either the Government of Canada or the Government of British Columbia which occurs prior to the proclamation of Heritage Day, the new holiday will be added to the list of statutory holidays recognized pursuant to Article 13.01, and this Article 13.08 will cease to have any further effect. In the meantime, the provisions of Paragraphs (b), (c), (d) and (e) shall remain in effect.
- (b) An employee shall establish his eligibility for a holiday in recognition of his anniversary of employment, in accordance with the same requirements which are established by Articles 13.02 and 13.06 for the purpose of statutory holiday eligibility.
- (c) An employee who has established his eligibility for an anniversary of employment holiday in accordance with the provisions of Paragraph (b), shall be entitled to a payment in recognition of his anniversary of employment, such payment to amount to a normal day's pay calculated in accordance with the provisions of Article 13.05.
- (d) In the case of an employee other than one whose anniversary of employment occurs during a period of his annual vacation, and who, having received a normal day's pay, wishes to be granted a day off without further pay in recognition of his anniversary of employment holiday, he shall provide reasonable advance notice in writing, indicating whether he wishes to be granted a day off on his actual anniversary or on some other day within the thirty (30) calendar day period immediately following such anniversary. The Employer will make every reasonable effort to accommodate such a request, and nothing contained herein shall prevent the Employer and the employee from extending the aforementioned thirty (30) calendar day period, if they mutually agree to do so.
- (e) In the case of an employee whose anniversary of employment occurs during a period of his annual vacation, who is eligible to receive a normal day's pay, and who wishes to be granted a day off without further pay in recognition of his anniversary of employment, the provisions of Article 13.06(d) shall apply

in the same manner as they apply to a statutory holiday which occurs during a period of an employee's annual vacation.

ARTICLE 14 ANNUAL VACATIONS

14.01 ANNUAL VACATION PAY: EMPLOYEES WITH LESS THAN ONE YEAR OF SERVICE

Employees with less than one year of completed service, will receive annual vacation pay in accordance with the provisions of applicable legislation.

14.02 ANNUAL VACATION AND VACATION PAY ENTITLEMENTS: EMPLOYEES WITH ONE YEAR OR MORE OF COMPLETED CONSECUTIVE SERVICE

(a) Employees are entitled to annual vacation and annual vacation pay, according to their completed years of consecutive service, calculated from their date of hire, as follows:

| Completed Years of Service | Annual Vacation Time | Annual Vacation Pay | |
|------------------------------------|----------------------|------------------------|--|
| 1 year, but less than 3 years | 2 weeks | 4% | |
| 3 years, but less than 7 years | 3 weeks | 6% | |
| 7 years, but less than 17 years | 4 weeks | 8% | |
| 17 years, but less than 23 | 5 weeks | 10% | |
| 23 years or more | 6 weeks | 12% | |

(b) "Consecutive years" as used herein, shall be understood to mean consecutive years of service with the same establishment subject to Articles 10.03 and 14.04 of this Agreement.

- (c) Annual vacation pay shall be calculated using the applicable percentage from (a) above, as a percentage of the employee's gross earnings for the preceding anniversary year.
- (d) "Gross earnings" as used herein, shall be understood to mean the total earnings realized by an employee from the payment of wage rates for straight time, overtime, statutory holiday pay and annual vacation pay.
- (e) Any employee will be entitled to receive his annual vacation pay on the last pay day immediately preceding his annual vacation, providing he files an application in writing at least fourteen (14) calendar days in advance of such pay day.

14.03 DEFINITIONS

For the purposes of this Article 14,

- (a) "Annual vacation year" means the period commencing on April 1st one year, and running to and including the 31st day of March in the immediately succeeding year.
- (b) "Calendar year" means the period commencing on January 1st of one year, and running to and including December 31st of the same year.

14.04 APPLICATIONS FOR ANNUAL VACATIONS

- (a) All applications for annual vacation shall be filed with the Employer in writing. Upon receipt of any such written application the Employer shall respond in writing, indicating whether or not the application has been granted.
- (b) Employer's response shall be given
 - (i) in the case of any written application received prior to the commencement of an annual vacation year, not later than March 15th of the same calendar year; and
 - (ii) in the case of any written application received after the commencement of an annual vacation year, within fourteen (14) calendar days.

14.05 SCHEDULING OF ANNUAL VACATIONS

(a) One employee within each department shall be entitled to be absent on annual vacation at any time during each annual vacation year. Additional employees shall be entitled to be absent on annual vacation at any time during each annual vacation year, providing they have filed written

applications by the last day of February for annual vacation to be taken in the immediately following annual vacation year, and providing that the total number of employees absent on annual vacation from any department at any time

may never exceed five per cent (5%) without the Employer's consent.

- (b) Nothing in this Article 14.05 shall preclude the Employer from permitting more than five per cent (5%) of the employees in any department to be absent on annual vacation at any time.
- (c) For the purposes of determining the number of employees who are entitled to be absent on annual vacation from any department at any time, five per cent (5%) shall be applied to the amount which results from dividing the total number of hours actually worked by all employees in that department during the immediately preceding calendar year, by eighteen hundred and twenty (1820). The final result shall be calculated to two decimal places, and shall be rounded to the nearest whole number. For example, 2.49 shall become 2, and 2.50 shall become 3.
- (d) Except for those vacations selected by employees pursuant to Paragraph (a), all annual vacations will be scheduled on the basis of the mutual consent of the Employer and the employee.
- (e) Notwithstanding the foregoing provisions of this Article 14.05, in any case where an employee applies for his annual vacation after the last day of February, his application may be denied by the Employer if the number of employees already absent, and reasonably expected to continue being absent from his department during the time covered by his application, through a combination of annual vacation, weekly indemnity or workers' compensation leaves, exceeds the limits set out in Paragraph (a) of this Article.
- (f) In any case where the Employer has approved an application for annual vacation, that approval may not be withdrawn without the employee's consent.

14.06 PREFERENCE FOR ANNUAL VACATIONS BY SENIORITY

- (a) Providing he files his application for annual vacation in writing by the last day of February for the immediately following annual vacation year, and subject to the 5% limit contained in Article 14.05(a), a senior employee shall have preference within his department over all junior employees with respect to the selection of his annual vacations.
- (b) Subject only to the provisions of Paragraph (c), a senior employee shall not be entitled to have any preference over junior employees with respect to any

- application filed after the last day of February for the immediately following annual vacation year.
- (c) With respect to all written applications filed after the last day of February for the immediately following annual vacation year, preference will be given to those which are first received, provided that where more than one such application is received on the same day, preference will be given the senior employee.
- (d) For the purposes of this Article 14.06, Regular employees shall be considered to be senior to Part-Time employees, regardless of their relative length of service within the hotel or within their department or classification.

14.07 SPLITTING OF ANNUAL VACATIONS

- (a) All employees are entitled to split their annual vacation entitlement into as many segments as they have weeks of entitlement in any year, providing that each segment shall consist of one or more full weeks.
- (b) Notwithstanding the provisions of Paragraph (a), by mutual consent of the Employer and an employee, the employee may be granted one or more segments of his annual vacation entitlement in blocks of less than one full week.
- (c) Whenever an employee takes only a part of his annual vacation entitlement, he shall be paid a strictly prorated portion of his accumulated vacation pay for the annual vacation period in question.

14.08 ANNUAL VACATIONS TO BE TAKEN WITHIN TIME LIMITS

- (a) Subject to the provisions of Paragraph (b), an employee must take the annual vacation to which he is entitled, not later than twelve (12) months after the anniversary date upon which he became entitled to it.
- (b) An employee may not defer any of his annual vacation entitlement beyond the limit of twelve (12) months established in Paragraph (a), except with the express written consent of the Employer. In order to be considered for a deferral, the employee must apply in writing prior to the expiry of the said twelve (12) months, and must provide valid reasons for seeking an exception to the general rule that annual vacations should be taken within the time limit established herein.
- (c) The Employer shall respond in writing to any such application within fourteen (14) calendar days, and shall not unreasonably withhold consent.
- (d) In any case where

- (i) the employee has not taken all of his annual vacation entitlement prior to the expiry of the time limit established in Paragraph (a); and
- (ii) the Employer has not consented in writing to permit the deferral of the outstanding portion of his entitlement,

the employee shall be required to take such outstanding annual vacation entitlement within the twelve (12) months immediately following the expiry of the time limit established by Paragraph (a), and at a time to be determined by the Employer. Notwithstanding its right to determine the time at which such non-conforming annual vacation entitlement is to be taken by an employee, the Employer will endeavour to comply with the employee's preference.

- (e) In any case where the Employer has consented to the deferral of an employee's vacation, such employee may exercise seniority preference over junior employees within his department, with respect to the deferred vacation entitlement.
- (f) In any case where the Union notifies the Employer in writing that the senior employees within a department have agreed to permit a junior employee to select his annual vacations ahead of them, the Employer will honour such agreement, and will permit the junior employee to select the combination of current and deferred annual vacations at a time which, subject to the restrictions imposed by this Article 14, is of his choosing.

14.09 POSTING OF ANNUAL VACATION SCHEDULES

Not later than March 20th in each year, the Employer shall post in each department an annual vacation schedule which shall set forth the periods of annual vacation entitlement which have been selected by those employees who had filed written applications by the last day of the immediately preceding February, and whose applications had been approved by the Employer.

ARTICLE 15 HEALTH CARE PLAN, PENSION PLAN AND WAGE LOSS PROTECTION

15.01 CONTRIBUTIONS FOR HEALTH CARE AND PENSION PLANS

(a) Effective November 1, 2006, the total amount of the Employer's contribution for the Health Care Plan will be one dollar and fourty-eight cents (\$1.48) for each hour of employment performed by an employee covered by the Agreement.

- (b) The total amount of each employee's contribution for fringe benefits will be nine cents (9ϕ) for each hour of employment performed by him/her.
- (c) Each year after 2003, the Employers' contribution to the Health Care Plan will be established at an hourly contribution which, in the Plan Administrator's view, will ensure that the benefit levels as of January 2003 are maintained and an adequate Fund Reserve is in place. The Dental Fee Schedule each year will be the current schedule for that year.
- (d) If, during the term of the Collective Agreement, the benefit levels under any of the Health Care Plans under the control of the Administrator are improved beyond the levels applicable to employees in this bargaining unit, the Employers and the Union will meet to discuss what changes should be made to the benefit levels applicable to employees in this bargaining unit. Should they be unable to reach agreement on this matter, it will be referred to Brian Foley for a binding decision.
- (e) The current Employer contribution per hour will be maintained. If, during the term of this collective agreement, the Plan Administrator deems that an increase or decrease is appropriate in the Employer contribution while still maintaining the benefit levels, that increase or decrease will be implemented.
- (f) GVHEA (HIR) and the Union will recommend to the Health and Welfare Plan Trustees that the following changes be made to the medical plan:
 - Effective from January 1, 2008, an employee will be eligible for the Group 1 Health Care Plan if the employee has worked at least 324 hours in the three consecutive calendar months immediately before the date of a claim.
 - Effective from January 1, 2008, the Vision Care Plan will provide as follows; one set of new lenses, including contact lenses selected in place of spectacle lenses, not more frequently than once in 24 months, or in 12 months in the case of individuals under 19 years of age, to the following limits:

| Single vision lenses or laser eye surgery | \$200.00 |
|---|-------------------|
| Bifocal lenses | \$240.00 |
| Trifocal lenses | \$280.00 |
| Contact lenses or laser eye surgery | \$500.00 lifetime |

• Effective January 1, 2008, an employee and the employee's dependants will qualify for Dental Care benefits once the employee has been eligible for coverage in the GVHEA – Local 40 Health Care Plan for four consecutive months.

- Effective with ratification, continue MSP coverage for employees on long term illness.
- As soon as can be arranged by the Plan Administrator, the Health and Welfare Plan will be amended to include a Prescription Drug Card.

15.02 CONTRIBUTIONS FOR PENSION

- (a) Effective October 1, 2007, the Employers contribution to the Pension Plan will be fifty-seven cents (57ϕ) for each hour of employment performed by an employee covered by the Agreement.
- (b) Effective July 1, 2008, the Employers contribution to the Pension Plan will be sixty-three cents (63ϕ) for each hour of employment performed by an employee covered by the Agreement.
- (c) Effective July 1, 2009, the Employers contribution to the Pension Plan will be sixty-nine cents (69¢) for each hour of employment performed by an employee covered by the Agreement.

15.03 PENSION PLAN QUALIFICATIONS

The Employer agrees that all employees who qualify for benefits under the provisions of the Trust Agreement known as the Hotel, Restaurant and Culinary Employees and Bartenders Union, Local 40 Pension Plan, shall be covered by the pension fund as set out in the said Trust Agreement.

15.04 DEEDS OF TRUST

It is mutually agreed between the Union and the Employer that all terms and conditions of the Trust Deed between the Union and the Greater Vancouver Hotel Employers Association in regards to Health Care, and all terms and conditions of the Deed of Trust between the Union and the B.C. Hotels Association in regards to Pension, shall be binding on the signing parties. This shall at no time determine the hourly rates of contribution as defined within Article 15 of the Agreement.

15.05 PAYMENT OF CONTRIBUTIONS TO ADMINISTRATORS

The Employer agrees to forward all monies payable by him in respect of fringe benefits, on or before the 10th day of the month following the actual performance of work, and shall forward said contributions to the respective Administrators.

15.06 MONTHLY STATEMENTS WITH CONTRIBUTIONS

The Employer also agrees to remit monthly statements setting out the names of the employees in respect of whom the monthly contributions are made, together with their Social Insurance Numbers and the number of hours worked by them.

15.07 PENALTIES FOR DEFAULT

- (a) In the event the Employer fails to remit contributions to the Pension Plan in conformity with the provisions of Article 15.05, the Employer shall, if in default more than ten (10) days after notification by the Union, pay the monies due thereunder and in addition thereto pay the Pension Plan a penalty in the amount of \$50.00. The Employer shall also be responsible for loss of benefits to any employee because of the Employer's default action.
- (b) Payments of contributions to the Health Care Plan under Article 15.05 which are not paid on the due date, will bear interest from the date the contribution was due until it is paid in full, at the prime rate per annum charged by the bank appointed from time to time by the Trustees under the Trust Deed between the Association and the Union to its most credit worthy customers at the date such contribution was due plus one percentage point. The Employer shall also be responsible for loss of benefits to any employee because of the Employer's default action.

15.08 INVESTIGATION OF THE EMPLOYER'S PAYROLL RECORDS

Upon receipt of a reasonable period of advance notice, the Employer shall allow the properly authorized representative of the Union, the Plan Administrators or such other properly authorized representatives of the Board of Trustees to investigate all of the Employer's relevant records for the purpose of ensuring that the proper contributions have been remitted pursuant to the provisions of this Article 15.

15.09 EMPLOYER CEASING TO BE A MEMBER OF THE GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION

In the event the Employer ceases to be a member of the Greater Vancouver Hotel Employers Association during the term of the Agreement, the parties shall continue to be bound by all of the obligations established by this Article 15, until such date as they enter into the immediately following renewal Collective Agreement.

15.10 EMPLOYER CEASING TO PARTICIPATE IN THE HEALTH CARE PLAN

In the event it ceases to participate in the Greater Vancouver Hotel Employers Association-Local 40 Health Care Plan, the Employer shall remain responsible for any and all residual liabilities associated with, or arising from its period of participation in the Plan.

15.11 TABLE OF CONTRIBUTIONS

| DESCRIPTION | NOV. 1 2006 | OCT. 1 2007 | JAN. 1 2008 | JAN. 1 2009 |
|---------------------------|----------------|----------------|----------------|----------------|
| Health & Welfare Employer | \$1.48* | \$1.48* | \$1.48* | \$1.48* |
| Health & Welfare Employee | \$0.09 | \$0.09 | \$0.09 | \$0.09 |
| Pension | \$0.51 | \$0.57 | \$0.63 | \$0.69 |
| | | | | |

^{*} Subject to review by Plan Administrator on January 1, 2008 and January 1, 2009

15.12 WAGE LOSS PROTECTION

After two (2) years employment, an employee who is eligible for weekly indemnity under the Greater Vancouver Hotel Employers Association - Local 40 Health Care Plan, shall be reimbursed for the first three (3) days of wage loss, once in each calendar year.

15.13 WORK INJURY PAY

An employee who is injured on the job during his working hours, and who as a consequence

- (a) is required to leave his place of work in order to obtain treatment; and
- (b) receives neither weekly indemnity benefits pursuant to the GVHEA/Local 40 Health Care Plan nor time-loss benefits from the Workers' Compensation Board for the balance of the shift during which he was injured;

shall be paid by the Employer at his straight time rate for all of the hours he was scheduled to work on the day on which he was so injured.

15.14 CONTRIBUTIONS ON BEHALF OF EMPLOYEES IN RECEIPT OF WORKERS' COMPENSATION TIME-LOSS BENEFITS

In the case of an employee who is in receipt of time-loss benefits from the Workers' Compensation Board, the Employer shall remit contributions to the Health Care Plan and to the Local 40 Pension Plan pursuant to the provisions of subsections (a) and (c) of Article 15.02, on the basis of the hours the employee would have been working other than for his temporary disability.

15.15 POSSIBLE INTRODUCTION OF GOVERNMENT PROGRAMS

The provisions of Articles 15.01 and 15.02 of this Agreement shall be subject to the provisions of the Letter of Understanding which has the same heading, and which forms a part of this Agreement.

ARTICLE 16 LEAVES OF ABSENCE

16.01 LEAVE OF ABSENCE: EMPLOYEE ELECTED TO UNION OFFICE

- (a) The Employer shall grant an unpaid leave of absence to an employee who is appointed or elected to a Union Office or to a full-time position as a staff representative for a period of up to and including five (5) years.
- (b) A request for such an approved leave must be given to the Employer by the Union, in writing at least fourteen (14) calendar days prior to the commencement of such leave, on Union letterhead and signed by an Officer of the Union.
- (c) An employee who obtains such a leave of absence shall return to his employment within thirty (30) calendar days after the completion of his employment with the Union.
- (d) The Employer is not obligated to grant such leave to more than one employee at a time from any one department.

16.02 LEAVE OF ABSENCE: UNION CONVENTIONS AND EDUCATIONAL PROGRAMS

- (a) Subject to the requirements of Paragraph (c), the Employer, upon receipt of written notice from the Union, shall grant leave of absence without pay to employees who
 - (i) have been elected as delegates to attend Union conventions;
 - (ii) have been appointed to act as members of the Union's negotiating committee;
 - (iii) have been elected as members of the Union's Executive Board; or
 - (iv) have been selected by the Union to attend bona fide shop steward education programs of up to fifteen (15) working days.

- (b) The Employer may grant further unpaid leaves of absence to employees for the purpose of attending mutually agreed upon educational programs within the hospitality industry. Written applications for such leave must be received at least seven (7) days prior to the commencement of such leaves.
- (c) The Employer is not obligated to grant any leave pursuant to this Article 16.02 to more than one employee at a time from any one department.
- (d) Written notice shall be given at least seven (7) calendar days prior to the commencement of any leave granted pursuant to this Article 16.02.

16.03 LEAVE TO APPEAR AS WITNESS

- (a) Subject to the provisions of Paragraph (b), any employee covered by this Agreement who is required to attend any commission, court or hearing, to give evidence in any civil or criminal case respecting the hotel in which he is employed, shall be compensated at the same hourly rate as called for in this Agreement, with a minimum of four (4) hours pay and a maximum of eight (8) hours pay for each day of attendance.
- (b) For the purposes of this Article 16.03, a commission, court or hearing does not include a grievance arbitration hearing, an industrial inquiry commission, the Labour Relations Board or any other tribunal or hearing which is concerned with the labour-management relationship between the Employer and the Union.
- (c) Notwithstanding Paragraph (b), an employee who is called by the Employer to give evidence at a grievance arbitration hearing, an industrial inquiry commission, the Labour Relations Board or any other tribunal or hearing which is concerned with the labour-management relationship between the Employer and the Union, shall be compensated in accordance with the provisions of Paragraph (a).

16.04 BEREAVEMENT LEAVE

- (a) A regular employee or a Part-Time employee will be granted three (3) days off without loss of pay in the event of the death of an immediate family member. If an employee is traveling outside of North America for the funeral of an immediate family member, they will receive five days of paid bereavement leave.
- (b) For the purposes of this Article, "immediate family" means the employee's mother, father, step-parents, son, daughter, sister, brother, spouse, father-in-law, mother-in-law and grandparents.

- (c) For purposes of this Article, "spouse" shall be defined to include a commonlaw spouse with whom the employee has cohabited for a minimum of one (1) year.
- (d) The Employer is entitled to require proof of death, relationship and, when necessary to travel outside of North America, proof of travel.

16.05 MATERNITY LEAVE

- (a) Subject to the provisions of Paragraphs (b), (c), (d) and (e) of this Article 16.05, an employee who has completed her probationary period, shall be entitled to a leave of absence without pay in accordance with all of the provisions of Part 7 of the Employment Standards Act, as it was enacted at the date this Agreement was executed, or as it might subsequently be amended from time to time. It is understood and agreed that the provisions of the said Paragraphs (b), (c), (d) and (e) are intended to clarify and to supplement, but not to limit the application of Part 7 of the Employment Standards Act.
- (b) An employee who resumes employment on the expiration of a leave of absence granted pursuant to this Article, shall be reinstated by the Employer in the position previously occupied by the employee, or in a comparable position, and with all increments to wages and benefits to which the employee would have been entitled had the leave of absence not been taken.
- (c) Where the employee elects to continue to pay her contributions to the Health Care Plan, the Employer shall be required to continue making its contributions to the Health Care Plan and to the Local 40 Pension Plan in accordance with Article 15.02, and on the basis of those hours which result from application of the formula determined by the Trustees of the Health Care Plan and in effect at the time.
- (d) An employee who has been granted a leave of absence pursuant to this Article 16.05, shall retain her seniority during the period of the leave of absence.
- (e) The period of service of an employee who has been granted a leave of absence pursuant to this Article, shall be considered to have been uninterrupted for purposes of determining annual vacation entitlement.

16.06 ADOPTION LEAVE

(a) Upon request, an employee who has completed his probationary period, will be granted adoption leave without pay in accordance with the provisions of

- the Employment Standards Act as it was enacted on March 22, 1991, or as it might subsequently be amended from time to time.
- (b) The period of adoption leave shall commence within two (2) weeks of the date of the adoption.
- (c) The provisions of Paragraphs (b), (c), (d) and (e) of Article 16.05 shall apply to an employee who has been granted adoption leave pursuant to this Article 16.06.
- (d) An employee who is the spouse of another employee who has already been granted adoption leave pursuant to this Article 16.06, is not entitled to such adoption leave.

16.07 PATERNITY LEAVE

- (a) In the case of the birth of his child, an employee who has completed his probationary period, shall be entitled to a leave of absence without pay in accordance with the provisions of the Employment Standards Act as it was enacted on March 22, 1991, or as it might subsequently be amended from time to time.
- (b) The provisions of Paragraphs (b), (c), (d) and (e) of Article 16.05 shall apply to an employee who has been granted paternity leave pursuant to this Article 16.07.

16.08 PAID PERSONAL DAYS

Effective January 1, 2008, all Regular employees will receive one (1) paid personal day per year. (Increased to two (2) paid personal days per year effective January 1, 2009). These days must be used within the year, cannot carry forward from one year to the next, and cannot be paid out in lieu of. Paid personal days will not be unreasonable denied.

16.09 LEAVE FOR MILITARY SERVICE

Members of the Union called up for the Military, Air Force or Naval Services, Red Cross or other combat relief service of Canada during the life of this Agreement will be considered on leave of absence and be returned to their former position upon honourable discharge from the service, provided they are physically and mentally capable and make application within two (2) months.

16.10 LEAVE FOR JURY DUTY OR AS WITNESS FOR CROWN

Employees who serve on a jury or as a witness for the Crown shall be granted leave of absence for this purpose and provided that the employee concerned deposits with

the Employer any pay received the employee shall continue to receive his full wages for such period of time.

16.11 GENERAL LIMITATION ON LEAVES OF ABSENCE

- (a) All leaves of absence provided for in this Agreement are leaves without pay, unless it is specifically provided in the appropriate article that the particular leave of absence is to be granted with pay.
- (b) Leaves of absence other than those specifically provided for in this Agreement may be granted to employees where it is deemed appropriate to do so by the Employer, but the granting of such leaves is within the discretion of the Employer. The granting of such leave will be in writing.

16.12 MULTIPLE APPLICATIONS FOR LEAVES OF ABSENCE

- (a) With respect to all applications made by employees for leaves of absence, preference will be given to those which are first received, provided that where more than one such application is received on the same day, preference will be given to the senior employee.
- (b) In cases where an employee's application is not made in writing, the Employer will be responsible for recording it in writing and dating it.
- (c) The Employer undertakes to make every reasonable effort to reach its decision

on such applications within a reasonable period of time following the application, and to communicate it to the affected employee with a minimum of further delay.

ARTICLE 17 ROOM ATTENDANT WORKLOAD

During the term of this Agreement, the Employers will implement the following measures with respect to the working conditions of Room Attendants:

17.01 WORKLOAD CONDITIONS

- (a) Room Attendants will not be assigned more than 15 credits or rooms per day.
- (b) When a Room Attendant is assigned 8 or more check-outs per day, the daily room assignment shall be reduced by (1) room or credit; in addition, when a Room Attendant is assigned 11 or more check-outs

- in a day, the daily assignment shall be reduced by two (2) rooms or credits (inclusive of the one (1) room credit for 8 or more check-outs.)
- (c) VIP floors will have no more than 14 rooms or credits assigned per day.
- (d) When a room attendant is assigned nine or more double/double rooms in one section, the daily room assignment will be reduced by one credit.
- (e) In any case where a Room Attendant is required to clean a full quota of rooms on more than two (2) floors, she shall be relieved of one (1) credit.
- (f) A Room Attendant shall not be required to make up more than three (3) cots or cribs.
- (g) In any case where a room is deemed to be in extraordinary condition, the Room Attendant shall immediately notify a Supervisor who will inspect the room to determine what assistance may be necessary.
- (h) The delivery of robes to guest floors will not be the responsibility of the Room Attendant.
- (i) Each Room Attendant will be provided with adequate supplies.
- (j) All practices which are superior to the conditions listed above will continue.

17.02 ROOM ATTENDANT WORKLOAD COMMITTEE

In order to provide for regular reviews of the factors which affect the workload of Room Attendants, each Employer will hold a meeting every six months at which will be present a representative group of Room Attendants, their Shop Steward, the Shop Chairperson, their Department Head, the Director of Human Resources and the General Manager or another member of the Hotel's Executive Committee.

17.03 REST PERIODS

Notwithstanding the provisions of Article 9.08 [Payment for Time In Lieu of Breaks], it is acknowledged that it is unacceptable for Room Attendants to intentionally forego their rest periods and/or their meal breaks. Each of the GVHEA General Managers will ensure that this is understood by all subordinate managers and supervisors and that they will take the necessary steps to ensure that their employees are aware of this policy.

ARTICLE 18 MISCELLANEOUS EMPLOYEE ENTITLEMENTS

18.01 PROTECTED WORKING CONDITIONS

- (a) All working conditions at present in force which are not specifically mentioned in this Agreement and which are not contrary to its general purpose and intent shall continue in full force and effect unless cancelled or terminated in accordance with the terms of this Article.
- (b) Any working condition which was implemented by the Employer on a conditional basis can be terminated when the terms of the condition have been exhausted or fulfilled, or the condition has been withdrawn.
- (c) Any other working condition which was granted by the Employer but which is not specifically provided for in this Agreement may be cancelled by the Employer by:
 - (i) serving the Union with written notice within thirty (30) days of the ratification of this Agreement; or
 - (ii) serving the Union with written notice of cancellation effective on the last day of each year of this Collective Agreement.

18.02 CAFETERIA, KITCHEN AND DINING LOUNGE MEAL ALLOWANCE

A wholesome meal shall be supplied by the Employer with no deduction from the employee's wages on the following basis:

All shifts in excess of five (5) hours worked shall receive one (1) meal per day.

18.03 EMPLOYEE ATTENDANCE AT STAFF MEETINGS

- (a) Where an employee is directed by the Employer to attend a staff meeting during his regular working hours the employee shall be compensated at his regular hourly rate for the time spent in such attendance.
- (b) An employee who is directed to attend a staff meeting is not entitled to claim overtime pay for such attendance, unless the time spent in the meeting results in the employee working more than eight (8) hours in a day, or more than forty (40) hours in a week.
- (c) Where the attendance of an employee at a staff meeting is voluntary, in response to an invitation and not a direction of the Employer, the Employer is

- not obligated to compensate the employee for the time spent in such attendance.
- (d) Where an employee is directed by the Employer to attend a staff meeting during his regular days off, the employee shall be compensated at his regular hourly rate for the time spent in such meeting.
- (e) Notwithstanding any of the foregoing provisions of this Article 17.03, any employee hired after June 30, 1998, may be required to attend in-house training sessions of general application to other employees, at which it is expected that he will acquire new or enhanced skills or knowledge, and, providing he is given at least fourteen (14) days notice of the date of the session, he shall be paid at his regular straight time hourly rate for such attendance to an annual maximum of sixteen (16) hours, providing such hours do not exceed ten (10) on any working day, eight (8) on any day off, or forty-eight (48) in any week.

18.04 EMPLOYEES RETURNING TO WORK AFTER ILLNESS OR INJURY

- (a) In any case where an employee provides notice to the Employer of his intention to return to work following an absence due to illness or injury, including absences covered by Workers' Compensation Board benefits, the employee will be entitled to be reinstated in his former position as soon as the Employer has been able, without incurring any penalty, to provide the appropriate notice, as required by the provisions of Article 9.13, to any other employee who might have been scheduled to work in place of the returning employee during that employee's absence.
- (b) Prior to reinstating the employee, the Employer is entitled to require documentation from a physician or from the Workers' Compensation Board, certifying that the employee is physically able to resume the performance of the duties.

18.05 NO INDIVIDUAL CONTRACTS OR AGREEMENTS

- (a) No employee shall be compelled to or allowed to enter into any individual contract or agreement with his Employer concerning the conditions of employment varying the conditions of employment contained herein.
- (b) No employee shall be asked to make a written or verbal agreement with the Employer covering hours of work, wages or conditions during the term of this Agreement.
- (c) Notwithstanding the provisions of subsections (a) and (b), an employee may legitimately be required to sign for the receipt of cash or other assets of his Employer which have been entrusted to his safekeeping.

- (d) Notwithstanding the provisions of subsections (a) and (b), an employee may legitimately be required to sign for the receipt of a written policy or procedure, providing it is made clear on the receipt form that by signing his acknowledgement, he will not forfeit any of his rights pursuant to the grievance procedure if he is subsequently disciplined for his failure to follow the policy or procedures.
- (e) The receipt form referred to in Paragraph (d), shall be the standard form as follows:

ACKNOWLEDGEMENT OF RECEIPT OF WRITTEN POLICIES OR PROCEDURES

| I, | , hereby acknowledge the receipt of a copy of a document |
|-----------------------------|--|
| entitled | which is dated |
| | nce with Article 17.05(d) of the Collective Agreement which stipulates how d to acknowledge the receipt of their Employer's written policies and |
| 0 0 | ipt does not in any way diminish the rights to which employees are entitled redure contained in the Collective Agreement. |
| | E OF THE EMPLOYEE OR OF A SHOP STEWARD MADE ON OF THE EMPLOYEE AND IN THE EMPLOYEE'S PRESENCE |
| DATED thisBritish Columbia. | day of, 200 at the City of Vancouver in the Province of |

18.06 HEALTH LAWS AND LOCKUP FACILITIES FOR EMPLOYEES

(a) It shall be a joint responsibility of the Employer and the employees to maintain strict observance of all sanitation and health laws and regulations, insofar as they affect the working conditions of the employees.

- (b) The Employer agrees to provide adequate lockup facilities for its employees' personal effects, namely purses and/or wallets. The Employer is not entitled to enter an employee's locker except in the presence of one of the following:
 - (i) the employee;
 - (ii) the Shop Chairperson;
 - (iii) one of the Shop Stewards; or
 - (iv) in an emergency when none of the aforementioned persons is immediately available, another member of the bargaining unit.

18.07 COMPENSATION TO EMPLOYEES RE: ENFORCEMENT OF HOUSE RULES FOR PATRONS

- (a) Upon presentation of a written bona fide claim by an employee, the Employer shall compensate the employee for the replacement cost of, or repair of, any wearing apparel, false teeth, eye glasses or contact lenses, or hearing aids, damaged or destroyed as a consequence of the employee's participation in the enforcement of House Rules and/or Liquor Control Board Regulations and/or at the direction of management or a person appointed by management.
- (b) If as a consequence of the proper performance of his duties, an employee is charged with a criminal offence or becomes the object of a civil suit for damages, he will be entitled to compensation for the reasonable costs of his legal defence which,
 - (i) in the case of a criminal charge, and where either the charge is subsequently dropped or the employee is tried and acquitted, shall be paid by the Employer to a maximum of five thousand dollars (\$5,000.00); and
 - (ii) in the case of a civil suit, shall be totally paid by the Employer.

18.08 SEVERANCE PAY

- (a) Each employee having completed at least one (1) year of service with his/her Employer, shall, upon retirement, permanent lay-off and loss of seniority, resignation and in cases of dismissal for non-culpable reasons be entitled to severance pay in accordance with the provisions of Paragraph (b), (c) or (d) below.
- (b) For employees employed by the Hyatt Regency Vancouver Hotel, severance pay shall consist of twelve (12) hours' pay at the employee's current hourly rate for each one thousand eight hundred and twenty (1,820) hours actually

- worked by the employee during his/her last period of continuous employment subsequent to January 1st, 1972.
- (c) For employees employed by the Renaissance Vancouver Hotel Harbourside, severance pay shall consist of twelve (12) hours' pay at the employee's current hourly rate for each one thousand eight hundred and twenty (1,820) hours actually worked by the employee during his last period of continuous employment subsequent to December 31st, 1987, and twelve (12) hours' pay at the employee's current hourly rate for each year of continuous service prior to 1988.
- (d) For employees employed by the Four Seasons Hotel and The Westin Bayshore Resort & Marina, severance pay shall consist of twelve (12) hours' pay at the employee's current hourly rate for each one thousand eight hundred and twenty (1,820) hours actually worked by the employee during his/her last period of continuous employment.
- (e) At the time he/she receives his/her severance payment, the employee shall be given a written statement by the Employer, clearly identifying the basis for calculating it.
- (f) The calculation of an employee's severance pay, will consist of
 - 1) dividing the total number of hours actually worked by the employee by 1,820;
 - 2) taking the calculation to two decimal places;
 - 3) multiplying the resulting number by twelve (12) times the employee's current straight-time hourly rate; and
 - 4) taking the product to the nearest cent.

18.09 SUCCESSOR OWNER OBLIGATION

In the event of the sale of the establishment it is the obligation of the successor owner to abide by all the terms and conditions of this Agreement.

18.10 LIMITATION ON EMPLOYEE ENTITLEMENTS

Employees who are not actively employed are only entitled to continue to receive such rights, entitlements and benefits, as are specifically given to them by the express terms and conditions of this Agreement, or by applicable legislation.

18.11 INDUSTRIAL FIRST AID TRAINING OF EMPLOYEES

In any case where the Employer agrees that it is in the best interests of the hotel that an employee undertake training to obtain, maintain or upgrade a recognized Industrial First Aid Certificate, and where the employee does undertake such training, the employee shall be granted leave of absence without loss of pay for the time required to undertake such training.

18.12 BAGGAGE HANDLING FEES

The Employers agree that 100% of all porterage fees collected will be distributed to the bargaining unit employees providing the porterage service. Complete disclosure of all of the porterage charges and the distribution to employees will be provided to the employees on a regular basis.

18.13 ROOM SERVICE GRATUITIES

The Employers at the four hotels agree that a fixed Room Service gratuity will be levied at not less that 15% and will not be reduced during the term of the collective agreement. The Employers also agree that 100% of the Room Service gratuities collected will be distributed to the bargaining unit employees providing the service. Complete disclosure of all the Room Service gratuities and distribution to employees will be provided to the employees on a regular basis.

This will become effective coincident with the printing of new Room Service menus, but not later than January 1, 2008.

ARTICLE 19 EMPLOYEE CONDUCT AND DRESS

19.01 HOUSE RULES GOVERNING CONDUCT OF EMPLOYEES

It is mutually agreed that the Employer will post House Rules for the conduct of employees and file a copy of those house rules with the Union before enforcing same. Filing with the Union office is accomplished by delivery of a copy of the house rules through registered mail.

19.02 CONTROL OF ABSENTEEISM

- (a) The Employer may require an employee to provide a medical certificate as evidence of the employee's illness or injury as a cause for the employee's absence from work.
- (b) Other than in exceptional circumstances, the Employer will not require an employee to provide a medical certificate as evidence of the employee's illness or injury as a cause of the employee's absence from work for a period

- of three days or less, unless the Employer has already developed, and communicated in writing to the employee its concern about the employee's record of attendance.
- (c) In any case where the Employer decides to communicate in writing to an employee its concern about the employee's unacceptable pattern of absenteeism, its communication shall be delivered to the employee at a meeting at which the employee shall be entitled to have present either a Shop Steward or another bargaining unit employee of his choice, and, from that time onwards, in addition to his obligation to report his absence to the Employer pursuant to Paragraph (d) of this Article, the employee shall be required personally to contact his department head or a designated alternate during the employee's normal working hours on each day of his continuing absence from his scheduled shifts, unless the absence is prolonged, bona fide and supported by medical documentation, in which case the daily reporting requirement shall not apply.
- (d) Every employee who is unable to report for work due to illness or injury are obligated to provide the Employer with notice at the earliest possible time to allow the employer time to cover the absence. Employees who are unable to reach their own manager will notify a Manager on Duty and, in addition to providing their name and department, will provide their absence reason. Voice mail or electronic mail are not acceptable means of reporting an absence. In the event that the Employer is not satisfied by objective evidence that there is proper justification or reason for an employee's absence, such an absence will be just and reasonable cause for discipline.
- (e) Where the Employer is satisfied by the objective evidence that an employee is unwilling to maintain a satisfactory attendance record in fulfillment of the employment relationship with the Employer, the Employer may terminate the services of the employee.
- (f) In relation to any provision in this Collective Agreement where an Employer is entitled to require medical evidence of an employee's ability to return to work or to continue to work, the Employer may require that the employee be examined by and present a medical certificate from a physician selected by the Trustees of the Health and Welfare Plan as identified in Article 15.04. In the event that an Employer requires an employee to submit to such an examination, any resulting charge by the doctor will be paid by the Employer.
- (g) In any case where the Employer considers it necessary to obtain medical evidence of an employee's ability to return to work or to continue to work, the Employer may so advise the Administrator of the Health Care Plan referred to in Article 15.04 of this Agreement, and the Administrator shall

thereupon refer the matter to a panel of two Trustees of the said Health Care Plan.

The two Trustees, one of whom shall have been appointed from among the Employer Trustees and the other of whom shall have been appointed from among the Union Trustees, shall, if in agreement with the Employer's determination, be authorized to designate a physician to conduct a medical examination of the employee and to submit a medical certificate to the Employer indicating whether or not the employee is able to return to work or to continue to work.

If the two Trustees do not agree with the Employer's determination, the Employer shall be advised of that fact forthwith, and the Employer shall have the right to refer the matter to the full Board of Trustees for its decision.

In order for the panel of Trustees or the full Board of Trustees to make an expeditious decision, the Trustees shall maintain a register of physicians from among whom the panel or the full Board may select an appropriate name in any case.

(h) If the absence is for three days or longer, and no notice or reasonable explanation is provided, at the time of absence or as soon as practical afterwards, the employee will be considered to have abandoned his/her job and his/her employment will be terminated.

19.03 AUTHORITY RE: CHEQUES AND CREDIT

When an employee is authorized to cash cheques, honour credit cards or credit accounts, he or she will not be held responsible for any losses provided he or she has followed management's instructions, but where an employee assumes responsibility of cashing cheques, honouring credit cards or credit accounts without such authorization from management he or she will be held responsible.

19.04 UNIFORM MODE OF DRESS

In consideration of the endeavor to improve the standards of the beverage dispensers industry, it is agreed that a proper uniform mode of dress shall be adopted, i.e. dark trousers or skirts, white shirts or blouses, and uniform tie. This dress and the cleaning thereof will be the responsibility of the employee and shall be effective upon the signing of this contract.

19.05 UNCONVENTIONAL MODE OF DRESS

Where an unconventional mode of dress or uniform is required by management, it is agreed the dress or uniform shall not be such as to cause discomfort, ridicule or embarrassment to the employee.

19.06 SPECIAL UNIFORMS

If any special uniform shall be required, such as tuxedo, white jackets, etc., it is agreed that the Employer shall supply same and be responsible for the cleaning thereof. All uniforms or special articles of wearing apparel worn by the employee while on duty shall be supplied and laundered by the Employer free of cost to the employee.

19.07 TELEPHONE AND MAILING ACCESS

Every employee shall provide the Employer with a current mailing address and telephone number at which he may be reached or contacted in an emergency, and shall be responsible for notifying the Employer of any subsequent changes to either one. The Employer agrees to respect the confidentiality of this information, and undertakes not to use it or to make it available unnecessarily.

ARTICLE 20 LIQUOR CONTROL LEGISLATION AND REGULATIONS

20.01 EMPLOYEES SERVING LIQUOR

If the employee is directed by a person designated by the Employer to serve a person whose age is in doubt, the Employer shall accept and bear the full responsibility and shall pay any fines or penalties incurred by the employee as a consequence of such service.

20.02 CHANGES IN REGULATIONS

It is mutually agreed that upon the implementation of any changes in the Liquor Control Board Regulations governing licensed premises, the Union and Management will negotiate an agreement on the problem that could arise therefrom.

ARTICLE 21 DISCIPLINE AND DISCHARGE OF EMPLOYEES

21.01 DISCIPLINE AND DISCHARGE OF EMPLOYEES

(a) Pursuant to Section 84(1) of the Labour Relations Code, the following standards shall be applied:

- (i) Employees who have successfully completed their probation period may only be disciplined or discharged for just and reasonable cause.
- (ii) During the probation period specified in this Agreement, an employee may be discharged if he is unsuitable for status as a non-probationary employee.
- (b) In the event that a non-probationary employee is discharged for just and reasonable cause, the Shop Chairperson will be notified by the Employer and will be provided with the reasons for the discharge.

21.02 PERSONNEL FILES

Upon the provision of reasonable notice to the Employer, an employee or a Union representative duly authorized in writing by the employee, shall be entitled to inspect the employee's personnel file.

21.03 WRITTEN RESPONSES TO WRITTEN DISCIPLINE

In any case where an employee has received from the Employer a written confirmation of a disciplinary penalty which the Employer intends to place on the employee's personnel file, the employee shall be entitled to submit to the Employer a written response which shall also be placed on the employee's personnel file.

21.04 COMMUNICATION DURING GRIEVANCES

In the absence of the Union's consent, the Employer may not initiate discussions with an employee who has filed a grievance, or on whose behalf the Union has filed a grievance, provided that nothing shall prevent the Employer, without the need for the Union's consent, from carrying on discussions concerning a grievance with a grievor who initiates such discussions.

21.05 EMPLOYEE'S RIGHT TO REPRESENTATION

- (a) Subject to the provisions of Paragraph (c), in any instance where the Employer issues a written warning or other, more severe form of disciplinary sanction to an employee which it
 - (i) presents to the employee at a meeting attended by more than one (1) representative of the Employer; and
 - (ii) intends to record in the employee's personnel file,

the employee shall have the right to have his Shop Steward or, in the absence of his Shop Steward, the Shop Chairperson present at the meeting.

- (b) In any instance where the employee referred to in Paragraph (a) is himself a Shop Steward or the Shop Chairperson, he shall be entitled to have another Union official present at the meeting.
- (c) In the case of any meeting which has been called by the Employer for the purpose of issuing a written warning or other, more severe form of disciplinary sanction, the function of the Shop Steward, Shop Chairperson or other Union official during that portion of the meeting which is devoted to the issuing of the disciplinary sanction, shall be expressly limited to the roles of witness and observer. No such representative of the Union may interfere in any way with the issuing of the disciplinary sanction during that portion of the meeting which is devoted to such purpose.

21.06 THE EXPUNGING OF CERTAIN DISCIPLINARY PENALTIES

Effective the date of ratification of this Agreement, any verbal or written warning, covering any matter other than sexual or personal harassment, theft, breach of trust, or acts of violence, that has been placed on the file of an employee, will subsequently be removed from his file as soon as the employee has been employed for a further continuous period of one (1) year without incurring an additional disciplinary penalty of any kind.

ARTICLE 22 GRIEVANCE AND ARBITRATION PROCEDURES

22.01 DEFINITION AND RECOGNITION OF A GRIEVANCE

- (a) Any complaint, disagreement or difference of opinion between the parties respecting the interpretation, application, operation or alleged violation of this Collective Agreement, including any dispute with regard to discipline or discharge, shall be considered to be a grievance.
- (b) Any such complaint, disagreement or difference of opinion will not be recognized as a grievance unless the grievance procedure is followed.

22.02 GRIEVANCE PROCEDURE

(a) **Informal Step:**

As an informal step, the employee is encouraged to make an earnest effort to resolve the grievance directly with the management person to whom he reports. At his option, the employee may be accompanied by the Shop Steward for the department in which the employee works.

(b) **Step One:**

At this step, notice in writing of the grievance must be filed with a person designated by the Employer, within ten (10) working days after the occurrence of the alleged grievance or of the date on which the employee first has knowledge of it.

The notice in writing shall briefly but clearly describe the nature of the incident or occurrence which gave rise to the grievance, and it shall clearly state the provision of the Agreement which has been violated.

Any meeting between the parties at this step must involve the employee, his Shop Steward and a person from management other than the employee's immediate supervisor.

The Employer's representative must answer the grievance in writing within ten (10) days.

(c) **Step Two:**

In the event that a resolution of the grievance, satisfactory to the Union and the Employer, does not result at Step One, an attempt to resolve the grievance shall be made between the employee, the Shop Chairperson and/or a Union Representative and a person or persons designated by the Employer.

This step must be taken by notice in writing within five (5) days of the date on which the written answer was delivered in Step One.

(d) **Step Three:**

In the event that a resolution of the grievance, satisfactory to the Union and the Employer, does not result at Step Two, either the Union or the Employer may advance the grievance to the next step. The next step involves a selection from the following alternatives:

- (i) the Optional Grievance Procedure provided for in Article 21.06;
- (ii) to a single Arbitrator as agreed in Paragraph (e) below; or
- (iii) directly to full arbitration at Step Four; or
- (iv) use the Fast Track Med/Arb process in Article 21.11.

In the case of a grievance arising from the dismissal of an employee, this step must be taken by notice in writing within thirty (30) calendar days of the date on which the grievance was advanced to Step Two.

In the case of any other grievance, this step must be taken by notice in writing within sixty (60) calendar days of the date on which the grievance was advanced to Step Two.

(e) If both parties agree, the grievance may be heard by a single Arbitrator. The parties shall have five (5) working days to agree on an Arbitrator. Failing such agreement, either party may request the Minister of Labour to appoint such Arbitrator. Articles 21.04 and 21.05 shall apply to such single Arbitrator.

(f) **Step Four:**

The final step of the grievance procedure shall be full arbitration as provided herein, unless the parties have previously agreed to be bound by the recommendations of an officer appointed by the Labour Relations Board, or by the recommendations of the investigator under the optional grievance procedure.

(g) Union and Employer Policy or General Grievances:

The Union or the Employer may file policy, or general grievances. Such grievances shall be filed at Step Two of the Grievance Procedure.

(h) **Step One and Step Two Responses**

Whether done so orally or in writing, no answer provided by the Employer at Step One, nor any element of the discussions between the parties at Step One or Step Two, may be introduced as evidence at any later step in the grievance procedure, unless done by mutual consent in any particular case.

22.03 ESTABLISHING THREE MEMBER BOARD

Seven (7) full days (excluding Sundays and holidays) shall be allowed for the setting up of a Board of Arbitration. It shall be composed of one (1) representative of the Union and one (1) representative of the Employer, and these two (2) members shall then select an impartial chairman. In the event of failure of these two (2) representatives agreeing on a chairman, the Minister of Labour shall be asked to appoint one.

22.04 ARBITRATION HEARING AND AWARD

(a) As soon as a chairman has been appointed, the Arbitration Board will be encouraged to commence the hearing within five (5) days and further encouraged to render a decision within fourteen (14) days.

- (b) In order to expedite the arbitration process, the parties agree that they will meet to discuss their understanding of the issue or issues to be placed before the Arbitration Board and to prepare a statement of all facts which are not in dispute. The identification of the issue or issues and the statement of agreed facts will be prepared in written form and placed before the Arbitration Board by agreement of the parties.
- (c) Each party to the arbitration will bear the expense of its nominee, and one-half (1/2) of the expense associated with the appointment of the chairman.
- (d) The parties recognize that they are bound by a decision of the Board, a majority of the Board, or by the Chairman of the Board in accordance with Section 94 of the Labour Relations Code.

22.05 AUTHORITY OF THE ARBITRATION BOARD

The parties to the arbitration recognize that the authority of the Arbitration Board is set out in Section 89 of the Labour Relations Code.

22.06 OPTIONAL INVESTIGATOR PROCEDURE

The parties have agreed to initiate an optional grievance investigation procedure on a trial basis, for the specified term of this Agreement, in accordance with the following:

(a) **Purpose and Scope:**

Recognizing that there are times and circumstances in which it may be necessary to seek third-party assistance in the resolution of grievances, and in an attempt to find a way in which to bring about such resolutions without incurring the costs and delays associated with formal arbitration proceedings, the parties have agreed to provide for an optional grievance investigation procedure.

The process is intended to complement the grievance and arbitration procedures otherwise provided for in this Agreement. It is not intended to replace those other procedures.

(b) **Optional Grievance Investigation Procedure:**

As provided for in Section 103 of the Labour Relations Code, where a difference arises between the parties relating to the dismissal, discipline or suspension of an employee, or to the interpretation, application, operation or alleged violation of this Agreement, including any question as to whether a matter is arbitrable, during the term of this Agreement, the parties will

appoint one of the persons named herein as "Investigators", or a substitute agreed to by the parties, to

- (i) investigate the difference;
- (ii) define the issue in the difference; and
- (iii) make written recommendations to resolve the difference

within thirty (30) days of the date of the receipt of the appointment and, for those thirty (30) days from that date, time does not run in respect of the Grievance Procedure.

(c) Cost Sharing:

As provided for in Section 103 of the Labour Relations Code, each party shall pay one-third (1/3) of the cost incurred in relation to the reasonable remuneration, travelling and out of pocket expenses of the Investigator or his substitute. The remaining one-third (1/3) will be paid by the Provincial Government.

Each of the parties shall be separately responsible for all other costs incurred by each of them in relation to the preparation and presentation of their respective cases and submissions to the Investigator.

(d) Investigators - Agreed Alternates and Selection:

The parties have agreed that for the term of this Agreement the persons named in a Letter of Understanding will be recognized as their "Investigators" for purposes of this investigation procedure, subject to receiving their respective consents to their appointment.

Selection of a particular named individual to serve in each instance shall be by agreement of the parties. Should the parties fail to agree on the selection, then the person next on the list after the last appointment shall be chosen.

(e) **Option Choice and Timing:**

Either party may choose to implement the investigation procedure, provided that all steps of the Grievance Procedure, prior to reference to arbitration, have been exhausted without a resolution of the difference.

The party wishing to use the investigation procedure shall notify the other party of the decision, within five (5) working days of the receipt of the reply at the last step of the Grievance Procedure. Such notification must be in writing.

The party receiving notification may refuse to accept the investigator procedure, in which case the arbitration provisions of this Agreement are then available and the time limit contained in that article begins to run from the date of the refusal decision being delivered in writing. No reasons for the refusal need be given, and such refusal must be submitted within five (5) working days.

(f) **Binding Recommendations:**

While the grievance investigation process is intended to yield only nonbinding recommendations, the parties may agree that the recommendations will represent a binding award, in the manner of an arbitration award. Such agreement must be made in advance of the appointment of the Investigator.

22.07 POWERS OF ARBITRATION BOARD

The powers of the Arbitration Board are limited to the application and interpretation of the Collective Agreement as written and the Board is not authorized to make any decision inconsistent with the provisions of this Agreement nor to alter, modify or amend any part of this Agreement.

22.08 TIME LIMITS ARE DIRECTORY NOT DESCRIPTIVE

A grievance or dispute shall commence within the time limits provided, otherwise it shall be deemed to be abandoned.

22.09 PERSONS AUTHORIZED TO DEAL WITH GRIEVANCES

- (a) The Union agrees to provide the Employer with a written list of the names of any persons other than Shop Stewards, who are authorized to deal with the adjustment or resolution of grievances on behalf of the Union, and to provide further written advice of changes made in the list from time to time.
- (b) The Employer agrees to provide the Union with a written list of the names of any persons who are authorized to deal with the adjustment or resolution of grievances on behalf of the Employer, and to provide further written advice of changes made in the list from time to time.

22.10 TROUBLESHOOTER PROCEDURE

(a) In an effort, among other things, to establish and maintain a workplace that is free of sexual and other forms of personal harassment, directed against members of the bargaining unit by either management personnel or other members of the bargaining unit, and in the belief that there are occasions when it will be mutually agreed by the parties that an expeditious and

- objective analysis of the nature of a dispute between two or more employees, is warranted, the parties have agreed to establish the following procedure.
- (b) If either party to the Agreement perceives a need for an unbiased, objective and independent third party investigation of a dispute arising in the workplace, whether between two or more bargaining unit employees or between a management person and one or more bargaining unit employees, that party may propose to the other party that a Troubleshooter be appointed for the purpose of attempting to resolve the dispute. When the Union's consent is required for the utilization of this procedure, it may only be provided by an officer.
- (c) The parties will by mutual consent establish and maintain a list of Troubleshooters which shall be large enough to ensure that it will always be possible for them to secure the services of one such Troubleshooter at short notice. In order to qualify for appointment to the list, a person will have acquired a reputation for objective fact-finding, and will be expected to possess, or to have the capacity quickly to gain a knowledge of the hospitality industry and its unique characteristics.
- (d) Following the appointment of a Troubleshooter to a particular dispute, a representative of each party will meet with the Troubleshooter, and will submit to the Troubleshooter and to the representative of the other party, all of the relevant information contained on file, an overview of the issues, concerns and interests involved in the dispute, and a list of the names and telephone numbers of all his witnesses.
- (e) The parties will seek as often as possible to be represented in the Troubleshooter proceedings by others than legal counsel.
- (f) The Troubleshooter may, at whatever stage in his mandate commends itself to him, conduct a preliminary interview with the grievor or with the employee who has been accused of an offence, and at the completion of his investigation will conduct a further interview at which the opportunity will be provided to the grievor to refute allegations made against him.
- (g) The Troubleshooter may, after having notified the parties in advance of his intention, interview other witnesses than those selected by the parties.
- (h) At the conclusion of his investigation, the Troubleshooter shall submit a preliminary report of his findings to the representatives of the parties, and will be encouraged at that point to attempt to mediate a resolution of the dispute.
- (i) In the event the Troubleshooter is unable to effect a mediated resolution of the dispute, he will be required to submit to both parties his recommendations

- for resolution contained in a report written in plain and intelligible layperson's language.
- (j) The costs of the Troubleshooter will be borne in three equal portions by the Employer, the Union and the Government of British Columbia pursuant to Section 103 of the *Labour Relations Code*.

22.11 FAST TRACK MED/ARB PROCESS

Recognizing that there are times when an expedited arbitration may be desirable, the parties agree that the following process may be used as a substitute for the formal Grievance Procedure outlined in Article 21 of the Collective Agreement.

- The process can only be used by mutual agreement between the parties who are signatory to this Collective Agreement. (i.e. H.I.R. Local 40).
- 2) The outcome will be binding on the parties.
- The cost will be borne in accordance with Section 103 of the Labour Relations Code, i.e., Employer-1/3, Union-1/3, Government-1/3.
- 4) The procedure cannot be used should an application for a Settlement Officer under Section 87 of the Labour Relations Code have been made by either party.
- 5) No legal counsel will be used by either party. The Union will use elected officers or business representatives. H.I.R. will use employees of their Industrial Relations Division.
- 6) The number of cases to be heard at any given time will not exceed three (3).
- 7) The parties or their representative will try to get an agreed Statement of Facts for presentation to the arbitrator.
- 8) Wherever possible the arbitrator will attempt to mediate a settlement between the parties.
- 9) In such case that the arbitrator must write a decision, such decision shall be brief and to the point.
- 10) An agreed schedule for the process will be arranged in advance, based on a mutual assessment of the length of time needed to present each case.
- 11) General rules of evidence will be waived except for the rule of "onus".

12) **Procedure Guidelines**

- a) The Opening Statement: This should basically set out the case from each party's perspective. The arbitrator will aggressively seek at this point to define the issue and to determine what evidence is agreed to and what is not.
- b) The Hearing: Sufficient witnesses should be called to ensure the "story" is properly told. Where it is an issue of credibility or conflicting evidence, the key individuals must testify.
- c) The Argument: As agreed, the parties will not cite legal precedents but may refer to Brown and Beatty, Palmer, etc. However, it is imperative that the relevant provisions of the Collective Agreement be canvassed by counsel to ensure that all relevant clauses are put before the arbitrator.
- d) *Mediation*: Counsel must accept some responsibility at this stage to assist the arbitrator in assessing the evidence before the arbitrator. Specifically, if counsel can assist in assessing credibility and/or contradictory evidence, they should do so.
- e) The Decision: If mediation fails or is not appropriate and if the decision can be rendered after a short deliberation, the arbitrator will do so. By meeting first with counsel to explain the framework of the arbitrator's decision, the parties are provided with an opportunity to influence the exact terms of resolution. Within the framework of settlement as outlined by the arbitrator, the parties can work out exact terms which best suit the specifics of the case. Such an opportunity should not be wasted by continuing to argue the merits of the case.

With respect to grievances involving customer complaints, the following will apply:

- i) the person to whom the complaint was given be called to testify;
- ii) bargaining unit or staff employees who can provide direct evidence with respect to the evidence be called to testify;
- iii) wherever possible, the complaint be committed to writing, in the customer's own handwriting;
- iv) prior to the hearing, the parties discuss the evidence so there are no surprises.

The Mediator/Arbitrator will be selected from the list contained in

ARTICLE 23 DEFINITIONS AND JOB DESCRIPTIONS

23.01 OBJECTIVE INTERPRETATION

Where a specific definition of a word, expression, term or a phrase is not expressly provided in this Agreement, such word, expression term or phrase shall be interpreted objectively, not subjectively; and according to common and normal grammatical usage.

23.02 TIME SPAN REFERENCES

References to days, weeks, months or years shall be under- stood to mean calendar days, weeks, months or years, unless otherwise expressly provided in this Agreement.

23.03 SPECIFIC DEFINITIONS

The following specific definitions of words, expressions, terms or phrases have been agreed to by the parties, and shall be used to establish the intent and meaning of the language of this Agreement, unless a different definition is provided within the context of a particular article:

Temporary Employee: An employee hired for a specific term or project, for a

period not to exceed ninety (90) days, with no expectation of continuing employment beyond the specified term or completion of the particular project

for which the employee was hired.

Probationary Employee: An employee who was hired into probationary status

and who has not successfully completed the

probationary period.

Regular Employee: An employee who is prepared to make himself

available without restriction.

Part-Time Employee: An employee who is not prepared to make himself

available without restriction.

Department: For purposes of this Agreement, "department" shall be

understood to mean those departments identified by the Employer and listed in the seniority lists prepared

and posted by the Employer.

(At the time this Agreement was signed, the departments in each hotel were those set forth in the Appendix numbered 1 which is attached to, and which forms part of the Agreement.)

If there is any uncertainty regarding the identification of classifications within departments, the Employer will respond by indicating in the written list of departments the particular classifications within departments.

Mixerologist (Service):

A mixerologist (Bartender) who does not receive a direct monetary consideration from a customer. (A non-gratuity employee.)

ARTICLE 24 TECHNOLOGICAL CHANGE

24.01 TECHNOLOGICAL CHANGE

- (a) It is the purpose of this Article 23 to provide for technological change in accordance with Section 54 of the Labour Relations Code.
- (b) "Technological change" shall be defined as the introduction by the Employer of, or the intention of the Employer to introduce a measure, policy, practice or change that affects the terms, conditions or security of employment of a significant number of employees to whom this Agreement applies.
- (c) If the Employer introduces a technological change that will directly result in the termination of a significant number of regular employees within the bargaining unit, then
 - (i) the Employer will provide at least sixty (60) calendar days notice to the Union;
 - (ii) the Employer will meet with the Union to discuss the technological change; and
 - (iii) the provisions of Paragraphs (d) and (e) shall apply.
- (d) Any regular employee whose position is terminated as the result of a technological change, shall be eligible for retraining to equip him to operate any new equipment. If such retraining is practical, it will be provided by the Employer without cost to the employee.

- (e) In any case where retraining is not practical, and where he has not taken advantage of the provisions of this Agreement to obtain another position within the hotel, a regular employee shall be entitled to the following:
 - (i) Two (2) weeks' pay or notice in lieu thereof if he has completed one (1) year of continuous service;
 - (ii) Three (3) weeks' pay or notice in lieu thereof if he has completed three (3) years of continuous service; and
 - (iii) One (1) additional week's pay or notice in lieu thereof for each further continuous year of service, up to a maximum of eight (8) weeks' pay or notice in lieu thereof.
- (f) In any case where it becomes necessary to displace existing employees, the parties will diligently seek opportunities to retrain such employees for other positions within the bargaining unit.

ARTICLE 25 THE RIGHTS OF EMPLOYEES TO PRIVACY AT THE WORKPLACE

The Employer and the Union acknowledge that the Employer has the need to establish and maintain a form of surveillance of its premises. The Employer and the Union also acknowledge that the Employer's need for surveillance must be balanced against the need of employees to enjoy the maximum possible privacy.

ARTICLE 26 LETTERS OF UNDERSTANDING

- **26.01** It is clearly understood and agreed by the Employer and the Union that the only Letters of Understanding that will be considered valid for the duration of this Collective Agreement are those Letters of Understanding renewed or amended from the Collective Agreement that expired on June 30, 2003.
- **26.02** The Letters of Understanding contained in the Collective Agreement that expired on June 30, 2003 and not contained in the renewal Collective Agreement are null and void.

| IN WITNESS WHEREOF the parties hereto had under the hands of the proper officers this | |
|---|---|
| SIGNED ON BEHALF OF: | SIGNED ON BEHALF OF: |
| HOSPITALITY INDUSTRIAL RELATIONS | GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION |
| Jack Butterworth Chairman of the Board | FOUR SEASONS HOTEL |
| Ron Schmidt Director | HYATT REGENCEY VANCOUVER |
| | RENAISSANCE VANCOUVER HOTEL HARBOURSIDE |
| | THE WESTIN BAYSHORE, VANCOUVE |
| | SIGNED ON BEHALF OF: |
| | UNITE HERE Local 40 |
| | Jim Pearson President/Administrator |

THIS IS THE APPENDIX 1 TO THE 2007-2010 COLLECTIVE AGREEMENT BETWEEN GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION

- AND -

UNITE HERE Local 40

LIST OF DEPARTMENTS

FOUR SEASONS HOTEL:

Kitchen

Bake Shop

Stewarding

Fine Dining Room

Restaurant

Lounge

Banquets

Room Service

Private Bar

Front Desk

Guest Services (including Bell-Doorman and Car Jockey)

Communications

Concierge Desk

Housekeeping (including Laundry)

Maintenance

Stores/Purchasing

Pool

HYATT REGENCY VANCOUVER:

Latte Café & Bistro

Banquets

Beverage

Front Office

Night Audit/Cashiers

Bell/Door Service

Housekeeping

Stewards

Stores/Receiving

Switchboard

Restaurant - Mosaic Bar & Grill

Kitchen

Engineering

Room Service

RENAISSANCE VANCOUVER HOTEL HARBOURSIDE:

Front Office (including Bell/Doorman)

Housekeeping (including Laundry)

Kitchen

Purchasing/Receiving

Restaurants (Patina/Vistas)

Room Service (including Mini-Bars)

Club Lounge

Lounge Services (Including Bartenders)

Banquets

Maintenance

Stewarding

Plantations

Reservations

THE WESTIN BAYSHORE, VANCOUVER:

Accounting

Banquets

Currents at Bayshore

Engineering

Front Office

Guest Services

Housekeeping

Kitchen

Receiving

Reservations

Seawall Bar & Grill

Service Express

Stewarding

THIS IS THE APPENDIX 2 TO THE 2007-2010 COLLECTIVE AGREEMENT BETWEEN GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION - AND UNITE HERE Local 40

It should be noted that certain of the rates for the First 6 Months, when calculated in accordance with the requirements of Article 12.01(b), fall below the British Columbia Minimum Wage Rate of \$8.00 that came into effect on November 1, 2001.

As a consequence, all such rates have been replaced on the wage rate schedules in this Collective Agreement, by the Minimum Wage Rate in effect at the time of executing this Agreement, and each such substituted rate is denoted by an asterisk (*).

THIS IS THE APPENDIX 3 TO THE 2007-2010 COLLECTIVE AGREEMENT -BETWEEN-

GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION -AND-

UNITE HERE Local 40

For the purposes of Article 9.1.05(d) of the Agreement, the work week shall, for the respective individual Hotels, be specifically defined as follows.

FOUR SEASONS HOTEL:

12.01 a.m. Monday to 11.59 p.m. Sunday

HYATT REGENCY VANCOUVER:

12.01 a.m. Monday to 11.59 p.m. Sunday

RENAISSANCE VANCOUVER HOTEL HARBOURSIDE:

12.01 a.m. Sunday to 11.59 p.m. Saturday

THE WESTIN BAYSHORE, VANCOUVER:

12.01 a.m. Sunday to 11.59 p.m. Saturday

THIS IS THE APPENDIX 4 TO THE 2007-2010 COLLECTIVE AGREEMENT -BETWEEN-

GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION -AND-

UNITE HERE Local 40

The maximum number of employees who may be granted Regular status in the Employers' Banquet departments at any one time, shall be as follows:

Four Seasons Hotel

Servers - 13;

Hyatt Regency Vancouver

Servers - 40;

Renaissance Vancouver Hotel Harbourside

Servers - 7;

The Westin Bayshore, Vancouver

Servers - 6.

- 1. With respect to the Four Seasons Hotel and Hyatt Regency Vancouver, any one or more of the above numbers may be reduced by the unilateral action of the Employer, but only if and when one or more of the stipulated number of Regular positions has or have been vacated by the incumbent(s).
- 2. With respect to the Hyatt Regency Vancouver, the maximum number of Servers may be reduced by the unilateral action of the Employer, but only if and when one or more of the stipulated number of Regular positions has or have been vacated by the incumbent(s).
- 3. With respect to the Hyatt Regency Vancouver, the Employer commenced on February 17, 1997 employing only 30 Regular Servers, and although the Employer has the unilateral right to employ up to 40 Regular Servers, it agrees to notify the Union at least two (2) weeks in advance of implementation, of its intention to increase the number above that which is established at that time.

Those who, on February 17, 1997, regardless of whether or not they were qualified for Regular status, applied therefor and who fell between numbers 31 and 40 in terms of seniority, both numbers inclusive, were placed on a waiting list, and are entitled to the benefits set forth in Paragraph 4 below.

APPENDIX 4 (cont'd) BANQUET DEPARTMENT

- 4. If a vacancy occurs among the Regular Servers in the Hyatt Regency Vancouver's Banquet department, it shall be offered first to the most senior of the Part-Time employees on the waiting list referred to in Paragraph 3 above. Any such employee on the waiting list who declines an offer of regular status, shall be removed from the list immediately, and shall no longer be eligible for the entitlement set forth in this Paragraph.
 - Any employee who attains Regular status pursuant to the provision contained in this Paragraph 4, shall be granted a Regular seniority start date that coincides with the date of his transfer to Regular status.
- 5. In the event an individual Hotel covered by this appendix wishes to increase the number of regular status employees on the current seniority list in the banquet department, then it will be accomplished by mutual agreement between the parties.

LETTER OF UNDERSTANDING #1 between GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION and UNITE HERE Local 40

CONTRACTED SERVICES

(Reference: Article 2.04)

This Letter of Understanding serves to qualify the language contained in Article 2.04 of the 2003-2007 Collective Agreement between the parties, to the extent that it is agreed between those same parties that Article 2.04 may not be applied or enforced in such a way as to interfere with any contracts which were entered into by any GVHEA hotel prior to March 13, 1995.

It is further agreed between the parties that the list of the areas which are to be protected by this Letter of Understanding, is that list which was furnished by the GVHEA to the Union by covering letter dated March 13, 1995.

Any new contract entered into by any GVHEA hotel which falls outside the areas protected by this Letter of Understanding, will constitute a violation of Article 2.04 of the Collective Agreement.

| DATED this day of _ | , 2007 at the City of Vancouver in the | |
|--|--|--|
| Province of British Columbia. | · | |
| | | |
| TO COMPANY AND LICED AND | | |
| HOSPITALITY INDUSTRIAL RELATIONS UNITE HERE Local 40 | | |
| | | |
| | | |
| Ron Schmidt | Jim Pearson | |
| Director | President/Administrator | |

LETTER OF UNDERSTANDING #1 - Contracted Services (cont'd)

March 13, 1995

Mr. Nick Worhaug President/Administrator Hotel, Restaurant & Culinary Employees' & Bartenders' Union, Local 40 100 - 4853 East Hastings Street Burnaby, B.C. V5C 2L1

Dear Sir:

Re: Article 2.04 of the Agreement: Contracted Services

I am enclosing a List of Contracts for each of the four GVHEA hotels which sets out the areas which it is agreed by the parties should be protected from the provisions of Article 2.04, pursuant to the Letter of Understanding that is appended to the Agreement.

The said List of Contracts is identical to the List that was protected by the 1991-1995 and 1988-1991 Agreements, and set out in Graham Leslie's letter dated February 6, 1989.

Where the Employers have entered into contracts in other areas which are not included in the above list, those contracts do not relate to work that comes under the jurisdiction of the Union.

None of the four GVHEA Employers intends to change its historical use of external services or forces, and no hotel will use external contracts to erode the scope of the existing certified bargaining unit.

Yours very truly,

Joann Pfeifer GVHEA Chairman

Encl.

LETTER OF UNDERSTANDING #1 - Contracted Services (cont'd)

"LIST OF CONTRACTS" FILED PURSUANT TO THE LETTER DATED MARCH 13, 1995 FROM JOANN PFEIFER TO NICK WORHAUG

FOUR SEASONS HOTEL:

Security

Laundry and Dry Cleaning

Maintenance and Engineering Services

Transportation Services

Music and Entertainment

Flowers, Plants and Grounds Maintenance

HYATT REGENCY VANCOUVER:

Security

Laundry and Dry Cleaning

Maintenance and Engineering Services

Transportation Services

Music and Entertainment

Flowers, Plants and Grounds Maintenance

Night Cleaning

RENAISSANCE VANCOUVER HOTEL HARBOURSIDE:

Security

Guest Dry Cleaning and Laundry

Cleaning of Staff Uniforms

Laundry of Food and Beverage Table Linens

Maintenance and Engineering

Flowers, Plants and Grounds Maintenance

Music and Entertainment

Transportation Services

THE WESTIN BAYSHORE, VANCOUVER:

Security

Parking and Valet Attendants

Maintenance and Engineering Services

Transportation Services

Music and Entertainment

Flowers, Plants and Grounds Maintenance

Laundry, Dry Cleaning and Valet

LETTER OF UNDERSTANDING #2 between GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION and **UNITE HERE Local 40**

OPTIONAL INVESTIGATOR PROCEDURE

[Reference: Article 22.06(d)]

The parties have agreed that for the period July 1, 2003 to and including June 30, 2007, the following persons will be recognized as the "investigators" for purposes of the optional investigator procedure:

Paul Devine

| o or more |
|---------------|
| |
| couver in the |
| |
| |

Jim Pearson

President/Administrator

Judi Korbin

gvhea-local40 99

Ron Schmidt

Director

LETTER OF UNDERSTANDING #3 between GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION and UNITE HERE Local 40

HYATT REGENCY VANCOUVER

(Reference: Article 9)

This Letter of Understanding serves to qualify the current Collective Agreement in general, and Article 9 thereof in particular, by incorporating into it the substance of the award of Arbitrator Bruce H. McColl, Q.C. which was published on April 10, 1987 in the matter of a dispute between Hyatt Regency Hotel and the Hotel, Restaurant and Culinary Employees and Bartenders Union, Local 40.

It is agreed between the parties that until such time as the Collective Agreement is specifically amended for the explicit purpose of altering the terms of the said award of the Arbitrator, its terms will prevail and will be applied in the interpretation and in the administration of the Collective Agreement.

| DATED this day of Province of British Columbia. | , 2007 at the City of Vancouver in the | |
|--|--|--|
| HOSPITALITY INDUSTRIAL RELATIONS UNITE HERE Local 40 | | |
| Ron Schmidt Director | Jim Pearson President/Administrator | |

LETTER OF UNDERSTANDING #4

between GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION and UNITE HERE Local 40

BANQUET SERVER EXTRAS

(Reference: Article 4)

- 1. Any employee who is engaged as a Banquet Server for the first time, shall, prior to commencing his employment, sign an authorization of check-off form in accordance with the requirements of Article 4.03(a).
- 2. Any such employee who believes that the number of hours of work available to him /her is very limited, may, with the consent of the Employer, elect to be covered by a Union permit, in which case the employee will not be required to pay an initiation fee to the Union.
- 3. There shall be no limit on the number of employees who elect to be covered by a Union permit or the length of time any one such employee may be covered.
- 4. Any employee covered by a Union permit shall be paid at the hourly rate of \$10.00, net only of the appropriate income tax and other statutory deductions. No part of the \$10.00 may be paid from the bargaining unit gratuity pool.
- 5. In the case of employees covered by a Union permit, the Employer shall be required to pay to the Union a Permit fee of One Dollar (\$1.00) for each hour worked by such employees, and the minimum amount of monthly dues as stipulated by the Union's current dues schedule.
- 6. For any month in which one or more employees covered by a Union permit, actually worked, the Employer shall at the time of remitting the required amount of Union dues and Permit fees, provide the Union with a statement in writing which shall list the names of those on whose behalf Union dues are being remitted, and the number of hours worked during the month by all employees covered by Union permits.

LETTER OF UNDERSTANDING #4 - Banquet Server Extras (Art. 4) (cont'd)

- 7. In the case of employees covered by a Union permit, the Employer shall not be required to pay any contributions to the Health Care Plan, Pension Plan or any Additional Union Assessment.
- 8. Employees covered by a Union permit are not entitled to receive any portion of the bargaining unit gratuity and service charge pool, and the Employer is not entitled to divert any portion of such pool to defray the cost of paying the Union permit fees.
- 9. At any time, an employee covered by a Union permit may apply to be initiated or reinstated into membership in the Union. From such time forth, the Employer will be required to deduct and to remit to the Union, the fee currently applicable in the particular situation, together with the appropriate amounts for Health Care, Pension and Additional Union Assessment, and will no longer be required to remit a Union permit fee.

| DATED this day of Province of British Columbia. | , 2007 at the City of Vancouver in the |
|--|--|
| HOSPITALITY INDUSTRIAL REI | ATIONSUNITE HERE Local 40 |
| Ron Schmidt Director | Jim Pearson President/Administrator |

LETTER OF UNDERSTANDING #5

gvhea-local40 102

DATED 41

between

GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION

and

UNITE HERE Local 40

PROVISIONS FOR BREAKING SENIORITY TIES

In any instance where the Employer starts up a new facility, the highest seniority ranking amongst employees in the new facility shall be granted to the employee having the earliest seniority start date in the department in which the facility is situated; or, as between two or more employees who are transferred into such department from other departments of the Hotel, the highest seniority ranking shall be granted to the employee having the earliest seniority start date in the department from which he was transferred; or, if two or more employees possess identical departmental seniority start dates, the highest seniority ranking shall be granted to the employee having the earliest hotel seniority start date.

In any instance where two or more employees are transferred into an existing facility, the highest seniority ranking shall be granted to the employee having the earliest seniority start date in the department in which the facility is situated; or, as between employees who are transferred into such department from other departments of the Hotel, the highest seniority ranking shall be granted to the employee having the earliest seniority start date in the department from which he was transferred; or, if two or more employees possess identical departmental seniority start dates, the highest seniority ranking shall be granted to the employee having the earliest hotel seniority start date.

In all of the above instances, existing employees shall be granted higher seniority ranking than newly hired employees.

LETTER OF UNDERSTANDING #5 - Provisions for Breaking Seniority Ties (cont'd)

| In any of the above instances, where a tie remains shall be broken in favour of the employee whose earliest month of whichever year each one was I would be granted higher seniority ranking than one | birthday occurs on the earliest day of the born. Thus, one born on March 6, 1966 |
|--|--|
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| DATED this day of | , 2007 at the City of Vancouver in the |
| Province of British Columbia. | , 2007 at the City of Vancouver in the |
| HOSPITALITY INDUSTRIAL RELATIONSUNIT | E HERE Local 40 |
| | |
| Ron Schmidt Director | Jim Pearson President/Administrator |

LETTER OF UNDERSTANDING #6 between GREATER VANCOUVER HOTEL EMPLOYERS ASSOCIATION

and UNITE HERE Local 40

RETIREMENT POLICIES

On behalf of its individual members, the Association agrees that as soon as possible following the ratification of a new collective agreement, each one will formulate and publicize a retirement policy containing the following principles.

- 1. The normal retirement age for all employees will be 65.
- 2. At some point during an employee's 65th year, but in any event no later than three months prior to his 65th birthday, the Employer will in writing advise the employee of his forthcoming 65th birthday, and will request the employee to complete a declaration form that will indicate whether or not the employee wishes to have his employment extended beyond his 65th birthday.
- 3. The declaration form shall state quite clearly that in a case where the employee opts to continue working beyond his 65th birthday, that option will be made subject to a determination by the Employer that the employee retains the ability to continue performing the full measure of the duties of his position.
- 4. An employee who receives from his Employer the written advice referred to in Paragraph 2, will be required to return his completed declaration form to the Employer, within one (1) calendar month of the date he received it.
- 5. In any case where an employee declares his intention to remain employed beyond his 65th birthday, the Employer shall in writing acknowledge his election, and, if the Employer has determined that the employee possesses the capacity to perform the full measure of the duties of his position, shall specify the period of such extended employment which may be for any period up to twelve (12) months.
- 6. In any case where an employee declares his intention to remain employed beyond his 65th birthday, but the Employer determines that the employee does not possess the capacity to perform the full measure of the duties of his position, the Employer shall notify the employee in writing of such determination within two (2) weeks of receiving the employee's completed declaration form, and shall simultaneously draw the attention of the employee to this Letter of Understanding.

LETTER OF UNDERSTANDING #6 - Retirement Policies (cont'd)

- 7. In the case of any employee who, at the time the renewed Collective Agreement is ratified, had already attained the age of 65 years, both the Employer and the employee shall conform with the requirements of paragraphs 1 to 6 above which shall be applied to the year in question and to the employee's immediately following birthday.
- 8. In any case where an Employer has extended an employee's period of employment beyond his 65th birthday, the same procedures as are outlined in Paragraphs 1 to 6 herein, shall be followed by both the Employer and the employee with respect to each succeeding extended period.
- 9. In any case where an Employer has extended an employee's period of employment beyond his 65th birthday, the employee shall be advised that he will be subject to an ongoing review and evaluation of his performance.
- 10. In any case where an employee who elected to remain employed beyond his 65th birthday or beyond any extended period thereafter, is denied the opportunity to continue working by his Employer, and is of the opinion that he retains the capacity to perform the full measure of the duties of his position, he shall be entitled to lodge a grievance pursuant to the procedure contained in Article 21 of the Agreement, within fourteen (14) calendar days of the date when he receives the Employer's notification to that effect.

| DATED this day of | , 2007 at the City of Vancouver in the | |
|--|--|--|
| Province of British Columbia. | · | |
| | | |
| | | |
| HOSPITALITY INDUSTRIAL RELATIONS UNITE HERE Local 40 | | |
| | | |
| | | |
| Ron Schmidt | Jim Pearson | |
| | | |
| Director | President/Administrator | |

gvhea-local40 106

LETTER OF UNDERSTANDING #7 between GREATER VANCOUVER HOTEL EMPLOYERS' ASSOCIATION and UNITE HERE Local 40

HYATT REGENCY VANCOUVER HOTEL THE SALE OF SUNDRY ITEMS ON THE EMPLOYER'S PREMISES

- WHEREAS the Employer intends to operate a lobby-based counter which will involve both the sale of food and beverages and the sale of sundry items;
- AND WHEREAS the Employer intends to operate the counter using employees who will become members of its bargaining unit, and who will receive pay and benefits as agreed to by the two parties;

AND WHEREAS for the purposes of the lobby-based counter

- (a) the Employer agrees that the sale of food and beverages constitutes work coming within the jurisdiction of the Union, and should therefore be performed by employees who are members of the Union, as specified by Article 2.04 of the Collective Agreement between the parties; and
- (b) the Union agrees that the sale of sundry items which exclude food and beverages does not constitute work coming within its jurisdiction.

NOW THEREFORE IT IS AGREED THAT

1. Commencing on May 5, 1997, the current Collective Agreement between the parties shall be amended by adding to the Employer's List of Departments, the following:

B & G Provision Company

2. Commencing on May 5, 1997, a new classification shall be established in the new Department as follows:

Sales Agent / Server

3. The hourly wage scale for the new classification shall be as follows (March 1, 1997 scale):

| | Entry Level | <u>6 Months</u> | 12 Months |
|--------------------|--------------------|-----------------|-----------|
| Sales Agent/Server | \$ 9.64 | \$11.24 | \$12.85 |

LETTER OF UNDERSTANDING #7 - HYATT REGENCY VANCOUVER (cont'd)

- 4. Effective May 5, 1997, all the terms and conditions of the Collective Agreement shall apply to employees of the new Department.
- 5. The Union acknowledges and agrees that the performance of work falling outside its jurisdiction by employees who will be its members, will not be held to have the effect of transferring into it jurisdiction the sale of sundry items.
- 6. More specifically, if the Employer subsequently finds itself unwilling to continue the direct sale of sundry items using its own employees, the Union agrees that the right that is now possessed by the Employer to contract out the sale of sundry items, shall be retained by the Employer.
- 7. In order to clarify the intentions of the parties in the above regard, the following examples are provided.
 - (a) In the event the Employer decides at a later date to close the lobby-based counter, it shall be entitled to contract out the sale of sundry items to a third party, but the third party shall not be entitled to sell any food or beverage items.
 - (b) In the event the Employer decides at a later date to continue operating the lobby-based counter, but to cease the sale of sundry items, it may contract out the sale of sundry items to a third party, providing the third party does not engage in the sale of food and beverage items.
- 8. In the event the Employer decides at a later date to close the new Department, and such closure occurs prior to May 5, 1998, any employee who had been transferred into it from another department before February 5, 1998, shall have the trial period set out in Article 11.03 of the Agreement, extended to twelve (12) months.

| DATED this day of Province of British Columbia. | , 2007 at the City of Vancouver in the |
|--|--|
| HOSPITALITY INDUSTRIAL RELAT | ΓΙΟΝSUNITE HERE Local 40 |
| Ron Schmidt | Jim Pearson |
| Director | President/Administrator |

LETTER OF UNDERSTANDING #8 between GREATER VANCOUVER HOTEL EMPLOYERS' ASSOCIATION and UNITE HERE Local 40

RENAISSANCE VANCOUVER HOTEL HARBOURSIDE THE SALE OF SUNDRY ITEMS ON THE EMPLOYER'S PREMISES

- 1. The Employer agrees that the sale of food and beverages constitutes work coming within the jurisdiction of the Union, and should therefore be performed by employees who are members of the Union, as specified by Article 2.04 of the Collective Agreement between the parties.
- 2. The Union agrees that the sale of sundry articles which exclude food and beverages, does not constitute work coming within its jurisdiction.
- 3. The Employer now intends to operate a lobby-based kiosk or counter which will amalgamate the sale of food and beverages with the sale of sundries.
 - The Employer's operation will be the successor to those currently independently contracted operations of United Cigar Stores and Bainbridge Daniels, and will commence on May 28, 1994.
- 4. The Employer intends to operate the kiosk/counter using employees who will become members of its bargaining unit, who will receive pay and benefits as agreed to by the two parties, and who will perform a combination of work falling within the union's jurisdiction and work falling outside the Union's jurisdiction.
- 5. Starting May 28, 1994, the current Collective Agreement between the parties shall be

amended by adding to the Employer's List of Departments the following:

Plantations

6. Starting May 28, 1994, two new classifications shall be recognized within the new Department of Plantations as follows:

Server (Plantations) and Head Server (Plantations)

LETTER OF UNDERSTANDING #8

- RENAISSANCE VANCOUVER HOTEL HARBOURSIDE – (cont'd.)

7. Except as expressly provided otherwise, the hourly wage scales for the two new classifications shall be as follows:

| | Entry Level | <u>6 Months</u> | 12 Months |
|-------------|--------------------|-----------------|-----------|
| Server | \$ 6.31 | \$ 7.36 | \$ 8.41 |
| Head Server | 6.78 | 6.91 | 9.04 |

- 8. Notwithstanding the wage rates set forth in Paragraph 7 above, no employee shall be paid less than the minimum wage rate prevailing in B.C. at any given time.
- 9. The five persons employed on May 27, 1994 by Bainbridge Daniels in the area known as *Plantations*, and whose names are listed in the order of their seniority in Appendix 1 attached to this Letter of Understanding, shall become employees of the Employer of May 28, 1994 in the classification of Server (Plantations).
- 10. It is agreed that notwithstanding the provisions of the Paragraph numbered 7 above, the five employees referred to in Paragraph 9 above, shall with effect from May 28, 1994, receive an hourly wage rate of not less than \$7.48 for their next six months.
- 11. Commencing on May 28, 1994, the five employees referred to in Paragraph 9 above, shall start to serve a period of probation as stipulated in Article 7.02 of the Collective Agreement.
- 12. Commencing on May 30, 1994, Kirsten McCormick shall be promoted to the classification of Head Server (Plantations) and, notwithstanding the provisions of Paragraph 7 above, shall receive an hourly wage rate of \$7.91.
- 13. Effective December 1, 1994, each of the five employees referred to in Paragraph 9 above who is still employed by the Employer in either of the two classifications listed in Paragraph above, shall commence receiving the twelve (12) months wage rate for his/her respective classifications.
- 14. Each new employee hired by the Employer to work in the new Plantations Department subsequent to May 27, 1994, shall commence employment at the entry level wage rate, and shall progress through the wage scale in accordance with the provisions of the Collective Agreement.
- 15. Effective May 28, 1994, all of the terms and conditions of the Collective Agreement shall apply to employees of the of the new Plantations Department.

LETTER OF UNDERSTANDING #8 - RENAISSANCE VANCOUVER HOTEL HARBOURSIDE – (cont'd.)

16. The Union acknowledges and agrees that the performance of work falling outside its jurisdiction by employees who will become its members, will not be held to have the effect of transferring into its jurisdiction the sale of sundries. More particularly, in the event the Employer fails to carry through with the intention stated in Paragraph 3 above, or, having carried through with it, finds itself unwilling to continue the direct sale of sundries using its own employees, the Union agrees that the right that is now possessed by the Employer to contract out sundry sales, shall be retained by the Employer. 17. If the Employer subsequently contracts out the sale of sundry articles, no employee of the contractor will be permitted to perform any of the work which it is agreed falls with the jurisdiction of the Union. DATED this ______, 2007 at the City of Vancouver in the Province of British Columbia. **HOSPITALITY INDUSTRIAL RELATIONS UNITE HERE Local 40**

LETTER OF UNDERSTANDING #9

Jim Pearson

President/Administrator

gvhea-local40 111

Ron Schmidt

Director

between GREATER VANCOUVER HOTEL EMPLOYERS' ASSOCIATION and UNITE HERE Local 40

PERFORMANCE OF BARGAINING UNIT WORK BY MANAGEMENT PERSONNEL

This is to confirm the understanding between the Greater Vancouver Hotel Employers' Association and the Hotel, Restaurant and Culinary Employees and Bartenders' Union, Local 40 respecting the application and interpretation of Article 3.03 of the Collective Agreement between the parties (that is, the provision dealing with the performance of bargaining unit work by management personnel.)

An Employer is expected to plan its work schedules for bargaining unit members in a reasonable and sensible manner so that there are sufficient bargaining unit employees scheduled to work to be able to handle the volume of bargaining unit work that is expected to arise during any particular shift.

An Employer should not and cannot establish work schedules for bargaining unit employees that are so unreasonable and unrealistic that the bargaining unit employees who would be working could not be expected to cope with that workload. Based on an Employer's knowledge of, or anticipation of the workload that will arise during any particular shift, sufficient bargaining unit employees will be scheduled to work to be able to deal with that workload.

Furthermore, an Employer is expected to build into its planning process for the establishment of work schedules for bargaining unit employees reasonable consideration for unplanned, unanticipated and unpredictable needs/situations that might arise.

Assuming that the work schedules for bargaining unit employees have been established based on this governing principle (that is, an Employer is expected to adequately and properly schedule and staff bargaining unit employees), management personnel may perform bargaining unit work under the following type of circumstances.

First, during rare and abnormally hectic time periods when there are extraordinary customer needs and all bargaining unit employees are working full hours or are on requested time off.

Second, when, on a particular shift in a particular department, all bargaining unit employees who have been scheduled to work do not show up for work and the Employer is unable to readily replace the absent bargaining unit members.

LETTER OF UNDERSTANDING #9 - PERFORMANCE OF BARGAINING UNIT WORK BY MANAGEMENT PERSONNEL – (cont'd.)

Third, when there is an unexpected/unanticipated/emergent customer service need, and the bargaining unit employees on that particular shift are otherwise occupied/busy and cannot address the customer service need in a timely/efficient manner.

Fourth, when the bargaining unit work that needs to be performed is of a minor nature in terms of the time involved or the actual work to be performed and bargaining unit employees are not otherwise readily available to perform that work.

| DATED this day of Province of British Columbia. | , 2007 at the City of Vancouver in the |
|--|--|
| HOSPITALITY INDUSTRIAL RELAT | TIONSUNITE HERE Local 40 |
| Ron Schmidt Director | Jim Pearson President/Administrator |

FOUR SEASONS HOTEL

| | JULY 1, 2007 | | |
|--|--------------|--------------|--------------|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Department | | | |
| First Bartender | \$15.26 | \$11.45 | \$13.35 |
| Second Bartender | \$15.05 | \$11.29 | \$13.17 |
| First Bartender (Service) | \$19.25 | \$14.44 | \$16.84 |
| Second Bartender (Service | \$19.05 | \$14.29 | \$16.67 |
| Server | \$13.97 | \$10.48 | \$12.22 |
| Fine Dining Room, Room Service, Restaurant | | | |
| Food Captain | \$13.75 | \$10.31 | \$12.03 |
| Host/Hostess | \$13.15 | \$9.86 | \$11.51 |
| Host/Hostess (Food) | \$16.38 | \$12.29 | \$14.33 |
| Server | \$12.23 | \$9.17 | \$10.70 |
| Bus Person | \$14.76 | \$11.07 | \$12.92 |
| Cashier | \$17.22 | \$12.92 | \$15.07 |
| Banquets Department | | | |
| Banquet Captain | \$13.63 | \$10.22 | \$11.93 |
| Banquet Server | \$11.52 | \$8.64 | \$10.08 |
| Banquet Server (Regular) | \$12.36 | \$9.27 | \$10.82 |
| Head Banquet Porter | \$16.44 | \$12.33 | \$14.39 |
| Banquet Porter | \$15.14 | \$11.36 | \$13.25 |
| Private Bar Department | | | |
| Private Bar Supervisor | \$17.67 | \$13.25 | \$15.46 |
| Private Bar Attendant | \$17.67 | \$11.74 | \$13.69 |
| Filvate bai Atteridant | φ15.05 | φ11.74 | φ13.09 |
| Kitchen and Bake Shop Departments | | | |
| 1st Cook (Chef de Partie, Head Breakfast Cook, | | | |
| Saucier, Gardemanger, Grill Cook, | | | |
| Tournant Entremetier) | \$19.14 | \$14.36 | \$16.75 |
| 2nd Cook (General Cook, Breakfast Cook, | | | |
| Experienced Short Order Cook) | \$18.09 | \$13.57 | \$15.83 |
| 3rd Cook (Commis Cuisine, Cafeteria Staff, Short | | | |
| Order Cook, Pantry Person) | \$17.38 | \$13.04 | \$15.21 |
| Cook's Helper (Pastry Helper, Vegetable Cleaner, | | | |
| Apprentice) | \$17.13 | \$12.85 | \$14.99 |
| Butcher | \$17.67 | \$13.25 | \$15.46 |
| Pattissier | \$17.56 | \$13.17 | \$15.37 |
| Stewarding Department | | | |
| Working Steward | \$17.32 | \$12.99 | \$15.16 |
| Kitchen Cleaner (Silver) | \$16.82 | \$12.62 | \$14.72 |
| Dishwasher | \$16.82 | \$12.62 | \$14.72 |
| FOUR OF ADOMO HOTEL | | | |

FOUR SEASONS HOTEL

| | JU | JULY 1, 2007 | | |
|---|--------------|--------------|--------------|--|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos | |
| Front Desk Department | | | | |
| Front Office Supervisor | \$19.03 | \$14.27 | \$16.65 | |
| Receptionist | \$17.65 | \$13.24 | \$15.44 | |
| Communications Department | | | | |
| Communications Supervisor | \$17.56 | \$13.17 | \$15.37 | |
| Switchboard Operator | \$17.06 | \$12.80 | \$14.93 | |
| Concierge Desk Department | | | | |
| Concierge | \$16.63 | \$12.47 | \$14.55 | |
| Bell Captain | \$15.60 | \$11.70 | \$13.65 | |
| Bell Person | \$12.07 | \$9.05 | \$10.56 | |
| Door Person | \$12.75 | \$9.56 | \$11.16 | |
| Housekeeping Department (Including Laundry) | | | | |
| Floor Inspector | \$17.01 | \$12.76 | \$14.88 | |
| Night Supervisor | \$17.50 | \$13.13 | \$15.31 | |
| House Officer/Window | \$17.41 | \$13.06 | \$15.23 | |
| House Person Cleaner | \$17.05 | \$12.79 | \$14.92 | |
| Room Attendant | \$16.95 | \$12.71 | \$14.83 | |
| Valet Supervisor | \$18.68 | \$14.01 | \$16.35 | |
| Uniform Attendant, Valet, Linen, | | | | |
| Room Assistant & Seamstress | \$17.01 | \$12.76 | \$14.88 | |
| Laundry Attendant | \$16.95 | \$12.71 | \$14.83 | |
| Maintenance Department | | | | |
| Maintenance (General) | \$18.89 | \$14.17 | \$16.53 | |
| Maintenance (Major) | \$19.25 | \$14.44 | \$16.85 | |
| Chief Maintenance | \$19.75 | \$14.81 | \$17.28 | |
| Maintenance Wage Adjustment - 10/01/07 | | | | |
| Maintenance (General) | \$19.40 | \$14.55 | \$16.98 | |
| Maintenance (Major) | \$19.75 | \$14.81 | \$17.28 | |
| Chief Maintenance | \$20.25 | \$15.19 | \$17.72 | |
| Stores/Purchasing Department | | | | |
| Storeroom Helper (Inventory, Bar Stock) | \$17.73 | \$13.30 | \$15.51 | |
| Receiver | \$17.90 | \$13.43 | \$15.66 | |
| | | | | |

| FOUR SEASONS HOTEL | | | |
|--|--------------|--------------|--------------|
| | JI | JLY 1, 20 | 08 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Department | | | |
| First Bartender | \$15.57 | \$11.68 | \$13.62 |
| Second Bartender | \$15.35 | \$11.51 | \$13.43 |
| First Bartender (Service) | \$19.64 | \$14.73 | \$17.19 |
| Second Bartender (Service | \$19.43 | \$14.57 | \$17.00 |
| Server | \$14.25 | \$10.69 | \$12.47 |
| Fine Dining Room, Room Service, Restaurant | | | |
| Food Captain | \$14.03 | \$10.52 | \$12.28 |
| Host/Hostess | \$13.41 | \$10.06 | \$11.73 |
| Host/Hostess (Food) | \$16.71 | \$12.53 | \$14.62 |
| Server | \$12.47 | \$9.35 | \$10.91 |
| Bus Person | \$15.06 | \$11.30 | \$13.18 |
| Cashier | \$17.56 | \$13.17 | \$15.37 |
| Banquets Department | | | |
| Banquet Captain | \$13.90 | \$10.43 | \$12.16 |
| Banquet Server | \$11.75 | \$8.81 | \$10.28 |
| Banquet Server (Regular) | \$12.61 | \$9.46 | \$11.03 |
| Head Banquet Porter | \$16.77 | \$12.58 | \$14.67 |
| Banquet Porter | \$15.44 | \$11.58 | \$13.51 |
| Private Bar Department | | | |
| Private Bar Supervisor | \$18.02 | \$13.52 | \$15.77 |
| Private Bar Attendant | \$15.96 | \$11.97 | \$13.97 |
| Kitchen and Bake Shop Departments | | | |
| 1st Cook (Chef de Partie, Head Breakfast Cook, | | | |
| Saucier, Gardemanger, Grill Cook, | | | |
| Tournant Entremetier) | \$19.52 | \$14.64 | \$17.08 |
| 2nd Cook (General Cook, Breakfast Cook, | | | |
| Experienced Short Order Cook) | \$18.45 | \$13.84 | \$16.14 |
| 3rd Cook (Commis Cuisine, Cafeteria Staff, Short | | | |
| Order Cook, Pantry Person) | \$17.73 | \$13.30 | \$15.51 |
| Cook's Helper (Pastry Helper, Vegetable Cleaner, | | | |
| Apprentice) | \$17.47 | \$13.10 | \$15.29 |
| Butcher | \$18.02 | \$13.52 | \$15.77 |
| Pattissier | \$17.91 | \$13.43 | \$15.67 |
| Stewarding Department | | | |
| Working Steward | \$17.67 | \$13.25 | \$15.46 |
| Kitchen Cleaner (Silver) | \$17.16 | \$12.87 | \$15.02 |
| Dishwasher | \$17.16 | \$12.87 | \$15.02 |
| | | | |

| FOUR SEASONS HOTEL | | | |
|---|--------------|--------------|--------------|
| | JI | ULY 1, 20 | 08 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Front Desk Department | | | |
| Front Office Supervisor | \$19.41 | \$14.56 | \$16.98 |
| Receptionist | \$18.00 | \$13.50 | \$15.75 |
| Communications Department | | | |
| Communications Supervisor | \$17.91 | \$13.43 | \$15.67 |
| Switchboard Operator | \$17.40 | \$13.05 | \$15.23 |
| Concierge Desk Department | | | |
| Concierge | \$16.96 | \$12.72 | \$14.84 |
| Bell Captain | \$15.91 | \$11.93 | \$13.92 |
| Bell Person | \$12.31 | \$9.23 | \$10.77 |
| Door Person | \$13.01 | \$9.76 | \$11.38 |
| Housekeeping Department (Including Laundry) | | | |
| Floor Inspector | \$17.35 | \$13.01 | \$15.18 |
| Night Supervisor | \$17.85 | \$13.39 | \$15.62 |
| House Officer/Window | \$17.76 | \$13.32 | \$15.54 |
| House Person Cleaner | \$17.39 | \$13.04 | \$15.22 |
| Room Attendant | \$17.29 | \$12.97 | \$15.13 |
| Valet Supervisor | \$19.05 | \$14.29 | \$16.67 |
| Uniform Attendant, Valet, Linen, | | | |
| Room Assistant & Seamstress | \$17.35 | \$13.01 | \$15.18 |
| Laundry Attendant | \$17.29 | \$12.97 | \$15.13 |
| Maintenance Department | | | |
| Maintenance (General) | \$19.79 | \$14.84 | \$17.32 |
| Maintenance (Major) | \$20.15 | \$15.11 | \$17.63 |
| Chief Maintenance | \$20.66 | \$15.50 | \$18.08 |
| Stores/Purchasing Department | | | |
| Storeroom Helper (Inventory, Bar Stock) | \$18.08 | \$13.56 | \$15.82 |
| Receiver | \$18.26 | \$13.70 | \$15.98 |

| FOUR SEASONS HOTEL | | | |
|--|--------------|--------------|--------------|
| | JAN | UARY 1, | 2009 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Department | | | |
| First Bartender | \$15.88 | \$11.91 | \$13.90 |
| Second Bartender | \$15.66 | \$11.75 | \$13.70 |
| First Bartender (Service) | \$20.03 | \$15.02 | \$17.53 |
| Second Bartender (Service | \$19.82 | \$14.87 | \$17.34 |
| Server | \$14.54 | \$10.91 | \$12.72 |
| Fine Dining Room, Room Service, Restaurant | | | |
| Food Captain | \$14.31 | \$10.73 | \$12.52 |
| Host/Hostess | \$13.68 | \$10.26 | \$11.97 |
| Host/Hostess (Food) | \$17.04 | \$12.78 | \$14.91 |
| Server | \$12.72 | \$9.54 | \$11.13 |
| Bus Person | \$15.36 | \$11.52 | \$13.44 |
| Cashier | \$17.91 | \$13.43 | \$15.67 |
| | | | |
| Banquets Department | | | |
| Banquet Captain | \$14.18 | \$10.64 | \$12.41 |
| Banquet Server | \$11.99 | \$8.99 | \$10.49 |
| Banquet Server (Regular) | \$12.86 | \$9.65 | \$11.25 |
| Head Banquet Porter | \$17.11 | \$12.83 | \$14.97 |
| Banquet Porter | \$15.75 | \$11.81 | \$13.78 |
| Private Bar Department | | | |
| Private Bar Supervisor | \$18.38 | \$13.79 | \$16.08 |
| Private Bar Attendant | \$16.28 | \$12.21 | \$14.25 |
| Kitchen and Bake Shop Departments | | | |
| 1st Cook (Chef de Partie, Head Breakfast Cook, | | | |
| Saucier, Gardemanger, Grill Cook, | | | |
| Tournant Entremetier) | \$19.91 | \$14.93 | \$17.42 |
| 2nd Cook (General Cook, Breakfast Cook, | | | |
| Experienced Short Order Cook) | \$18.82 | \$14.12 | \$16.47 |
| 3rd Cook (Commis Cuisine, Cafeteria Staff, Short | | | |
| Order Cook, Pantry Person) | \$18.08 | \$13.56 | \$15.82 |
| Cook's Helper (Pastry Helper, Vegetable Cleaner, | | | |
| Apprentice) | \$17.82 | \$13.37 | \$15.59 |
| Butcher | \$18.38 | \$13.79 | \$16.08 |
| Pattissier | \$18.27 | \$13.70 | \$15.99 |
| Stewarding Department | | | |
| Working Steward | \$18.02 | \$13.52 | \$15.77 |
| Kitchen Cleaner (Silver) | \$17.50 | \$13.13 | \$15.77 |
| Dishwasher | \$17.50 | \$13.13 | \$15.31 |
| 2.540.101 | ψ17.50 | ψ10.10 | ψ10.01 |

| FOUR SEASONS HOTEL | | | |
|---|--------------|--------------|--------------|
| | JAN | UARY 1, | 2009 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Front Desk Department | | | |
| Front Office Supervisor | \$19.80 | \$14.85 | \$17.33 |
| Receptionist | \$18.36 | \$13.77 | \$16.07 |
| Communications Department | | | |
| Communications Supervisor | \$18.27 | \$13.70 | \$15.99 |
| Switchboard Operator | \$17.75 | \$13.31 | \$15.53 |
| Concierge Desk Department | | | |
| Concierge | \$17.30 | \$12.98 | \$15.14 |
| Bell Captain | \$16.23 | \$12.17 | \$14.20 |
| Bell Person | \$12.56 | \$9.42 | \$10.99 |
| Door Person | \$13.27 | \$9.95 | \$11.61 |
| Housekeeping Department (Including Laundry) | | | |
| Floor Inspector | \$17.70 | \$13.28 | \$15.49 |
| Night Supervisor | \$18.21 | \$13.66 | \$15.93 |
| House Officer/Window | \$18.12 | \$13.59 | \$15.86 |
| House Person Cleaner | \$17.74 | \$13.31 | \$15.52 |
| Room Attendant | \$17.64 | \$13.23 | \$15.44 |
| Valet Supervisor | \$19.43 | \$14.57 | \$17.00 |
| Uniform Attendant, Valet, Linen, | | | |
| Room Assistant & Seamstress | \$17.70 | \$13.28 | \$15.49 |
| Laundry Attendant | \$17.64 | \$13.23 | \$15.44 |
| Maintenance Department | | | |
| Maintenance (General) | \$20.19 | \$15.14 | \$17.67 |
| Maintenance (Major) | \$20.55 | \$15.41 | \$17.98 |
| Chief Maintenance | \$21.07 | \$15.80 | \$18.44 |
| Stores/Purchasing Department | | | |
| Storeroom Helper (Inventory, Bar Stock) | \$18.44 | \$13.83 | \$16.14 |
| Receiver | \$18.63 | \$13.97 | \$16.30 |

| FOUR SEASONS HOTEL | | | |
|--|-----------------|----------------|----------------|
| | JI | JLY 1, 20 | 09 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Department | | | |
| First Bartender | \$16.20 | \$12.15 | \$14.18 |
| Second Bartender | \$15.97 | \$11.98 | \$13.97 |
| First Bartender (Service) | \$20.43 | \$15.32 | \$17.88 |
| Second Bartender (Service | \$20.22 | \$15.17 | \$17.69 |
| Server | \$14.83 | \$11.12 | \$12.98 |
| Fine Dining Room, Room Service, Restaurant | | | |
| Food Captain | \$14.60 | \$10.95 | \$12.78 |
| Host/Hostess | \$13.95 | \$10.46 | \$12.21 |
| Host/Hostess (Food) | \$17.38 | \$13.04 | \$15.21 |
| Server | \$12.97 | \$9.73 | \$11.35 |
| Bus Person | \$15.67 | \$11.75 | \$13.71 |
| Cashier | \$18.27 | \$13.70 | \$15.99 |
| Parameter Parameter | | | |
| Banquets Department | * 4.4.40 | # 40.05 | # 40.05 |
| Banquet Captain | \$14.46 | \$10.85 | \$12.65 |
| Banquet Server | \$12.23 | · | \$10.70 |
| Banquet Server (Regular) | \$13.12 | | \$11.48 |
| Head Banquet Porter | \$17.45 | | \$15.27 |
| Banquet Porter | \$16.07 | \$12.05 | \$14.06 |
| Private Bar Department | | | |
| Private Bar Supervisor | \$18.75 | \$14.06 | \$16.41 |
| Private Bar Attendant | \$16.61 | \$12.46 | \$14.53 |
| Kitchen and Bake Shop Departments | | | |
| 1st Cook (Chef de Partie, Head Breakfast Cook, | | | |
| Saucier, Gardemanger, Grill Cook, | | | |
| Tournant Entremetier) | \$20.31 | \$15.23 | \$17.77 |
| 2nd Cook (General Cook, Breakfast Cook, | | | |
| Experienced Short Order Cook) | \$19.20 | \$14.40 | \$16.80 |
| 3rd Cook (Commis Cuisine, Cafeteria Staff, Short | | | |
| Order Cook, Pantry Person) | \$18.44 | \$13.83 | \$16.14 |
| Cook's Helper (Pastry Helper, Vegetable Cleaner, | | | |
| Apprentice) | \$18.18 | \$13.64 | \$15.91 |
| Butcher | \$18.75 | \$14.06 | \$16.41 |
| Pattissier | \$18.64 | \$13.98 | \$16.31 |
| Stewarding Department | | | |
| Working Steward | \$18.38 | \$13.79 | \$16.08 |
| Kitchen Cleaner (Silver) | \$17.85 | \$13.79 | \$15.62 |
| Dishwasher | \$17.85 | \$13.39 | \$15.62 |
| DISTINGUITE | ψ17.03 | ψ13.33 | ψ13.02 |

| FOUR SEASONS HOTEL | | | |
|---|--------------|--------------|--------------|
| | Jl | JLY 1, 200 | 09 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Front Desk Department | | | |
| Front Office Supervisor | \$20.20 | \$15.15 | \$17.68 |
| Receptionist | \$18.73 | \$14.05 | \$16.39 |
| Communications Department | | | |
| Communications Supervisor | \$18.64 | \$13.98 | \$16.31 |
| Switchboard Operator | \$18.11 | \$13.58 | \$15.85 |
| Concierge Desk Department | | | |
| Concierge | \$17.65 | \$13.24 | \$15.44 |
| Bell Captain | \$16.55 | \$12.41 | \$14.48 |
| Bell Person | \$12.81 | \$9.61 | \$11.21 |
| Door Person | \$13.54 | \$10.16 | \$11.85 |
| Housekeeping Department (Including Laundry) | | | |
| Floor Inspector | \$18.05 | \$13.54 | \$15.79 |
| Night Supervisor | \$18.57 | \$13.93 | \$16.25 |
| House Officer/Window | \$18.48 | \$13.86 | \$16.17 |
| House Person Cleaner | \$18.09 | \$13.57 | \$15.83 |
| Room Attendant | \$17.99 | \$13.49 | \$15.74 |
| Valet Supervisor | \$19.82 | \$14.87 | \$17.34 |
| Uniform Attendant, Valet, Linen, | | | |
| Room Assistant & Seamstress | \$18.05 | \$13.54 | \$15.79 |
| Laundry Attendant | \$17.99 | \$13.49 | \$15.74 |
| Maintenance Department | | | |
| Maintenance (General) | \$20.59 | \$15.44 | \$18.02 |
| Maintenance (Major) | \$20.96 | \$15.72 | \$18.34 |
| Chief Maintenance | \$21.49 | \$16.12 | \$18.80 |
| Stores/Purchasing Department | | | |
| Storeroom Helper (Inventory, Bar Stock) | \$18.81 | \$14.11 | \$16.46 |
| Receiver | \$19.00 | \$14.25 | \$16.63 |

| FOUR SEASONS HOTEL | | | |
|--|-----------------|--------------|--------------|
| | JANUARY 1, 2010 | | 2010 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Department | | | |
| First Bartender | \$16.52 | \$12.39 | \$14.46 |
| Second Bartender | \$16.29 | \$12.22 | \$14.25 |
| First Bartender (Service) | \$20.84 | \$15.63 | \$18.24 |
| Second Bartender (Service | \$20.62 | \$15.47 | \$18.04 |
| Server | \$15.13 | \$11.35 | \$13.24 |
| Fine Dining Room, Room Service, Restaurant | | | |
| Food Captain | \$14.89 | \$11.17 | \$13.03 |
| Host/Hostess | \$14.23 | \$10.67 | \$12.45 |
| Host/Hostess (Food) | \$17.73 | \$13.30 | \$15.51 |
| Server | \$13.23 | \$9.92 | \$11.58 |
| Bus Person | \$15.98 | \$11.99 | \$13.98 |
| Cashier | \$18.64 | \$13.98 | \$16.31 |
| Banquets Department | | | |
| Banquet Captain | \$14.75 | \$11.06 | \$12.91 |
| Banquet Server | \$12.47 | \$9.35 | \$10.91 |
| Banquet Server (Regular) | \$13.38 | \$10.04 | \$11.71 |
| Head Banquet Porter | \$17.80 | \$13.35 | \$15.58 |
| Banquet Porter | \$16.39 | \$12.29 | \$14.34 |
| Private Bar Department | | | |
| Private Bar Supervisor | \$19.13 | \$14.35 | \$16.74 |
| Private Bar Attendant | \$16.94 | \$12.71 | \$14.82 |
| Kitchen and Bake Shop Departments | | | |
| 1st Cook (Chef de Partie, Head Breakfast Cook, | | | |
| Saucier, Gardemanger, Grill Cook, | | | |
| Tournant Entremetier) | \$20.72 | \$15.54 | \$18.13 |
| 2nd Cook (General Cook, Breakfast Cook, | | | |
| Experienced Short Order Cook) | \$19.58 | \$14.69 | \$17.13 |
| 3rd Cook (Commis Cuisine, Cafeteria Staff, Short | | | |
| Order Cook, Pantry Person) | \$18.81 | \$14.11 | \$16.46 |
| Cook's Helper (Pastry Helper, Vegetable Cleaner, | | | |
| Apprentice) | \$18.54 | \$13.91 | \$16.22 |
| Butcher | \$19.13 | \$14.35 | \$16.74 |
| Pattissier | \$19.01 | \$14.26 | \$16.63 |
| Stewarding Department | | | |
| Working Steward | \$18.75 | \$14.06 | \$16.41 |
| Kitchen Cleaner (Silver) | \$18.21 | \$13.66 | \$15.93 |
| Dishwasher | \$18.21 | \$13.66 | \$15.93 |

| FOUR SEASONS HOTEL | | | |
|---|-----------------|--------------|--------------|
| | JANUARY 1, 2010 | | |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Front Desk Department | | | |
| Front Office Supervisor | \$20.60 | \$15.45 | \$18.03 |
| Receptionist | \$19.10 | \$14.33 | \$16.71 |
| Communications Department | | | |
| Communications Supervisor | \$19.01 | \$14.26 | \$16.63 |
| Switchboard Operator | \$18.47 | \$13.85 | \$16.16 |
| Concierge Desk Department | | | |
| Concierge | \$18.00 | \$13.50 | \$15.75 |
| Bell Captain | \$16.88 | \$12.66 | \$14.77 |
| Bell Person | \$13.07 | \$9.80 | \$11.44 |
| Door Person | \$13.81 | \$10.36 | \$12.08 |
| Housekeeping Department (Including Laundry) | | | |
| Floor Inspector | \$18.41 | \$13.81 | \$16.11 |
| Night Supervisor | \$18.94 | \$14.21 | \$16.57 |
| House Officer/Window | \$18.85 | \$14.14 | \$16.49 |
| House Person Cleaner | \$18.45 | \$13.84 | \$16.14 |
| Room Attendant | \$18.35 | \$13.76 | \$16.06 |
| Valet Supervisor | \$20.22 | \$15.17 | \$17.69 |
| Uniform Attendant, Valet, Linen, | | | |
| Room Assistant & Seamstress | \$18.41 | \$13.81 | \$16.11 |
| Laundry Attendant | \$18.35 | \$13.76 | \$16.06 |
| Maintenance Department | | | |
| Maintenance (General) | \$21.00 | \$15.75 | \$18.38 |
| Maintenance (Major) | \$21.38 | \$16.04 | \$18.71 |
| Chief Maintenance | \$21.92 | \$16.44 | \$19.18 |
| Stores/Purchasing Department | | | |
| Storeroom Helper (Inventory, Bar Stock) | \$19.19 | \$14.39 | \$16.79 |
| Receiver | \$19.38 | \$14.54 | \$16.96 |

| HYATT REGENCY VANCOUVER | | | |
|-----------------------------------|--------------|--------------|--------------|
| | JULY 1, 2007 | | 07 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Beverage Department | | | |
| First Mixerologist (Service) | \$19.45 | \$14.59 | \$17.02 |
| Second Mixerologist (Service) | \$19.23 | \$14.42 | \$16.83 |
| First Mixerologist | \$15.42 | \$11.57 | \$13.49 |
| Second Mixerologist | \$15.23 | \$11.42 | \$13.33 |
| Floorperson | \$18.23 | \$13.67 | \$15.95 |
| Sandwich/Grill | \$17.16 | \$12.87 | \$15.02 |
| Bar Porter - Service Bar | \$16.60 | \$12.45 | \$14.53 |
| Waitperson | \$13.73 | \$10.30 | \$12.01 |
| Food Waitperson | \$12.48 | \$9.36 | \$10.92 |
| Accounting/Night Audit Department | | | |
| Head Night Auditor | \$18.16 | \$13.62 | \$15.89 |
| Night Auditor | \$17.72 | \$13.29 | • |
| Tught / dollor | ψ2 | Ψ10.20 | ψ10.01 |
| Banquets Department | | | |
| Cashier | \$16.50 | \$12.38 | \$14.44 |
| Head Houseperson | \$16.53 | \$12.40 | \$14.46 |
| Houseperson | \$15.32 | \$11.49 | \$13.41 |
| Waitperson | \$12.20 | \$9.15 | \$10.68 |
| Bell/Door Service Department | | | |
| Head Bellperson | \$12.96 | \$9.72 | \$11.34 |
| Bellperson | \$12.27 | | • |
| Doorperson | \$12.27 | \$9.20 | \$10.74 |
| | Ť | , - | • |
| Dining Room - Mosaic Bar & Grille | | | |
| Host/Hostess | \$15.79 | \$11.84 | \$13.82 |
| Mixerologist/Sommelier | \$15.10 | \$11.33 | \$13.21 |
| Busperson | \$14.68 | \$11.01 | \$12.85 |
| Bar Waitperson | \$12.97 | \$9.73 | \$11.35 |
| Waitperson | \$12.20 | \$9.15 | \$10.68 |
| Engineering Department | | | |
| Maintenance (Electric) | \$21.77 | \$16.33 | \$19.05 |
| Maintenance (Plumber) | \$20.76 | \$15.57 | \$18.17 |
| Maintenance (Woodwork) | \$19.98 | \$14.99 | \$17.48 |
| Maintenance (Minor Repair) | \$18.97 | \$14.23 | \$16.60 |
| General Maintenance | \$18.20 | \$13.65 | \$15.93 |
| | | | , |
| | | | |

| | |
|---------|-----------|
| DECENCY | VANCOUVER |
| | |

| | JULY 1, 2007 | | 07 |
|--|--------------|--------------|--------------|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Engineering Wage Adjustment - 10/01/07 | | | |
| Maintenance (Electric) | \$22.27 | \$16.70 | \$19.49 |
| Maintenance (Plumber) | \$21.26 | \$15.95 | \$18.60 |
| Maintenance (Woodwork) | \$20.48 | \$15.36 | \$17.92 |
| Maintenance (Minor Repair) | \$19.47 | \$14.60 | \$17.04 |
| General Maintenance | \$18.70 | \$14.03 | \$16.36 |
| Front Office Department | | | |
| Night Clerk | \$17.72 | \$13.29 | \$15.51 |
| Senior Desk Clerk | \$17.56 | \$13.17 | \$15.37 |
| Front Desk Clerk | \$16.91 | \$12.68 | \$14.80 |
| Housekeeping Department | | | |
| Inspector/Inspectress | \$17.14 | \$12.86 | \$15.00 |
| Tailor/Seamstress | \$17.05 | \$12.79 | \$14.92 |
| Housekeeping House Attendant | \$17.01 | \$12.76 | \$14.88 |
| Housekeeper | \$16.74 | \$12.56 | \$14.65 |
| Linen Attendant | \$16.74 | \$12.56 | \$14.65 |
| Laundry Attendant | \$16.66 | \$12.50 | \$14.58 |
| Room Attendant | \$16.54 | \$12.41 | \$14.47 |
| Kitchen Department | | | |
| First Cook | \$19.54 | \$14.66 | \$17.10 |
| Second Cook | \$18.64 | \$13.98 | \$16.31 |
| Third Cook | \$17.66 | \$13.25 | \$15.45 |
| Pastry Cook | \$17.30 | \$12.98 | \$15.14 |
| Salad/Sandwich | \$17.06 | \$12.80 | \$14.93 |
| Cook's Helper | \$16.86 | \$12.65 | \$14.75 |
| PBX/Switchboard Department | | | |
| Chief Switchboard Operator | \$16.98 | \$12.74 | |
| Switchboard Operator | \$16.62 | \$12.47 | \$14.54 |
| Room Service Department | | | |
| Executive Bar Attendant | \$16.60 | \$12.45 | \$14.53 |
| Cashier | \$16.50 | \$12.38 | \$14.44 |
| Captain | \$13.36 | \$10.02 | \$11.69 |
| Waitperson | \$12.20 | \$9.15 | \$10.68 |
| Stewards Department | | | |
| Janitor | \$17.01 | \$12.76 | \$14.88 |
| Steward | \$16.91 | \$12.68 | \$14.80 |
| Dishwasher | \$16.50 | \$12.38 | \$14.44 |
| | | | |

| | | VANCOUVER | |
|--|--|-----------|--|
| | | | |
| | | | |

JULY 1, 2007

| | JULY 1, 2007 | | | |
|--------------------------------|--------------|--------------|--------------|--|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos | |
| Storeroom/Receiving Department | | | | |
| Head Storekeeper | \$18.22 | \$13.67 | \$15.94 | |
| Storekeeper | \$17.46 | \$13.10 | \$15.28 | |
| | | | | |
| Latte Café & Bistro | | | | |
| Bistro Supervisor | \$16.05 | \$12.04 | \$14.04 | |
| Bistro Server | \$14.97 | \$11.23 | \$13.10 | |

| HYATT REGENCY VANCOUVER | | | |
|-----------------------------------|--------------|--------------|--------------|
| | JULY 1, 2008 | | 08 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Beverage Department | | | |
| First Mixerologist (Service) | \$19.84 | \$14.88 | \$17.36 |
| Second Mixerologist (Service) | \$19.61 | | \$17.16 |
| First Mixerologist | \$15.73 | | \$13.76 |
| Second Mixerologist | \$15.53 | | \$13.59 |
| Floorperson | \$18.59 | | \$16.27 |
| Sandwich/Grill | \$17.50 | | \$15.31 |
| Bar Porter - Service Bar | \$16.93 | \$12.70 | \$14.81 |
| Waitperson | \$14.00 | | \$12.25 |
| Food Waitperson | \$12.73 | \$9.55 | \$11.14 |
| Accounting/Night Audit Department | | | |
| Head Night Auditor | \$18.52 | \$13.89 | \$16.21 |
| Night Auditor | \$18.07 | \$13.55 | \$15.81 |
| Banquets Department | | | |
| Cashier | \$16.83 | \$12.62 | \$14.73 |
| Head Houseperson | \$16.86 | | \$14.75 |
| Houseperson | \$15.63 | | \$13.68 |
| Waitperson | \$12.44 | \$9.33 | \$10.89 |
| Bell/Door Service Department | | | |
| Head Bellperson | \$13.22 | \$9.92 | \$11.57 |
| Bellperson | \$12.52 | | \$10.96 |
| Doorperson | \$12.52 | | \$10.96 |
| Dining Room - Mosaic Bar & Grille | | | |
| Host/Hostess | \$16.11 | \$12.08 | \$14.10 |
| Mixerologist/Sommelier | \$15.40 | \$11.55 | \$13.48 |
| Busperson | \$14.97 | \$11.23 | \$13.10 |
| Bar Waitperson | \$13.23 | \$9.92 | \$11.58 |
| Waitperson | \$12.44 | \$9.33 | \$10.89 |
| Engineering Department | | | |
| Maintenance (Electric) | \$22.72 | \$17.04 | \$19.88 |
| Maintenance (Plumber) | \$21.69 | \$16.27 | \$18.98 |
| Maintenance (Woodwork) | \$20.89 | \$15.67 | \$18.28 |
| Maintenance (Minor Repair) | \$19.86 | \$14.90 | \$17.38 |
| General Maintenance | \$19.07 | \$14.30 | \$16.69 |
| Front Office Department | | | |
| Night Clerk | \$18.07 | \$13.55 | \$15.81 |
| Senior Desk Clerk | \$17.91 | \$13.43 | \$15.67 |
| Front Desk Clerk | \$17.25 | \$12.94 | \$15.09 |
| | | | |

| ı | uv | ATT | DEC | ENCV | V/AN | COUVER | , |
|---|----|--------------|-----|--------|------|--------|---|
| - | пτ | \mathbf{A} | RFG | FINE.T | VAN | | |

| | JULY 1, 2008 | | |
|--------------------------------|--------------|--------------|--------------|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Housekeeping Department | | | |
| Inspector/Inspectress | \$17.48 | \$13.11 | \$15.30 |
| Tailor/Seamstress | \$17.39 | | \$15.22 |
| Housekeeping House Attendant | \$17.35 | \$13.01 | \$15.18 |
| Housekeeper | \$17.07 | \$12.80 | \$14.94 |
| Linen Attendant | \$17.07 | \$12.80 | \$14.94 |
| Laundry Attendant | \$16.99 | \$12.74 | \$14.87 |
| Room Attendant | \$16.87 | \$12.65 | \$14.76 |
| Kitchen Department | | | |
| First Cook | \$19.93 | \$14.95 | \$17.44 |
| Second Cook | \$19.01 | \$14.26 | \$16.63 |
| Third Cook | \$18.01 | \$13.51 | \$15.76 |
| Pastry Cook | \$17.65 | \$13.24 | \$15.44 |
| Salad/Sandwich | \$17.40 | \$13.05 | \$15.23 |
| Cook's Helper | \$17.20 | \$12.90 | \$15.05 |
| PBX/Switchboard Department | | | |
| Chief Switchboard Operator | \$17.32 | \$12.99 | \$15.16 |
| Switchboard Operator | \$16.95 | \$12.71 | \$14.83 |
| Room Service Department | | | |
| Executive Bar Attendant | \$16.93 | \$12.70 | \$14.81 |
| Cashier | \$16.83 | \$12.62 | \$14.73 |
| Captain | \$13.63 | \$10.22 | \$11.93 |
| Waitperson | \$12.44 | \$9.33 | \$10.89 |
| Stewards Department | | | |
| Janitor | \$17.35 | \$13.01 | \$15.18 |
| Steward | \$17.25 | \$12.94 | \$15.09 |
| Dishwasher | \$16.83 | \$12.62 | \$14.73 |
| Storeroom/Receiving Department | | | |
| Head Storekeeper | \$18.58 | \$13.94 | \$16.26 |
| Storekeeper | \$17.81 | \$13.36 | \$15.58 |
| Latte Café & Bistro | | | |
| Bistro Supervisor | \$16.37 | \$12.28 | \$14.32 |
| Bistro Server | \$15.27 | \$11.45 | \$13.36 |

| HYATT REGENCY VANCOUVER | | | |
|-----------------------------------|--------------|-----------------|--------------|
| | JAN | JANUARY 1, 2009 | |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Beverage Department | | | |
| First Mixerologist (Service) | \$20.24 | \$15.18 | \$17.71 |
| Second Mixerologist (Service) | \$20.00 | \$15.00 | \$17.50 |
| First Mixerologist | \$16.04 | \$12.03 | \$14.04 |
| Second Mixerologist | \$15.84 | \$11.88 | \$13.86 |
| Floorperson | \$18.96 | \$14.22 | \$16.59 |
| Sandwich/Grill | \$17.85 | \$13.39 | \$15.62 |
| Bar Porter - Service Bar | \$17.27 | \$12.95 | \$15.11 |
| Waitperson | \$14.28 | \$10.71 | \$12.50 |
| Food Waitperson | \$12.98 | \$9.74 | \$11.36 |
| Accounting/Night Audit Department | | | |
| Head Night Auditor | \$18.89 | \$14.17 | \$16.53 |
| Night Auditor | \$18.43 | \$13.82 | \$16.13 |
| Banquets Department | | | |
| Cashier | \$17.17 | \$12.88 | \$15.02 |
| Head Houseperson | \$17.20 | \$12.90 | \$15.05 |
| Houseperson | \$15.94 | \$11.96 | \$13.95 |
| Waitperson | \$12.69 | \$9.52 | \$11.10 |
| Bell/Door Service Department | | | |
| Head Bellperson | \$13.48 | \$10.11 | \$11.80 |
| Bellperson | \$12.77 | \$9.58 | \$11.17 |
| Doorperson | \$12.77 | \$9.58 | \$11.17 |
| Dining Room - Mosaic Bar & Grille | | | |
| Host/Hostess | \$16.43 | • | \$14.38 |
| Mixerologist/Sommelier | \$15.71 | | \$13.75 |
| Busperson | \$15.27 | | |
| Bar Waitperson | \$13.49 | \$10.12 | \$11.80 |
| Waitperson | \$12.69 | \$9.52 | \$11.10 |
| Engineering Department | | | |
| Maintenance (Electric) | \$23.17 | \$17.38 | \$20.27 |
| Maintenance (Plumber) | \$22.12 | \$16.59 | \$19.36 |
| Maintenance (Woodwork) | \$21.31 | \$15.98 | \$18.65 |
| Maintenance (Minor Repair) | \$20.26 | \$15.20 | \$17.73 |
| General Maintenance | \$19.45 | \$14.59 | \$17.02 |
| Front Office Department | | | |
| Night Clerk | \$18.43 | \$13.82 | \$16.13 |
| Senior Desk Clerk | \$18.27 | \$13.70 | \$15.99 |
| Front Desk Clerk | \$17.60 | \$13.20 | \$15.40 |
| | | | |

| HYATT REGENCY VANCOUVER | | | |
|--------------------------------|--------------|-----------------|--------------|
| | JAN | JANUARY 1, 2009 | |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Housekeeping Department | | | |
| Inspector/Inspectress | \$17.83 | \$13.37 | \$15.60 |
| Tailor/Seamstress | \$17.74 | \$13.31 | \$15.52 |
| Housekeeping House Attendant | \$17.70 | \$13.28 | \$15.49 |
| Housekeeper | \$17.41 | \$13.06 | \$15.23 |
| Linen Attendant | \$17.41 | \$13.06 | \$15.23 |
| Laundry Attendant | \$17.33 | \$13.00 | \$15.16 |
| Room Attendant | \$17.21 | \$12.91 | \$15.06 |
| Kitchen Department | | | |
| First Cook | \$20.33 | \$15.25 | \$17.79 |
| Second Cook | \$19.39 | \$14.54 | \$16.97 |
| Third Cook | \$18.37 | \$13.78 | \$16.07 |
| Pastry Cook | \$18.00 | \$13.50 | \$15.75 |
| Salad/Sandwich | \$17.75 | \$13.31 | \$15.53 |
| Cook's Helper | \$17.54 | \$13.16 | \$15.35 |
| PBX/Switchboard Department | | | |
| Chief Switchboard Operator | \$17.67 | \$13.25 | \$15.46 |
| Switchboard Operator | \$17.29 | \$12.97 | \$15.13 |
| Room Service Department | | | |
| Executive Bar Attendant | \$17.27 | \$12.95 | \$15.11 |
| Cashier | \$17.17 | \$12.88 | \$15.02 |
| Captain | \$13.90 | \$10.43 | \$12.16 |
| Waitperson | \$12.69 | \$9.52 | \$11.10 |
| Stewards Department | | | |
| Janitor | \$17.70 | \$13.28 | \$15.49 |
| Steward | \$17.60 | \$13.20 | \$15.40 |
| Dishwasher | \$17.17 | \$12.88 | \$15.02 |
| Storeroom/Receiving Department | | | |
| Head Storekeeper | \$18.95 | \$14.21 | \$16.58 |
| Storekeeper | \$18.17 | \$13.63 | \$15.90 |
| Latte Café & Bistro | | | |
| Bistro Supervisor | \$16.70 | \$12.53 | \$14.61 |
| Bistro Server | \$15.58 | \$11.69 | \$13.63 |
| | | | |

| Pull | HYATT REGENCY VANCOUVER | | | |
|---|---------------------------------------|---------------|---------|---------|
| Rate 6 Mos | | JULY 1, 2009 | | 09 |
| Second Mixerologist (Service) \$20.64 \$15.48 \$18.06 Second Mixerologist (Service) \$20.40 \$15.30 \$17.85 \$16.36 \$12.27 \$14.32 \$16.36 \$12.27 \$14.32 \$16.36 \$12.27 \$14.43 \$16.92 \$16.16 \$12.12 \$14.14 \$16.92 \$18.20 \$18.21 \$13.66 \$15.93 \$18.40 \$18.21 \$13.66 \$15.93 \$18.40 \$18.21 \$13.66 \$15.93 \$18.70 \$13.22 \$15.42 \$13.22 \$15.42 \$13.22 \$15.42 \$13.24 \$19.93 \$12.75 \$10.93 \$12.75 \$10.93 \$12.75 \$10.94 \$13.24 \$10.93 \$12.75 \$10.94 \$19.27 \$14.45 \$16.86 \$18.80 \$14.10 \$16.45 \$18.80 | | | | |
| Second Mixerologist (Service) \$20.40 | Beverage Department | | | |
| Sinst Mixerologist | First Mixerologist (Service) | \$20.64 | \$15.48 | \$18.06 |
| Second Mixerologist | Second Mixerologist (Service) | \$20.40 | \$15.30 | \$17.85 |
| Sandwich/Grill | First Mixerologist | \$16.36 | \$12.27 | \$14.32 |
| Sandwich/Grill \$18.21 \$13.66 \$15.93 Bar Porter - Service Bar \$17.62 \$13.22 \$15.42 Waitperson \$14.57 \$10.93 \$12.75 Food Waitperson \$13.24 \$9.93 \$11.59 Accounting/Night Audit Department Head Night Auditor \$19.27 \$14.45 \$16.86 Night Auditor \$18.80 \$14.10 \$16.45 Banquets Department Cashier \$17.51 \$13.13 \$15.32 Head Houseperson \$17.54 \$13.16 \$15.35 Houseperson \$16.26 \$12.20 \$14.23 Waitperson \$13.75 \$10.31 \$12.03 Bell/Door Service Department Head Bellperson \$13.03 \$9.77 \$11.40 Dorperson \$13.03 \$9.77 \$11.40 Dining Room - Mosaic Bar & Grille Host/Hostess \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 | Second Mixerologist | \$16.16 | \$12.12 | \$14.14 |
| Bar Porter - Service Bar \$17.62 \$13.22 \$15.42 Waitperson \$14.57 \$10.93 \$12.75 Food Waitperson \$13.24 \$9.93 \$11.59 Accounting/Night Audit Department Head Night Auditor \$19.27 \$14.45 \$16.86 Night Auditor \$18.80 \$14.10 \$16.45 Banquets Department Cashier \$17.51 \$13.13 \$15.32 Head Houseperson \$16.26 \$12.20 \$14.23 Waitperson \$16.26 \$12.20 \$14.23 Waitperson \$13.75 \$10.31 \$12.03 Bell/Door Service Department \$13.03 \$9.77 \$11.40 Department Head Bellperson \$13.03 \$9.77 \$11.40 Dorperson \$13.03 \$9.77 \$11.40 Dining Room - Mosaic Bar & Grille Host/Hostess \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$15.58 \$11.69 \$13.63 Bar Waitperson \$15.58 \$11.69 | Floorperson | \$19.34 | \$14.51 | \$16.92 |
| Waitperson \$14.57 \$10.93 \$12.75 Food Waitperson \$13.24 \$9.93 \$11.59 Accounting/Night Audit Department \$19.27 \$14.45 \$16.86 Night Auditor \$18.80 \$14.10 \$16.45 Banquets Department \$17.51 \$13.13 \$15.32 Head Houseperson \$17.54 \$13.16 \$15.33 Houseperson \$16.26 \$12.20 \$14.23 Waitperson \$12.94 \$9.71 \$11.32 Bell/Door Service Department \$13.03 \$9.77 \$11.40 Head Bellperson \$13.03 \$9.77 \$11.40 Dorperson \$13.03 \$9.77 \$11.40 Dining Room - Mosaic Bar & Grille \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department< | Sandwich/Grill | \$18.21 | \$13.66 | \$15.93 |
| Sample | Bar Porter - Service Bar | \$17.62 | \$13.22 | \$15.42 |
| Accounting/Night Audit Department \$19.27 \$14.45 \$16.86 Night Auditor \$18.80 \$14.10 \$16.45 Banquets Department \$17.51 \$13.13 \$15.32 Head Houseperson \$17.54 \$13.16 \$15.35 Houseperson \$16.26 \$12.20 \$14.23 Waitperson \$13.75 \$10.31 \$12.03 Bell/Door Service Department \$13.75 \$10.31 \$12.03 Bellperson \$13.03 \$9.77 \$11.40 Doorperson \$13.03 \$9.77 \$11.40 Dining Room - Mosaic Bar & Grille \$16.02 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$12.94 \$9.71 \$11.32 Engineering Department \$12.94 \$9.71 \$11.32 Engineering Department \$23.63 \$17.72 \$20.68 Maintenance (Electric) \$23.63 \$17.72 \$20.68 | Waitperson | \$14.57 | \$10.93 | \$12.75 |
| Head Night Auditor | Food Waitperson | \$13.24 | \$9.93 | \$11.59 |
| Head Night Auditor | Accounting/Night Audit Department | | | |
| Night Auditor \$18.80 \$14.10 \$16.45 Banquets Department Cashier \$17.51 \$13.13 \$15.32 Head Houseperson \$17.54 \$13.16 \$15.35 Houseperson \$16.26 \$12.20 \$14.23 Waitperson \$12.94 \$9.71 \$11.32 Bell/Door Service Department Head Bellperson \$13.75 \$10.31 \$12.03 Bell/Person \$13.03 \$9.77 \$11.40 Dining Room - Mosaic Bar & Grille Host/Hostess \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 < | | \$19.27 | \$14.45 | \$16.86 |
| Banquets Department Cashier \$17.51 \$13.13 \$15.32 Head Houseperson \$16.26 \$12.20 \$14.23 Houseperson \$16.26 \$12.20 \$14.23 Waitperson \$12.94 \$9.71 \$11.32 Bell/Door Service Department Head Bellperson \$13.75 \$10.31 \$12.03 Bellperson \$13.03 \$9.77 \$11.40 Dining Room - Mosaic Bar & Grille Host/Hostess \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 < | | | | |
| Cashier \$17.51 \$13.13 \$15.32 Head Houseperson \$17.54 \$13.16 \$15.35 Houseperson \$16.26 \$12.20 \$14.23 Waitperson \$12.94 \$9.71 \$11.32 Bell/Door Service Department Head Bellperson \$13.75 \$10.31 \$12.03 Bellperson \$13.03 \$9.77 \$11.40 Doorperson \$13.03 \$9.77 \$11.40 Dining Room - Mosaic Bar & Grille Host/Hostess \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Woodwork) \$22.56 \$16.92 \$19.74 Maintenance (Minor Repair) \$20.67 \$15. | g.t./iaalio/ | V.0.00 | Ψσ | Ψ.σσ |
| Head Houseperson | · · · · · · · · · · · · · · · · · · · | | | |
| Houseperson | | | • | \$15.32 |
| Waitperson \$12.94 \$9.71 \$11.32 Bell/Door Service Department Head Bellperson \$13.75 \$10.31 \$12.03 Bellperson \$13.03 \$9.77 \$11.40 Doorperson \$13.03 \$9.77 \$11.40 Dining Room - Mosaic Bar & Grille Host/Hostess \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$13.76 \$10.32 \$12.04 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | Head Houseperson | \$17.54 | \$13.16 | \$15.35 |
| Bell/Door Service Department Head Bellperson \$13.75 \$10.31 \$12.03 Bellperson \$13.03 \$9.77 \$11.40 Doorperson Dining Room - Mosaic Bar & Grille Host/Hostess Mixerologist/Sommelier \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | Houseperson | \$16.26 | \$12.20 | \$14.23 |
| Head Bellperson | Waitperson | \$12.94 | \$9.71 | \$11.32 |
| Bellperson \$13.03 \$9.77 \$11.40 Doorperson \$13.03 \$9.77 \$11.40 Dining Room - Mosaic Bar & Grille Host/Hostess \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | Bell/Door Service Department | | | |
| Dining Room - Mosaic Bar & Grille Host/Hostess \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department \$23.63 \$17.72 \$20.68 Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | Head Bellperson | \$13.75 | \$10.31 | \$12.03 |
| Dining Room - Mosaic Bar & Grille Host/Hostess \$16.76 \$12.57 \$14.67 Mixerologist/Sommelier \$16.02 \$12.02 \$14.02 Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department *** *** *** \$20.68 Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | Bellperson | \$13.03 | \$9.77 | \$11.40 |
| Host/Hostess | Doorperson | \$13.03 | \$9.77 | \$11.40 |
| Host/Hostess | Dining Room - Mosaic Bar & Grille | | | |
| Busperson \$15.58 \$11.69 \$13.63 Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | _ | \$16.76 | \$12.57 | \$14.67 |
| Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Waintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | Mixerologist/Sommelier | \$16.02 | \$12.02 | \$14.02 |
| Bar Waitperson \$13.76 \$10.32 \$12.04 Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Waintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | Busperson | \$15.58 | \$11.69 | \$13.63 |
| Waitperson \$12.94 \$9.71 \$11.32 Engineering Department Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | • | | | |
| Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | | \$12.94 | | |
| Maintenance (Electric) \$23.63 \$17.72 \$20.68 Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | Engineering Department | | | |
| Maintenance (Plumber) \$22.56 \$16.92 \$19.74 Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | | \$23.63 | \$17.72 | \$20.68 |
| Maintenance (Woodwork) \$21.74 \$16.31 \$19.02 Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | | | | |
| Maintenance (Minor Repair) \$20.67 \$15.50 \$18.09 General Maintenance \$19.84 \$14.88 \$17.36 | | | | |
| General Maintenance \$19.84 \$14.88 \$17.36 | | | | |
| | | | | |
| Front Office Department | Constantial Indication | ψ13.04 | ψ1-1.00 | ψ17.50 |
| - I | Front Office Department | | | |
| Night Clerk \$18.80 \$14.10 \$16.45 | Night Clerk | \$18.80 | \$14.10 | \$16.45 |
| Senior Desk Clerk \$18.64 \$13.98 \$16.31 | Senior Desk Clerk | \$18.64 | \$13.98 | \$16.31 |

| Front Desk Clerk HYATT REGENCY VANCOUVER | \$17.95 | \$13.46 | \$15.71 |
|--|--------------|--------------|--------------|
| THAT REGERET VARIOUS VER | JULY 1, 2009 | | |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Housekeeping Department | | | |
| Inspector/Inspectress | \$18.19 | \$13.64 | \$15.92 |
| Tailor/Seamstress | \$18.09 | \$13.57 | \$15.83 |
| Housekeeping House Attendant | \$18.05 | \$13.54 | \$15.79 |
| Housekeeper | \$17.76 | \$13.32 | \$15.54 |
| Linen Attendant | \$17.76 | \$13.32 | \$15.54 |
| Laundry Attendant | \$17.68 | \$13.26 | \$15.47 |
| Room Attendant | \$17.55 | \$13.16 | \$15.36 |
| Kitchen Department | | | |
| First Cook | \$20.74 | \$15.56 | \$18.15 |
| Second Cook | \$19.78 | \$14.84 | \$17.31 |
| Third Cook | \$18.74 | \$14.06 | \$16.40 |
| Pastry Cook | \$18.36 | \$13.77 | \$16.07 |
| Salad/Sandwich | \$18.11 | \$13.58 | \$15.85 |
| Cook's Helper | \$17.89 | \$13.42 | \$15.65 |
| PBX/Switchboard Department | | | |
| Chief Switchboard Operator | \$18.02 | \$13.52 | \$15.77 |
| Switchboard Operator | \$17.64 | \$13.23 | \$15.44 |
| Room Service Department | | | |
| Executive Bar Attendant | \$17.62 | \$13.22 | \$15.42 |
| Cashier | \$17.51 | \$13.13 | \$15.32 |
| Captain | \$14.18 | \$10.64 | \$12.41 |
| Waitperson | \$12.94 | \$9.71 | \$11.32 |
| Stewards Department | | | |
| Janitor | \$18.05 | \$13.54 | \$15.79 |
| Steward | \$17.95 | \$13.46 | \$15.71 |
| Dishwasher | \$17.51 | \$13.13 | \$15.32 |
| Storeroom/Receiving Department | | | |
| Head Storekeeper | \$19.33 | \$14.50 | \$16.91 |
| Storekeeper | \$18.53 | \$13.90 | \$16.21 |
| Latte Café & Bistro | | | |
| Bistro Supervisor | \$17.03 | \$12.77 | \$14.90 |
| Bistro Server | \$15.89 | \$11.92 | \$13.90 |
| | | | |

| HYATT REGENCY VANCOUVER | | | |
|-----------------------------------|-----------------|--------------|--------------|
| | JANUARY 1, 2010 | | |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Beverage Department | | | |
| First Mixerologist (Service) | \$21.05 | \$15.79 | \$18.42 |
| Second Mixerologist (Service) | \$20.81 | \$15.61 | \$18.21 |
| First Mixerologist | \$16.69 | \$12.52 | \$14.60 |
| Second Mixerologist | \$16.48 | \$12.36 | \$14.42 |
| Floorperson | \$19.73 | \$14.80 | \$17.26 |
| Sandwich/Grill | \$18.57 | \$13.93 | \$16.25 |
| Bar Porter - Service Bar | \$17.97 | \$13.48 | \$15.72 |
| Waitperson | \$14.86 | \$11.15 | \$13.00 |
| Food Waitperson | \$13.50 | \$10.13 | \$11.81 |
| Accounting/Night Audit Department | | | |
| Head Night Auditor | \$19.66 | \$14.75 | \$17.20 |
| Night Auditor | \$19.18 | \$14.39 | \$16.78 |
| Banquets Department | | | |
| Cashier | \$17.86 | \$13.40 | \$15.63 |
| Head Houseperson | \$17.89 | \$13.42 | \$15.65 |
| Houseperson | \$16.59 | \$12.44 | \$14.52 |
| Waitperson | \$13.20 | \$9.90 | \$11.55 |
| Bell/Door Service Department | | | |
| Head Bellperson | \$14.03 | \$10.52 | \$12.28 |
| Bellperson | \$13.29 | \$9.97 | \$11.63 |
| Doorperson | \$13.29 | \$9.97 | \$11.63 |
| Dining Room - Mosaic Bar & Grille | | | |
| Host/Hostess | \$17.10 | \$12.83 | \$14.96 |
| Mixerologist/Sommelier | \$16.34 | \$12.26 | \$14.30 |
| Busperson | \$15.89 | \$11.92 | \$13.90 |
| Bar Waitperson | \$14.04 | \$10.53 | \$12.29 |
| Waitperson | \$13.20 | \$9.90 | \$11.55 |
| Engineering Department | | | |
| Maintenance (Electric) | \$24.10 | \$18.08 | \$21.09 |
| Maintenance (Plumber) | \$23.01 | \$17.26 | \$20.13 |
| Maintenance (Woodwork) | \$22.17 | \$16.63 | \$19.40 |
| Maintenance (Minor Repair) | \$21.08 | \$15.81 | \$18.45 |
| General Maintenance | \$20.24 | \$15.18 | \$17.71 |
| Front Office Department | | | |
| Night Clerk | \$19.18 | \$14.39 | \$16.78 |
| Senior Desk Clerk | \$19.01 | \$14.26 | \$16.63 |
| Front Desk Clerk | \$18.31 | \$13.73 | \$16.02 |
| | | | |

| HYATT REGENCY VANCOUVER | | | |
|--------------------------------|-----------------|--------------|--------------|
| | JANUARY 1, 2010 | | |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Housekeeping Department | Rate | 0 11103 | 0 14103 |
| Inspector/Inspectress | \$18.55 | \$13.91 | \$16.23 |
| Tailor/Seamstress | \$18.45 | | \$16.14 |
| Housekeeping House Attendant | \$18.41 | | \$16.11 |
| Housekeeper | \$18.12 | • | |
| Linen Attendant | \$18.12 | | \$15.86 |
| Laundry Attendant | \$18.03 | | |
| Room Attendant | \$17.90 | \$13.43 | \$15.66 |
| Kitchen Department | | | |
| First Cook | \$21.15 | \$15.86 | \$18.51 |
| Second Cook | \$20.18 | \$15.14 | \$17.66 |
| Third Cook | \$19.11 | \$14.33 | \$16.72 |
| Pastry Cook | \$18.73 | \$14.05 | \$16.39 |
| Salad/Sandwich | \$18.47 | \$13.85 | \$16.16 |
| Cook's Helper | \$18.25 | \$13.69 | \$15.97 |
| PBX/Switchboard Department | | | |
| Chief Switchboard Operator | \$18.38 | \$13.79 | \$16.08 |
| Switchboard Operator | \$17.99 | \$13.49 | \$15.74 |
| Room Service Department | | | |
| Executive Bar Attendant | \$17.97 | \$13.48 | \$15.72 |
| Cashier | \$17.86 | \$13.40 | \$15.63 |
| Captain | \$14.46 | \$10.85 | \$12.65 |
| Waitperson | \$13.20 | \$9.90 | \$11.55 |
| Stewards Department | | | |
| Janitor | \$18.41 | \$13.81 | \$16.11 |
| Steward | \$18.31 | \$13.73 | \$16.02 |
| Dishwasher | \$17.86 | \$13.40 | \$15.63 |
| Storeroom/Receiving Department | | | |
| Head Storekeeper | \$19.72 | \$14.79 | \$17.26 |
| Storekeeper | \$18.90 | \$14.18 | \$16.54 |
| Latte Café & Bistro | | | |
| Bistro Supervisor | \$17.37 | \$13.03 | \$15.20 |
| Bistro Server | \$16.21 | \$12.16 | \$14.18 |

RENAISSANCE VANCOUVER HOTEL HARBOURSIDE

JULY 1, 2007

| | | , | |
|--|--------------|--------------|--------------|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Employees | | | |
| Club Lounge Attendant | \$17.02 | \$12.77 | \$14.89 |
| Second Mixerologist | \$15.04 | \$11.28 | \$13.16 |
| Second Mixerologist (Service) | \$19.05 | \$14.29 | \$16.67 |
| Cocktail Server | \$13.98 | \$10.49 | \$12.23 |
| Server (Food Only) | \$12.70 | \$9.53 | \$11.11 |
| Host/Hostess | \$17.02 | \$12.77 | \$14.89 |
| Service Employees | | | |
| Desk Clerk | \$17.65 | \$13.24 | \$15.44 |
| Night Desk Clerk | \$18.34 | \$13.76 | \$16.05 |
| Night Auditor | \$18.34 | \$13.76 | \$16.05 |
| Reservations Agent | \$17.65 | \$13.24 | \$15.44 |
| Switchboard Operator | \$17.20 | \$12.90 | \$15.05 |
| Room Controller | \$17.65 | \$13.24 | \$15.44 |
| Room Attendant | \$17.05 | \$12.79 | \$14.92 |
| Lead Housekeeper | \$17.14 | \$12.86 | \$15.00 |
| Head Housekeeper | \$17.15 | \$12.86 | \$15.01 |
| Laundry Help | \$17.05 | \$12.79 | \$14.92 |
| Laundry Supervisor | \$17.30 | \$12.98 | \$15.14 |
| Seamstress/Uniform Attendant | \$17.05 | \$12.79 | \$14.92 |
| Bell Captain | \$12.65 | \$9.49 | \$11.07 |
| Bell Person | \$12.10 | \$9.08 | \$10.59 |
| Doorperson | \$12.65 | \$9.49 | \$11.07 |
| General Maintenance | \$19.66 | \$14.75 | \$17.20 |
| General Maintenance - Wage Adjustment 10/01/07 | \$20.16 | \$15.12 | \$17.64 |
| Porter/Utility Janitor | \$16.95 | \$12.71 | \$14.83 |
| Banquet Houseperson | \$12.93 | \$9.70 | \$11.31 |
| Banquet Server | \$12.23 | \$9.17 | \$10.70 |
| Banquet Captain | \$13.70 | \$10.28 | \$11.99 |
| Houseperson | \$16.80 | \$12.60 | \$14.70 |
| Houseperson (Nights) | \$16.95 | \$12.71 | \$14.83 |
| Storeperson & Receiver | \$17.33 | \$13.00 | \$15.16 |
| Purchaser & Receiver | \$18.86 | \$14.15 | \$16.50 |
| Chef de Partie | \$19.98 | \$14.99 | \$17.48 |
| 1st Cook | \$19.00 | \$14.25 | \$16.63 |
| 2nd Cook | \$18.16 | \$13.62 | \$15.89 |
| 3rd Cook | \$17.19 | \$12.89 | \$15.04 |
| Apprentice | \$16.19 | \$12.14 | \$14.17 |
| Kitchen Helper | \$16.37 | \$12.28 | \$14.32 |
| Dishwasher | \$16.81 | \$12.61 | \$14.71 |
| Night Cleaner | \$16.95 | \$12.71 | \$14.83 |
| Dishwasher Supervisor | \$17.16 | \$12.87 | \$15.02 |
| Server (Restaurant) | \$12.23 | \$9.17 | \$10.70 |
| Busperson (Restaurant) | \$14.79 | \$11.09 | \$12.94 |
| , , | | | |

RENAISSANCE VANCOUVER HOTEL HARBOURSIDE

JULY 1, 2007

| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
|-----------------------------------|--------------|--------------|--------------|
| Host/Hostess/Cashier (Restaurant) | \$17.02 | \$12.77 | \$14.89 |
| Room Service Server | \$12.23 | \$9.17 | \$10.70 |
| Room Service Captain | \$17.02 | \$12.77 | \$14.89 |
| Mini Bar Attendant | \$16.40 | \$12.30 | \$14.35 |
| Plantations Server/Cashier | \$15.29 | \$11.47 | \$13.38 |
| Plantations Supervisor | \$16.37 | \$12.28 | \$14.32 |

| RENAISSANCE VANCOUVER HOTEL | |
|-----------------------------|--|
| HARBOURSIDE | |

JULY 1, 2008 1st

| | • | ., | |
|-------------------------------|--------------|--------------|--------------|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Employees | | | |
| Club Lounge Attendant | \$17.36 | \$13.02 | \$15.19 |
| Second Mixerologist | \$15.34 | \$11.51 | \$13.42 |
| Second Mixerologist (Service) | \$19.43 | \$14.57 | \$17.00 |
| Cocktail Server | \$14.26 | \$10.70 | \$12.48 |
| Server (Food Only) | \$12.95 | \$9.71 | \$11.33 |
| Host/Hostess | \$17.36 | \$13.02 | \$15.19 |
| Service Employees | | | |
| Desk Clerk | \$18.00 | \$13.50 | \$15.75 |
| Night Desk Clerk | \$18.71 | \$14.03 | \$16.37 |
| Night Auditor | \$18.71 | \$14.03 | \$16.37 |
| Reservations Agent | \$18.00 | \$13.50 | \$15.75 |
| Switchboard Operator | \$17.54 | \$13.16 | \$15.35 |
| Room Controller | \$18.00 | \$13.50 | \$15.75 |
| Room Attendant | \$17.39 | \$13.04 | \$15.22 |
| Lead Housekeeper | \$17.48 | \$13.11 | \$15.30 |
| Head Housekeeper | \$17.49 | \$13.12 | \$15.30 |
| Laundry Help | \$17.39 | \$13.04 | \$15.22 |
| Laundry Supervisor | \$17.65 | \$13.24 | \$15.44 |
| Seamstress/Uniform Attendant | \$17.39 | \$13.04 | \$15.22 |
| Bell Captain | \$12.90 | \$9.68 | \$11.29 |
| Bell Person | \$12.34 | \$9.26 | \$10.80 |
| Doorperson | \$12.90 | \$9.68 | \$11.29 |
| General Maintenance | \$20.56 | \$15.42 | \$17.99 |
| Porter/Utility Janitor | \$17.29 | \$12.97 | \$15.13 |
| Banquet Houseperson | \$13.19 | \$9.89 | \$11.54 |
| Banquet Server | \$12.47 | \$9.35 | \$10.91 |
| Banquet Captain | \$13.97 | \$10.48 | \$12.22 |
| Houseperson | \$17.14 | \$12.86 | \$15.00 |
| Houseperson (Nights) | \$17.29 | \$12.97 | \$15.13 |
| Storeperson & Receiver | \$17.68 | \$13.26 | \$15.47 |
| Purchaser & Receiver | \$19.24 | \$14.43 | \$16.84 |
| Chef de Partie | \$20.38 | \$15.29 | \$17.83 |
| 1st Cook | \$19.38 | \$14.54 | \$16.96 |
| 2nd Cook | \$18.52 | \$13.89 | \$16.21 |
| 3rd Cook | \$17.53 | \$13.15 | \$15.34 |
| Apprentice | \$16.51 | \$12.38 | \$14.45 |
| Kitchen Helper | \$16.70 | \$12.53 | \$14.61 |
| Dishwasher | \$17.15 | \$12.86 | \$15.01 |
| Night Cleaner | \$17.29 | \$12.97 | \$15.13 |
| Dishwasher Supervisor | \$17.50 | \$13.13 | \$15.31 |
| Server (Restaurant) | \$12.47 | \$9.35 | \$10.91 |
| Busperson (Restaurant) | \$15.09 | \$11.32 | \$13.20 |
| | | | |

RENAISSANCE VANCOUVER HOTEL HARBOURSIDE

JULY 1, 2008

| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
|-----------------------------------|--------------|--------------|--------------|
| Host/Hostess/Cashier (Restaurant) | \$17.36 | \$13.02 | \$15.19 |
| Room Service Server | \$12.47 | \$9.35 | \$10.91 |
| Room Service Captain | \$17.36 | \$13.02 | \$15.19 |
| Mini Bar Attendant | \$16.73 | \$12.55 | \$14.64 |
| Plantations Server/Cashier | \$15.60 | \$11.70 | \$13.65 |
| Plantations Supervisor | \$16.70 | \$12.53 | \$14.61 |

| RENAISSANCE | VANCOUVER HOTEL |
|--------------------|-----------------|
| HAPROLIPSIDE | |

| | JANUARY 1, 2009 | | |
|------------------------------------|------------------------|-------------------|--------------------|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Employees | A | | ^ |
| Club Lounge Attendant | \$17.71 | \$13.28 | \$15.50 |
| Second Mixerologist | \$15.65 | \$11.74 | \$13.69 |
| Second Mixerologist (Service) | \$19.82 | \$14.87 | \$17.34 |
| Cocktail Server | \$14.55 | \$10.91 | \$12.73 |
| Server (Food Only) Host/Hostess | \$13.21 \$17.71 | \$9.91 \$13.28 | \$11.56 \$15.50 |
| i iosui iosiess | φ17.71 | φ13.20 | φ15.50 |
| Service Employees | | | |
| Desk Clerk | \$18.36 | \$13.77 | \$16.07 |
| Night Desk Clerk | \$19.08 | \$14.31 | \$16.70 |
| Night Auditor | \$19.08 | \$14.31 | \$16.70 |
| Reservations Agent | \$18.36 | \$13.77 | \$16.07 |
| Switchboard Operator | \$17.89 | \$13.42 | \$15.65 |
| Room Controller | \$18.36 | \$13.77 | \$16.07 |
| Room Attendant | \$17.74 | \$13.31 | \$15.52 |
| Lead Housekeeper | \$17.83 | \$13.37 | \$15.60 |
| Head Housekeeper | \$17.84 | \$13.38 | \$15.61 |
| Laundry Help | \$17.74 | \$13.31 | \$15.52 |
| Laundry Supervisor | \$18.00 | \$13.50 | \$15.75 |
| Seamstress/Uniform Attendant | \$17.74 | \$13.31 | \$15.52 |
| Bell Captain | \$13.16 | \$9.87 | \$11.52 |
| Bell Person Doorperson | \$12.59 \$13.16 | \$9.44 \$9.87 | \$11.02 \$11.52 |
| General Maintenance | \$20.97 | \$15.73 | \$18.35 |
| Porter/Utility Janitor | \$17.64 | \$13.23 | \$15.44 |
| Banquet Houseperson | \$13.45 | \$10.09 | \$11.77 |
| Banquet Server | \$12.72 | \$9.54 | \$11.13 |
| Banquet Captain | \$14.25 | | \$12.47 |
| Houseperson | \$17.48 | \$13.11 | \$15.30 |
| Houseperson (Nights) | \$17.64 | \$13.23 | \$15.44 |
| Storeperson & Receiver | \$18.03 | \$13.52 | \$15.78 |
| Purchaser & Receiver | \$19.62 | \$14.72 | \$17.17 |
| Chef de Partie | \$20.79 | \$15.59 | \$18.19 |
| 1st Cook | \$19.77 | \$14.83 | \$17.30 |
| 2nd Cook | \$18.89 | \$14.17 | \$16.53 |
| 3rd Cook | \$17.88 | \$13.41 | \$15.65 |
| Apprentice | \$16.84 | \$12.63 | \$14.74 |
| Kitchen Helper | \$17.03 | \$12.77 | \$14.90 |
| Dishwasher | \$17.49 | \$13.12 | \$15.30 |
| Night Cleaner | \$17.64 | \$13.23 | \$15.44 |
| Dishwasher Supervisor | \$17.85 | \$13.39 | \$15.62 |
| Server (Restaurant) | \$12.72 | \$9.54 | \$11.13 |
| Busperson (Restaurant) | \$15.39 | \$11.54 | \$13.47 |

RENAISSANCE VANCOUVER HOTEL HARBOURSIDE

JANUARY 1, 2009

| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
|-----------------------------------|--------------|--------------|--------------|
| Host/Hostess/Cashier (Restaurant) | \$17.71 | \$13.28 | \$15.50 |
| Room Service Server | \$12.72 | \$9.54 | \$11.13 |
| Room Service Captain | \$17.71 | \$13.28 | \$15.50 |
| Mini Bar Attendant | \$17.06 | \$12.80 | \$14.93 |
| Plantations Server/Cashier | \$15.91 | \$11.93 | \$13.92 |
| Plantations Supervisor | \$17.03 | \$12.77 | \$14.90 |
| | | | |

| RENAISSANCE VANCOUVER HOTEL |
|-----------------------------|
| HARBOURSIDE |

JULY 1, 2009

| | 3021 1, 2009 | | |
|-------------------------------|--------------|--------------|--------------|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Employees | | | |
| Club Lounge Attendant | \$18.06 | \$13.55 | \$15.80 |
| Second Mixerologist | \$15.96 | \$11.97 | \$13.97 |
| Second Mixerologist (Service) | \$20.22 | \$15.17 | \$17.69 |
| Cocktail Server | \$14.84 | \$11.13 | \$12.99 |
| Server (Food Only) | \$13.47 | \$10.10 | \$11.79 |
| Host/Hostess | \$18.06 | \$13.55 | \$15.80 |
| Service Employees | | | |
| Desk Clerk | \$18.73 | \$14.05 | \$16.39 |
| Night Desk Clerk | \$19.46 | \$14.60 | \$17.03 |
| Night Auditor | \$19.46 | \$14.60 | \$17.03 |
| Reservations Agent | \$18.73 | \$14.05 | \$16.39 |
| Switchboard Operator | \$18.25 | \$13.69 | \$15.97 |
| Room Controller | \$18.73 | \$14.05 | \$16.39 |
| Room Attendant | \$18.09 | \$13.57 | \$15.83 |
| Lead Housekeeper | \$18.19 | \$13.64 | \$15.92 |
| Head Housekeeper | \$18.20 | \$13.65 | \$15.93 |
| Laundry Help | \$18.09 | \$13.57 | \$15.83 |
| Laundry Supervisor | \$18.36 | \$13.77 | \$16.07 |
| Seamstress/Uniform Attendant | \$18.09 | \$13.57 | \$15.83 |
| Bell Captain | \$13.42 | \$10.07 | \$11.74 |
| Bell Person | \$12.84 | \$9.63 | \$11.24 |
| Doorperson | \$13.42 | \$10.07 | \$11.74 |
| General Maintenance | \$21.39 | \$16.04 | \$18.72 |
| Porter/Utility Janitor | \$17.99 | \$13.49 | \$15.74 |
| Banquet Houseperson | \$13.72 | \$10.29 | \$12.01 |
| Banquet Server | \$12.97 | \$9.73 | \$11.35 |
| Banquet Captain | \$14.54 | \$10.91 | \$12.72 |
| Houseperson | \$17.83 | \$13.37 | \$15.60 |
| Houseperson (Nights) | \$17.99 | \$13.49 | \$15.74 |
| Storeperson & Receiver | \$18.39 | \$13.79 | \$16.09 |
| Purchaser & Receiver | \$20.01 | \$15.01 | \$17.51 |
| Chef de Partie | \$21.21 | \$15.91 | \$18.56 |
| 1st Cook | \$20.17 | \$15.13 | \$17.65 |
| 2nd Cook | \$19.27 | \$14.45 | \$16.86 |
| 3rd Cook | \$18.24 | \$13.68 | \$15.96 |
| Apprentice | \$17.18 | \$12.89 | \$15.03 |
| Kitchen Helper | \$17.37 | \$13.03 | \$15.20 |
| Dishwasher | \$17.84 | \$13.38 | \$15.61 |
| Night Cleaner | \$17.99 | \$13.49 | \$15.74 |
| Dishwasher Supervisor | \$18.21 | \$13.66 | \$15.93 |
| Server (Restaurant) | \$12.97 | \$9.73 | \$11.35 |
| Busperson (Restaurant) | \$15.70 | \$11.78 | \$13.74 |
| | | | |

| Host/Hostess/Cashier (Restaurant) | \$18.06 | \$13.55 | \$15.80 |
|-----------------------------------|---------|---------|---------|
| | | | |

RENAISSANCE VANCOUVER HOTEL HARBOURSIDE

JULY 1, 2009

| | · | | |
|----------------------------|--------------|--------------|--------------|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Room Service Server | \$12.97 | \$9.73 | \$11.35 |
| Room Service Captain | \$18.06 | \$13.55 | \$15.80 |
| Mini Bar Attendant | \$17.40 | \$13.05 | \$15.23 |
| Plantations Server/Cashier | \$16.23 | \$12.17 | \$14.20 |
| Plantations Supervisor | \$17.37 | \$13.03 | \$15.20 |
| | | | |

| RENAISSANCE | VANCOUVER HO | TEL |
|--------------------|--------------|-----|
| HARBOURSIDE | | |

| | JANUARY 1, 2010 | | 2010 |
|-------------------------------|--------------------|-------------------|--------------------|
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Lounge Employees | | | |
| Club Lounge Attendant | \$18.42 | \$13.82 | \$16.12 |
| Second Mixerologist | \$16.28 | \$12.21 | \$14.25 |
| Second Mixerologist (Service) | \$20.62 | \$15.47 | \$18.04 |
| Cocktail Server | \$15.14 | | |
| Server (Food Only) | \$13.74 | | |
| Host/Hostess | \$18.42 | \$13.82 | \$16.12 |
| Service Employees | | | |
| Desk Clerk | \$19.10 | \$14.33 | \$16.71 |
| Night Desk Clerk | \$19.85 | \$14.89 | \$17.37 |
| Night Auditor | \$19.85 | \$14.89 | \$17.37 |
| Reservations Agent | \$19.10 | \$14.33 | \$16.71 |
| Switchboard Operator | \$18.62 | \$13.97 | \$16.29 |
| Room Controller | \$19.10 | \$14.33 | \$16.71 |
| Room Attendant | \$18.45 | \$13.84 | \$16.14 |
| Lead Housekeeper | \$18.55 | \$13.91 | \$16.23 |
| Head Housekeeper | \$18.56 | \$13.92 | \$16.24 |
| Laundry Help | \$18.45 | \$13.84 | \$16.14 |
| Laundry Supervisor | \$18.73 | \$14.05 | \$16.39 |
| Seamstress/Uniform Attendant | \$18.45 | \$13.84 | \$16.14 |
| Bell Captain | \$13.69 | \$10.27 | \$11.98 |
| Bell Person | \$13.10 | \$9.83 | \$11.46 |
| Doorperson | \$13.69 | \$10.27 | \$11.98 |
| General Maintenance | \$21.82 | \$16.37 | \$19.09 |
| Porter/Utility Janitor | \$18.35 | \$13.76 | \$16.06 |
| Banquet Houseperson | \$13.99 | \$10.49 | \$12.24 |
| Banquet Server | \$13.23 | \$9.92 | \$11.58 |
| Banquet Captain | \$14.83 | | \$12.98 |
| Houseperson | \$18.19 | \$13.64 | |
| Houseperson (Nights) | \$18.35 | \$13.76 | \$16.06 |
| Storeperson & Receiver | \$18.76 | | |
| Purchaser & Receiver | \$20.41 | \$15.31 | \$17.86 |
| Chef de Partie | \$21.63 | \$16.22 | \$18.93 |
| 1st Cook | \$20.57 | \$15.43 | \$18.00 |
| 2nd Cook | \$19.66 | \$14.75 | \$17.20 |
| 3rd Cook | \$18.60 | \$13.95 | \$16.28 |
| Apprentice | \$17.52 | \$13.14 | \$15.33 |
| Kitchen Helper | \$17.72 | \$13.29 | \$15.51 \$15.03 |
| Dishwasher | \$18.20 | \$13.65 | \$15.93 |
| Night Cleaner | \$18.35 | \$13.76 | \$16.06 |
| Dishwasher Supervisor | \$18.57 | \$13.93 | \$16.25 |
| Server (Restaurant) | \$13.23 \$16.01 | \$9.92 \$12.01 | \$11.58 \$14.01 |
| Busperson (Restaurant) | \$16.01 | \$12.01 | \$14.01 |

| Host/Hostess/Cashier (Restaurant) | \$18.42 | \$13.82 | \$16.12 |
|--|--------------|--------------|--------------|
| RENAISSANCE VANCOUVER HOTEL HARBOURSIDE | | | |
| | JAN | UARY 1, | 2010 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Room Service Server | \$13.23 | \$9.92 | \$11.58 |
| Room Service Captain | \$18.42 | \$13.82 | \$16.12 |
| Mini Bar Attendant | \$17.75 | \$13.31 | \$15.53 |
| Plantations Server/Cashier | \$16.55 | \$12.41 | \$14.48 |
| Plantations Supervisor | \$17.72 | \$13.29 | \$15.51 |

JULY 1, 2007

| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
|--|--------------|--------------|--------------|
| CLASSIFICATIONS | | | |
| Accounting/Finance | | | |
| Accounting Agent | \$16.98 | \$12.74 | \$14.86 |
| A/R A/P | \$17.20 | \$12.90 | \$15.05 |
| Night Auditor | \$17.44 | \$13.08 | \$15.26 |
| Banquets | | | |
| Banquet Captain | \$14.08 | \$10.56 | \$12.32 |
| Banquet Server | \$12.44 | \$9.33 | \$10.89 |
| House Person | \$12.80 | \$9.60 | \$11.20 |
| Bartender | \$15.26 | \$11.45 | \$13.35 |
| Currents at Bayshore | | | |
| Cashier/Host/Hostess | \$17.46 | \$13.10 | \$15.28 |
| Server | \$12.98 | \$9.74 | \$11.36 |
| Bartender | \$15.04 | \$11.28 | \$13.16 |
| Engineering | | | |
| Maintenance Worker - Lead | \$20.76 | \$15.57 | \$18.17 |
| Maintenance Worker - General | \$19.97 | \$14.98 | \$17.47 |
| Power Engineering Ticket Premium | \$1.00 | | |
| Painter (to be deleted) | \$19.97 | \$14.98 | \$17.47 |
| Engineering Wage Adjustment - 10/01/07 | | | |
| Maintenance Worker - Lead | \$21.26 | \$15.95 | \$18.60 |
| Maintenance Worker - General | \$20.47 | \$15.35 | \$17.91 |
| Power Engineering Ticket Premium | \$1.00 | | |
| Painter (to be deleted) | \$20.47 | \$15.35 | \$17.91 |
| Front Office | | | |
| Front Office Agent | \$17.30 | \$12.98 | \$15.14 |
| Starbucks | | | |
| Barista | \$17.46 | \$13.10 | \$15.28 |
| Lead Barista | \$17.64 | \$13.23 | \$15.44 |
| Housekeeping | | | |
| Housekeeping - Lead | \$17.34 | \$13.01 | \$15.17 |
| Linen Room Assistant & Seamstress | \$17.25 | \$12.94 | \$15.09 |
| Room Attendant | \$17.19 | \$12.89 | \$15.04 |
| House Person | \$17.30 | \$12.98 | \$15.14 |
| Night Cleaner | \$17.47 | \$13.10 | \$15.29 |
| | | | |

| THE WESTIN BAYSHORE, VANCOUVER | | JLY 1, 200 | 0.7 |
|---|--------------|--------------|--------------|
| | 30 | JLT 1, 200 | J <i>i</i> |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| CLASSIFICATIONS | | | |
| Kitchen | | | |
| Chef de Partie | \$20.42 | \$15.32 | \$17.87 |
| First Cook | \$19.40 | \$14.55 | \$16.98 |
| Second Cook | \$18.19 | \$13.64 | \$15.92 |
| Third Cook | \$17.32 | \$12.99 | \$15.16 |
| Apprentice | \$16.42 | \$12.32 | \$14.37 |
| Purchasing | | | |
| Storekeeper | \$18.10 | \$13.58 | \$15.84 |
| Storekeeper - Helper | \$17.01 | \$12.76 | \$14.88 |
| Reservations | | | |
| Agent | \$17.24 | \$12.93 | \$15.09 |
| Service Express | | | |
| Door Person | \$12.97 | \$9.73 | \$11.35 |
| Service Express Agent - Concierge | \$17.47 | \$13.10 | \$15.29 |
| Service Express Agent | \$17.47 | \$13.10 | \$15.29 |
| Service Express Attendant-General/Lobby/Service | \$12.48 | \$9.36 | \$10.92 |
| Service Express - Valet | \$17.30 | \$12.98 | \$15.14 |
| Refreshment Centre Attendant | \$15.96 | \$11.97 | \$13.97 |
| Stewarding | | | |
| Steward Lead | \$17.54 | \$13.16 | \$15.35 |
| Steward | \$17.32 | | \$15.16 |
| Dishwasher | \$17.24 | | \$15.09 |
| House Person/Janitor | \$17.41 | \$13.06 | \$15.23 |
| Night Cleaner | \$17.41 | \$13.06 | \$15.23 |
| Porter | \$17.32 | \$12.99 | \$15.16 |
| Yardman | \$17.32 | | \$15.16 |
| Cafeteria Attendant | \$17.24 | \$12.93 | \$15.09 |
| Seawall Bar & Bistro | | | |
| Host/Hostess | \$17.15 | \$12.86 | \$15.01 |
| Server | \$11.81 | \$8.86 | \$10.33 |
| Bartender | \$13.97 | \$10.48 | \$12.22 |
| Chef de Partie | \$20.11 | \$15.08 | \$17.60 |
| Cook | \$17.89 | | \$15.65 |
| Steward | \$17.10 | \$12.83 | \$14.96 |
| Seawall Bar & Bistro - Wage Adjustment 10/01/07 | | | |
| Host/Hostess | \$17.65 | \$13.24 | \$15.44 |
| Server | \$12.31 | \$9.23 | \$10.77 |
| Bartender | \$14.47 | \$10.85 | \$12.66 |
| | • | | |

JULY 1, 2008

| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
|---|--------------|--------------|--------------|
| CLASSIFICATIONS | | | |
| Seawall Bar & Bistro - Wage Adjustment 01/01/08 | | | |
| Host/Hostess | \$18.15 | \$13.61 | \$15.88 |
| Server | \$12.81 | \$9.61 | \$11.21 |
| Bartender | \$14.97 | \$11.23 | \$13.10 |
| Accounting/Finance | | | |
| Accounting Agent | \$17.32 | \$12.99 | \$15.16 |
| A/R A/P | \$17.54 | \$13.16 | \$15.35 |
| Night Auditor | \$17.79 | \$13.34 | \$15.57 |
| Banquets | | | |
| Banquet Captain | \$14.36 | \$10.77 | \$12.57 |
| Banquet Server | \$12.69 | \$9.52 | \$11.10 |
| House Person | \$13.06 | \$9.80 | \$11.43 |
| Bartender | \$15.57 | \$11.68 | \$13.62 |
| Currents at Bayshore | | | |
| Cashier/Host/Hostess | \$17.81 | \$13.36 | \$15.58 |
| Server | \$13.24 | \$9.93 | \$11.59 |
| Bartender | \$15.34 | \$11.51 | \$13.42 |
| Engineering | | | |
| Maintenance Worker - Lead | \$21.69 | \$16.27 | \$18.98 |
| Maintenance Worker - General | \$20.88 | \$15.66 | \$18.27 |
| Power Engineering Ticket Premium | \$1.00 | | |
| Painter (to be deleted) | \$20.88 | \$15.66 | \$18.27 |
| Front Office | | | |
| Front Office Agent | \$17.65 | \$13.24 | \$15.44 |
| Starbucks | | | |
| Barista | \$17.81 | \$13.36 | \$15.58 |
| Lead Barista | \$17.99 | \$13.49 | \$15.74 |
| Housekeeping | | | |
| Housekeeping - Lead | \$17.69 | \$13.27 | \$15.48 |
| Linen Room Assistant & Seamstress | \$17.60 | \$13.20 | \$15.40 |
| Room Attendant | \$17.53 | \$13.15 | \$15.34 |
| House Person | \$17.65 | \$13.24 | \$15.44 |
| Night Cleaner | \$17.82 | \$13.37 | \$15.59 |
| | | | |

| THE WESTIN BAYSHORE, VANCOUVER | | | |
|---|--------------|--------------|--------------|
| | JI | JLY 1, 20 | 08 |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Kitchen | | | |
| Chef de Partie | \$20.83 | \$15.62 | \$18.23 |
| First Cook | \$19.79 | \$14.84 | \$17.32 |
| Second Cook | \$18.55 | | \$16.23 |
| Third Cook | \$17.67 | \$13.25 | \$15.46 |
| Apprentice | \$16.75 | \$12.56 | \$14.66 |
| Purchasing | | | |
| Storekeeper | \$18.46 | \$13.85 | \$16.15 |
| Storekeeper - Helper | \$17.35 | \$13.01 | \$15.18 |
| Reservations | | | |
| Agent | \$17.58 | \$13.19 | \$15.38 |
| Service Express | | | |
| Door Person | \$13.23 | \$9.92 | \$11.58 |
| Service Express Agent - Concierge | \$17.82 | \$13.37 | \$15.59 |
| Service Express Agent | \$17.82 | \$13.37 | \$15.59 |
| Service Express Attendant-General/Lobby/Service | \$12.73 | \$9.55 | \$11.14 |
| Service Express - Valet | \$17.65 | \$13.24 | \$15.44 |
| Refreshment Centre Attendant | \$16.28 | \$12.21 | \$14.25 |
| Stewarding | | | |
| Steward Lead | \$17.89 | \$13.42 | \$15.65 |
| Steward | \$17.67 | \$13.25 | \$15.46 |
| Dishwasher | \$17.58 | \$13.19 | \$15.38 |
| House Person/Janitor | \$17.76 | \$13.32 | \$15.54 |
| Night Cleaner | \$17.76 | \$13.32 | \$15.54 |
| Porter | \$17.67 | \$13.25 | \$15.46 |
| Yardman | \$17.67 | \$13.25 | \$15.46 |
| Cafeteria Attendant | \$17.58 | \$13.19 | \$15.38 |
| Seawall Bar & Bistro | | | |
| Host/Hostess | \$18.51 | \$13.88 | \$16.20 |
| Server | \$13.07 | \$9.80 | \$11.44 |
| Bartender | \$15.27 | \$11.45 | \$13.36 |
| Chef de Partie | \$20.51 | \$15.38 | \$17.95 |
| Cook | \$18.25 | \$13.69 | \$15.97 |
| Steward | \$17.44 | \$13.08 | \$15.26 |
| | | | |

JANUARY 1, 2009

| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
|-----------------------------------|--------------|--------------|--------------|
| CLASSIFICATIONS | | | |
| Accounting/Finance | | | |
| Accounting Agent | \$17.67 | \$13.25 | \$15.46 |
| A/R A/P | \$17.89 | \$13.42 | \$15.65 |
| Night Auditor | \$18.15 | \$13.61 | \$15.88 |
| Banquets | | | |
| Banquet Captain | \$14.65 | \$10.99 | \$12.82 |
| Banquet Server | \$12.94 | \$9.71 | \$11.32 |
| House Person | \$13.32 | \$9.99 | \$11.66 |
| Bartender | \$15.88 | \$11.91 | \$13.90 |
| Currents at Bayshore | | | |
| Cashier/Host/Hostess | \$18.17 | \$13.63 | \$15.90 |
| Server | \$13.50 | \$10.13 | \$11.81 |
| Bartender | \$15.65 | \$11.74 | \$13.69 |
| Engineering | | | |
| Maintenance Worker - Lead | \$22.12 | \$16.59 | \$19.36 |
| Maintenance Worker - General | \$21.30 | \$15.98 | \$18.64 |
| Power Engineering Ticket Premium | \$1.00 | | |
| Painter (to be deleted) | \$21.30 | \$15.98 | \$18.64 |
| Front Office | | | |
| Front Office Agent | \$18.00 | \$13.50 | \$15.75 |
| Starbucks | | | |
| Barista | \$18.17 | \$13.63 | \$15.90 |
| Lead Barista | \$18.35 | \$13.76 | \$16.06 |
| Housekeeping | | | |
| Housekeeping - Lead | \$18.04 | \$13.53 | \$15.79 |
| Linen Room Assistant & Seamstress | \$17.95 | \$13.46 | \$15.71 |
| Room Attendant | \$17.88 | \$13.41 | \$15.65 |
| House Person | \$18.00 | \$13.50 | \$15.75 |
| Night Cleaner | \$18.18 | \$13.64 | \$15.91 |
| Kitchen | | | |
| Chef de Partie | \$21.25 | \$15.94 | \$18.59 |
| First Cook | \$20.19 | \$15.14 | \$17.67 |
| Second Cook | \$18.92 | \$14.19 | \$16.56 |
| Third Cook | \$18.02 | \$13.52 | \$15.77 |
| Apprentice | \$17.09 | \$12.82 | \$14.95 |

| THE WESTIN BAYSHORE. | VANCOUVER |
|----------------------|-----------|
|----------------------|-----------|

JANUARY 2009

| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
|---|--------------|--------------|--------------|
| | raio | 0 11.00 | 0 11100 |
| Purchasing | | | |
| Storekeeper | \$18.83 | \$14.12 | \$16.48 |
| Storekeeper - Helper | \$17.70 | \$13.28 | \$15.49 |
| Reservations | | | |
| Agent | \$17.93 | \$13.45 | \$15.69 |
| Service Express | | | |
| Door Person | \$13.49 | \$10.12 | \$11.80 |
| Service Express Agent - Concierge | \$18.18 | \$13.64 | \$15.91 |
| Service Express Agent | \$18.18 | \$13.64 | \$15.91 |
| Service Express Attendant-General/Lobby/Service | \$12.98 | \$9.74 | \$11.36 |
| Service Express - Valet | \$18.00 | \$13.50 | \$15.75 |
| Refreshment Centre Attendant | \$16.61 | \$12.46 | \$14.53 |
| Stewarding | | | |
| Steward Lead | \$18.25 | \$13.69 | \$15.97 |
| Steward | \$18.02 | \$13.52 | \$15.77 |
| Dishwasher | \$17.93 | \$13.45 | \$15.69 |
| House Person/Janitor | \$18.12 | \$13.59 | \$15.86 |
| Night Cleaner | \$18.12 | \$13.59 | \$15.86 |
| Porter | \$18.02 | \$13.52 | \$15.77 |
| Yardman | \$18.02 | \$13.52 | \$15.77 |
| Cafeteria Attendant | \$17.93 | \$13.45 | \$15.69 |
| Seawall Bar & Bistro | | | |
| Host/Hostess | \$18.88 | \$14.16 | \$16.52 |
| Server | \$13.33 | \$10.00 | \$11.66 |
| Bartender | \$15.58 | \$11.69 | \$13.63 |
| Chef de Partie | \$20.92 | \$15.69 | \$18.31 |
| Cook | \$18.62 | \$13.97 | \$16.29 |
| Steward | \$17.79 | \$13.34 | \$15.57 |

JULY 1, 2009

| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
|-----------------------------------|--------------|--------------|--------------|
| CLASSIFICATIONS | | | |
| Accounting/Finance | | | |
| Accounting Agent | \$18.02 | \$13.52 | \$15.77 |
| A/R A/P | \$18.25 | | \$15.97 |
| Night Auditor | \$18.51 | \$13.88 | \$16.20 |
| Banquets | | | |
| Banquet Captain | \$14.94 | \$11.21 | \$13.07 |
| Banquet Server | \$13.20 | \$9.90 | \$11.55 |
| House Person | \$13.59 | \$10.19 | \$11.89 |
| Bartender | \$16.20 | \$12.15 | \$14.18 |
| Currents at Bayshore | | | |
| Cashier/Host/Hostess | \$18.53 | \$13.90 | \$16.21 |
| Server | \$13.77 | \$10.33 | \$12.05 |
| Bartender | \$15.96 | \$11.97 | \$13.97 |
| Engineering | | | |
| Maintenance Worker - Lead | \$22.56 | \$16.92 | \$19.74 |
| Maintenance Worker - General | \$21.73 | \$16.30 | \$19.01 |
| Power Engineering Ticket Premium | \$1.00 | | |
| Painter (to be deleted) | \$21.73 | \$16.30 | \$19.01 |
| Front Office | | | |
| Front Office Agent | \$18.36 | \$13.77 | \$16.07 |
| Starbucks | | | |
| Barista | \$18.53 | \$13.90 | \$16.21 |
| Lead Barista | \$18.72 | \$14.04 | \$16.38 |
| Housekeeping | | | |
| Housekeeping - Lead | \$18.40 | \$13.80 | \$16.10 |
| Linen Room Assistant & Seamstress | \$18.31 | \$13.73 | \$16.02 |
| Room Attendant | \$18.24 | \$13.68 | \$15.96 |
| House Person | \$18.36 | \$13.77 | \$16.07 |
| Night Cleaner | \$18.54 | \$13.91 | \$16.22 |
| Kitchen | | | |
| Chef de Partie | \$21.68 | \$16.26 | \$18.97 |
| First Cook | \$20.59 | \$15.44 | \$18.02 |
| Second Cook | \$19.30 | \$14.48 | \$16.89 |
| Third Cook | \$18.38 | \$13.79 | \$16.08 |

| Apprentice THE WESTIN BAYSHORE, VANCOUVER | \$17.43 | \$13.07 | \$15.25 |
|---|---------|------------|---------|
| | Jl | JLY 1, 200 | 09 |
| | Full | 1st | 2nd |
| | Rate | 6 Mos | 6 Mos |
| Purchasing | | | |
| Storekeeper | \$19.21 | \$14.41 | \$16.81 |
| Storekeeper - Helper | \$18.05 | \$13.54 | \$15.79 |
| Reservations | | | |
| Agent | \$18.29 | \$13.72 | \$16.00 |
| Service Express | | | |
| Door Person | \$13.76 | \$10.32 | \$12.04 |
| Service Express Agent - Concierge | \$18.54 | \$13.91 | \$16.22 |
| Service Express Agent | \$18.54 | \$13.91 | \$16.22 |
| Service Express Attendant-General/Lobby/Service | \$13.24 | \$9.93 | \$11.59 |
| Service Express - Valet | \$18.36 | \$13.77 | \$16.07 |
| Refreshment Centre Attendant | \$16.94 | \$12.71 | \$14.82 |
| Stewarding | | | |
| Steward Lead | \$18.62 | \$13.97 | \$16.29 |
| Steward | \$18.38 | \$13.79 | \$16.08 |
| Dishwasher | \$18.29 | \$13.72 | \$16.00 |
| House Person/Janitor | \$18.48 | \$13.86 | \$16.17 |
| Night Cleaner | \$18.48 | \$13.86 | \$16.17 |
| Porter | \$18.38 | \$13.79 | \$16.08 |
| Yardman | \$18.38 | \$13.79 | \$16.08 |
| Cafeteria Attendant | \$18.29 | \$13.72 | \$16.00 |
| Seawall Bar & Bistro | | | |
| Host/Hostess | \$19.26 | \$14.45 | \$16.85 |
| Server | \$13.60 | \$10.20 | \$11.90 |
| Bartender | \$15.89 | \$11.92 | \$13.90 |
| Chef de Partie | \$21.34 | \$16.01 | \$18.67 |
| Cook | \$18.99 | \$14.24 | \$16.62 |
| Steward | \$18.15 | \$13.61 | \$15.88 |

JANUARY 1, 2010

| | Full | 1st | 2nd |
|-----------------------------------|---------------|---------------|----------------|
| CLASSIFICATIONS | Rate | 6 Mos | 6 Mos |
| CLASSIFICATIONS | | | |
| Accounting/Finance | | | |
| Accounting Agent | \$18.38 | \$13.79 | \$16.08 |
| A/R A/P | \$18.62 | \$13.97 | \$16.29 |
| Night Auditor | \$18.88 | \$14.16 | \$16.52 |
| | | | |
| Banquets | | | |
| Banquet Captain | \$15.24 | \$11.43 | \$13.34 |
| Banquet Server | \$13.46 | \$10.10 | \$11.78 |
| House Person | \$13.86 | \$10.40 | \$12.13 |
| Bartender | \$16.52 | \$12.39 | \$14.46 |
| | | | |
| Currents at Bayshore | | | |
| Cashier/Host/Hostess | \$18.90 | \$14.18 | \$16.54 |
| Server | \$14.05 | \$10.54 | \$12.29 |
| Bartender | \$16.28 | \$12.21 | \$14.25 |
| Foodstanden | | | |
| Engineering | #00.04 | 047.00 | # 00.40 |
| Maintenance Worker - Lead | \$23.01 | \$17.26 | \$20.13 |
| Maintenance Worker - General | \$22.16 | \$16.62 | \$19.39 |
| Power Engineering Ticket Premium | \$1.00 | | |
| Painter (to be deleted) | \$22.16 | \$16.62 | \$19.39 |
| Front Office | | | |
| Front Office Agent | \$18.73 | \$14.05 | \$16.39 |
| Total Care August | V.S.1. | ψσσ | ψ.σ.σσ |
| Starbucks | | | |
| Barista | \$18.90 | \$14.18 | \$16.54 |
| Lead Barista | \$19.09 | \$14.32 | \$16.70 |
| | | | |
| Housekeeping | | | |
| Housekeeping - Lead | \$18.77 | \$14.08 | \$16.42 |
| Linen Room Assistant & Seamstress | \$18.68 | \$14.01 | \$16.35 |
| Room Attendant | \$18.60 | \$13.95 | \$16.28 |
| House Person | \$18.73 | \$14.05 | \$16.39 |
| Night Cleaner | \$18.91 | \$14.18 | \$16.55 |
| | | | |
| Kitchen | | | |
| Chef de Partie | \$22.11 | \$16.58 | \$19.35 |
| First Cook | \$21.00 | \$15.75 | \$18.38 |
| Second Cook | \$19.69 | \$14.77 | \$17.23 |
| Third Cook | \$18.75 | \$14.06 | \$16.41 |
| Apprentice | \$17.78 | \$13.34 | \$15.56 |
| | | | |

| THE WESTIN BAYSHORE, VANCOUVER | | | |
|---|-----------------|--------------|--------------|
| | JANUARY 1, 2010 | | |
| | Full Rate | 1st 6 Mos | 2nd 6 Mos |
| Purchasing | | | |
| Storekeeper | \$19.59 | \$14.69 | \$17.14 |
| Storekeeper - Helper | \$18.41 | \$13.81 | \$16.11 |
| Reservations | | | |
| Agent | \$18.66 | \$14.00 | \$16.33 |
| Service Express | | | |
| Door Person | \$14.04 | \$10.53 | \$12.29 |
| Service Express Agent - Concierge | \$18.91 | \$14.18 | \$16.55 |
| Service Express Agent | \$18.91 | \$14.18 | \$16.55 |
| Service Express Attendant-General/Lobby/Service | \$13.50 | \$10.13 | \$11.81 |
| Service Express - Valet | \$18.73 | \$14.05 | \$16.39 |
| Refreshment Centre Attendant | \$17.28 | \$12.96 | \$15.12 |
| Stewarding | | | |
| Steward Lead | \$18.99 | \$14.24 | \$16.62 |
| Steward | \$18.75 | \$14.06 | \$16.41 |
| Dishwasher | \$18.66 | \$14.00 | \$16.33 |
| House Person/Janitor | \$18.85 | \$14.14 | \$16.49 |
| Night Cleaner | \$18.85 | \$14.14 | \$16.49 |
| Porter | \$18.75 | \$14.06 | \$16.41 |
| Yardman | \$18.75 | \$14.06 | \$16.41 |
| Cafeteria Attendant | \$18.66 | \$14.00 | \$16.33 |
| Seawall Bar & Bistro | | | |
| Host/Hostess | \$19.65 | \$14.74 | \$17.19 |
| Server | \$13.87 | \$10.40 | \$12.14 |
| Bartender | \$16.21 | \$12.16 | \$14.18 |
| Chef de Partie | \$21.77 | \$16.33 | \$19.05 |
| Cook | \$19.37 | \$14.53 | \$16.95 |
| Steward | \$18.51 | \$13.88 | \$16.20 |